

# The Morphology of Loanwords in Urdu: the Persian, Arabic and English Strands

Submitted by:

Riaz Ahmed Islam

In fulfilment of the requirements for the degree of Doctor of Philosophy

Supervised by:

Dr. S. J. Hannahs & Prof. Maggie Tallerman

School of English Literature, Language & Linguistics Newcastle University

> October, 2011 © 2011 Riaz Ahmed Islam All rights reserved

#### **Abstract**

#### The Morphology of Loanwords in Urdu: the Persian, Arabic and English Strands

Language contact and the influence of one language on another are very common phenomena. Persian, Arabic and English have influenced various languages globally. Urdu is one of the recipient languages from these three sources, and shows linguistic features borrowed from them. This study focuses on the Persian, Arabic and English loanword morphology in Urdu.

Loanwords from Persian/Arabic are far older than English loans, and function like native Urdu words. Therefore, native Urdu morphological structures and those from Persian and Arabic are treated as the patterns for English loans. The discussion describes the patterns and then the processes involving English loans in the light of these patterns. The hypothesis is that the affixation, whether inflectional or derivational, may be based on native Urdu patterns but that the compounding of English loans is more frequent with Persian and Arabic loanwords. This is a major factor, which needs to be established. It is equally vital to know whether Urdu also borrows any derivations of an English loan, as it did with Persian/Arabic loans with or without any morphological changes.

Almost nothing is written on the morphology of loanwords, from the three languages, into Urdu. Furthermore, there is no theory on loanwords specifically dealing with the morphological adaptation of loans. So, the present work is descriptive and deals with the characteristics of the morphological structures from native Urdu, Persian, Arabic and English. Due to space restrictions, the primary focus is on gender/number and case morphology, and derivation of by affixation and by compounding.

The study is divided into six chapters. The discussion begins in the first chapter with an introduction to the study and an overview of the sociolinguistic background of Urdu. It also

discusses the influence of English loanwords on South Asian languages in general and Urdu in particular. The chapter exemplifies pluralisation of English loans and compound verb forms with the words of recipient languages. The next three chapters focus on inflectional morphology, derivation by affixation and derivation by compounding. In chapter 2, only the gender, number and case morphology of the Urdu noun with relevance to the three sources comes under discussion. Chapter 3 focuses mainly on derivational affixes in Urdu from the three sources. Although a discussion of morphological issues is the main concern, some phonological and semantic issues with relevance to morphology are also included. However, phonological issues are only discussed in connection with Arabic loanword phonology, which shares in lexical creations e.g. yonda 'scoundrel'. Derivation of new words is also very frequent by means of compounding. Constituents from two different sources very often interact and are rather more frequent than normal, i.e. native + native, combinations in Urdu. Therefore, rather than looking at their source languages, Chapter 4 focuses on various types of compounds i.e. endocentric, exocentric and copulative etc.

Chapter 5 discusses the features of English loanwords adaptation in the light of Persian and Arabic loanwords adaptations. Morphological changes occur both on the inflectional and derivational level. On the inflectional level, the changes are more frequent and based on the native Urdu patterns. Derivational changes are seen in various loans, but the adaptability is limited in the derivation of other categories irrespective of native Urdu or Persian and Arabic patterns. It is far less frequent than the adaptation seen in Persian and Arabic loans. It is more frequent with native Urdu affixes, but the formation of compounds is more frequent with Persian and Arabic loan constituents. Thus, the hypothesis made in the beginning of the study is supported. English loan affixes have not found a place in formal Urdu, although they are used informally. The chapter draws some conclusions. Chapter 6 then presents a summary of the discussion made in the thesis, and presents the implications of the study.

# Chapter 1

# Morphology and Loanword Morphology

#### 1.1. Introduction

Language contact and the influence of one language on the other are very common phenomena. Persian, Arabic and English are some of the great languages of the world. They have influenced various other languages globally. Although a great language in itself, Urdu is also one of the recipient languages. Lexical influence from various sources is a common occurrence, but it is significant to note that a recipient language shows the loan morphological and phonological features. The same can be seen in Urdu as well, but almost nothing has been studied on any of the linguistic aspects. A great many features are unexplored. This study attempts to bring some of these aspects to the surface. It focuses on the loanword morphology in Urdu and looks at loanwords from Persian, Arabic and English.

This study focuses equally on the morphology of loanwords from these languages. However, the loanwords from the former two are far older than English loans, and function like native Urdu words. They are far more morphologically integrated, which means that they have adopted various morphological changes. Therefore, in this perspective, the investigation into morphological adaptation of English loans is done by looking at the patterns of native Urdu morphological structures and the morphological structures of the older loans. Thus, the morphological structures of native Urdu, Persian and Arabic are treated as the model for the morphological adaptation of English loans, which is discussed in the final part of this study. As English loanwords are more recent, it is necessary to look briefly at some of the background of English loanwords in Urdu.

This is because the Urdu lexicon is basically an amalgamation of the native Urdu words and Persian and Arabic loans. Therefore, the morphological structures which are apparent as a whole are also an amalgamation of the morphological structures from these three sources. 'Native Urdu' is understood to mean a word belonging to Urdu, Hindi or even Sanskrit, provided that it is commonly used in Urdu. The reason for this is that linguists generally treat Urdu and Hindi as the same language, known by different names, i.e. Urdu, Hindi, Hindustani and Hindi-Urdu. According to Rai (2000: 11), "one man's Hindi is another man's Urdu". Linguists normally use the term Urdu-Hindi or Hindi-Urdu to study the common features despite the fact that some lexical, morphological and phonetic differences do exist. Sanskrit is considered to be the mother of both, but due to deep and distinct influences, Urdu is highly Persianised/Arabicised, while Hindi is highly Sanskritised.

Studying loanwords and loanword integration, the distinction and combination of two criteria of conformity, i.e. conformity of loanwords to the source language form and conformity to the target language system, help us better understand the diachronic processes of progressive integration of loanwords (Winter 2008: 156). By invoking conformity in both, one of two alternative possibilities must be addressed. (1) The properties of a loanword are unchanged in a borrowing language and they match those of the source language. (2) The properties of a loanword have changed, and they now match those of a target language system in that the loanword seems to be part of borrowing language vocabulary. Analysing conformity to the (SLF) Source Language Form and conformity to the (TLS) Target Language System, the main goal of this study is to establish first the relation between the Source Language (Persian and Arabic) and the Target Language (Native Urdu), with relevance to the morphology of loanwords. Then, treating the morphological structures of native Urdu, Persian and Arabic as the three faces of Urdu morphology as a whole, it is interesting to look at English loanword

morphology. The hypothesis is that the affixation, whether inflectional or derivational, may be on native Urdu patterns but that the compounding of English loans is more with the older loans (Persian and Arabic), which form a bigger part of the Urdu Lexicon. This is one of the factors which need to be established. Moreover, it is equally important to know whether Urdu also borrows any derivations, of an English loan, as it borrowed Persian and Arabic derivations with or without any morphological changes. The study generally observes the extent of the English loanword adaptation, in the light of Persian and Arabic loanword adaptation, and establishes if it is as complete as that of the two older loan influences. There are other related questions, e.g. what are the structural patterns after the loanword integration; how are the loanwords modified, and which of the elements are affected? It is also of interest to note whether Urdu borrows English affixes in the same way as it has borrowed Persian and Arabic affixes.

Two points in particular need to be considered. Regarding Persian and Arabic influence, there is extremely little written on Urdu linguistics in general and its morphology in particular. There is nothing written on English loanword influence on Urdu either. Therefore, the lack of any related literature is an obstacle. Moreover, there is no theory of loanwords dealing specifically with morphological adaptation of loans. In this context, the present work does not fit clearly into any specific theoretical framework. Therefore, there is need for good, careful descriptive work before serious theoretical analysis can be carried out. So, much of the current work is descriptive and deals with the characteristics of the morphological structures from the three prominent sources i.e. native Urdu, Persian and Arabic. Following this, the discussion of the morphological adaptation of English loans describes the processes involving English in the light of the three sources of morphological structures already shown.

Morphological adaptation includes all aspects of derivational and inflectional integration. However, due to space restrictions, the primary focus is on some major points, including gender/number and case morphology, the derivation of nouns, adjectives and verbs by affixation and by compounding. The proposed study is divided into six chapters. The examination starts in the first chapter with an introduction to English loanwords in Urdu, and more generally morphology and loanword morphology of the South Asian languages. A general picture is sketched regarding the status of English loanwords and how nouns and verbs in particular are adapted in some South Asian/Indo Aryan languages. In the subsequent chapters, the focus is on the morphology of the Urdu noun itself (Chapter 2), derivation by means of affixation (Chapter 3) and by compounding (chapter 4). Chapter 5 takes a close look at the adaptation of English loans. The final chapter (Chapter 6) concludes the discussion summarising Urdu morphological features and those of English loans. The key point in this chapter is to have a look at the general implications of the study.

In order to discuss these points, the first factor required is to understand morphology or word structure, in particular inflection and derivation. Beyond that, it is equally important to understand the issues involved in loanword morphology.

#### 1.2. Morphology

Morphology is the study of words and their internal structures mainly through the analysis of morphemes, affixation, reduplication (partial or full) and various sorts of compounding. Morphemes are various types of word parts, e.g. prefix, stem, infix and suffix. Thus, a single Urdu word, e.g. *laɪlmi* 'unawareness' can be broken into three morphemes, i.e. the Arabic prefix *la* 'un', Arabic stem *ılm* 'knowledge/awareness' and native Urdu suffix -*i*. The prefix *la* converts the noun *ılm* into an adjective *laɪlm* which is then converted into another noun *laɪlmi* 

by the suffix -i. Sometimes, independently occurring words in a compound can also participate in morphological processes. For example, the Urdu word kəmra-e-dʒəmaətə 'classroom' is a compound word and both the constituents kəmra 'room' and dʒəmaətə 'class' are two independently occurring words. The function of the infix -e- is to link both the constituents and give possessive meaning. It replaces the semantic genitive ka/ ki/ke, which cannot occur in compounds.

Many morphologists make a distinction between inflection and derivation, which are often associated with affixation (the attachment of a morpheme). Inflection does not change the category but rather the form of the same word, e.g.  $pl\omega t$  'plate',  $pl\omega t\tilde{\omega}$  'plates',  $pl\omega t\tilde{\omega}$  (plural oblique)  $pl\omega to$  (plural vocative). In the given Urdu examples,  $pl\omega t$  is the singular noun,  $-\tilde{\omega}$  is a plural marker in the second, while  $-\tilde{o}$  and -o show the oblique and vocative forms of the noun, respectively. Inflected forms are variants of one and the same word. Derivation, on the other hand, means new word formations from a base word. This may change the category of the base, but a derivation may also be in the same category as the base word. A noun, for instance  $\omega sul$  'principle', can be converted into the adjective  $\omega suli$  'in principle', the noun and adjective  $\omega suli$  '(someone) unprincipled' and an adverb  $\omega sulan$  'in principle'.

A loanword typically follows the morphological rules of a borrowing language. It may be helpful to know how languages influence each other and what morphological results occur.

#### 1.2.1. Loanword Morphology

When speakers of a language want to identify with each other, they may find themselves adjusting their speech to eliminate the more obvious differences in pronunciation or vocabulary. But when multilingual speakers interact with each other, it often involves

language contact. This can be seen in Kupwar, India, where multilingualism has led to the convergence between the local dialects of two Indic languages, Marathi and Urdu, and two Dravidian languages, Kannada and Telugu (Thomason 2001: 45). Kupwar is a small town of 3000 inhabitants who speak the four languages. Due to centuries of contact, the varieties of the four languages are similar to each other syntactically and are distinct from the varieties spoken in other parts of India. Gumperz and Wilson (1971) note that historically, grammar is thought to be the most persistent, while lexicon is the most changeable component of a linguistic system, but in Kupwar the grammar is adaptable and the lexicon is the most persistent. The words of each language have been retained and coupled with a common syntax. They observe that the adaptations are far-reaching and multilateral, and each language has adopted some features from the others.

Bilingualism and education are two important factors. Presently, electronic media is also an important factor. Note the following two examples. The first is Greece, where most people are educated and monolingual (Tatsioka 2008: 129), and the other is the Indian Subcontinent where most people are bilingual and uneducated. In both these regions, only a small number of people know English, but most people frequently use English words and phrases. Expressions like *oh God!* are very common. This is mainly due to media influence.

The consistent use of such borrowings makes them part of the borrowing language. However, loanwords and simple borrowings are two different terms which distinguish them from each other. Haugen (1950: 212) uses the term "borrowing" for *the attempted reproduction* of some patterns of one language into the other, and the term "loanwords" for *one type of borrowing*. Loanwords are single words or compounds, but borrowings may be stems or full phrases. Loanwords are also different from code-switching, as Poplack et al (1989: 390) state that the

former follows the rules of the recipient language, while code-switching is a linguistic term denoting the concurrent use of more than one language, or language variety, in conversation.

The study of how a language reacts to the presence of foreign words – whether it rejects, translates or freely accepts them – may give us insight into its formal tendencies. The adaptation of a loanword depends both on its word class (verb, noun, adjective) and on its similarity to native words and to loanwords from other languages already existing in the borrowing language.

The native Urdu affixes are productive and form new words in combination with the words from a different source. But some of the loan affixes e.g. Persian  $-ana \rightarrow zalmana$  'tyrant' also derive a good number of words with roots from other sources e.g. Arabic in this case. The productivity of these affixes in general, although systematic, shows some irregularity as well. Seidenberg and Gonnerman (2000: 354) state that, "Morphological systems are quasi regular. They are productive and systematic but also show many seemingly irregular forms." They note that the inconsistencies in word structures reflect the uncertainty about the nature of morphological units and greatly complicate the task of interpreting empirical results; for example, bakery is related to bake and cannery to can but what is groce in grocery? While looking at similar inconsistencies, Ahmed (2008: 2) states that Urdu "lexical items, prefixes, and suffixes are difficult to study on the spoken level since their occurrence is quite unpredictable." Although Urdu morphology follows some patterns and rules, the rules may be complex and may cut across linguistic components e.g. morphology and phonology. It cannot be termed 'unpredictable', as Ahmed claims. Nevertheless, there are certain deviations from the rules of each pattern.

The patterns of words used as nouns differ from those used as verbs or adjectives, in relation to both morphological marking and words' incorporation into the borrowing language structure. The incorporation of loans also indicates both alterations in the phonology of the loanwords and subsequent adjustments in the phonology of the recipient language. For example, with respect to regions, phonetically, there are two alternations in the English noun *street*. Punjabi people say *satrit*, but others pronounce it as *tstrit*. The loan also has two plural variants, *satrite/tstrite*. The subsequent adjustments in the phonology of the recipient language can be observed in Urdu also due to Arabic loan velar fricatives /x/ and /y/.

However, as the focus of this thesis is on the morphological integration of loanwords, rather than on loanword phonology, only some phonological aspects of relevance to morphology are discussed. Before discussing morphological adaptation, it is helpful to look at the sociolinguistic status of Urdu as compared with other languages of the Indian subcontinent.

# 1.3.1. Sociolinguistic Status of Urdu as Compared with other Languages

Urdu is an Indo-Aryan language of the Indo-Iranian branch from an Indo European family. Abbas (2002) traces its origin to the armies of Afghan emperor Mehmood Gaznavi in the 12<sup>th</sup> Century. However, as generally known, the term *Ordu* was first used for the language spoken by the soldiers of Mughal emperor Shahjehan (17<sup>th</sup> century) when he built the Red Fort in Delhi, and the surrounding town called *Ordu-e-Mu'alla*. Urdu abridged the several languages spoken by the soldiers in the Indian army and absorbed the loanwords from them. That's why it is often called *lafkari* (army) *zəban* (language) or the language of the army. There have been four major dialects of Urdu i.e. 1) Dakhani or now called Daccani 2) Pinjari 3) Rekhta and 4) Modern Vernacular Urdu.

The modern vernacular language is based on the *Khaqi Boli* dialect, of the Delhi region, spoken in 12<sup>th</sup> century. Urdu is most closely related to Assamese, Gujarati, Hindi, Marathi, Punjabi, Sindhi, Singhalese, and Romany. It is one of the largest languages of the Indian subcontinent with native speakers of over sixty million, and an official language in Pakistan and the Indian states of Uttar Pradesh, Kashmir and Delhi (Martindale (2009). It is one of the three most important languages in the modern Indian Subcontinent alongside Hindi and Bengali, while languages spoken throughout various regions are provincial languages. It is also widely spoken in Bangladesh, Afghanistan and Nepal and the Indian states of Bihar, Maharashtra, Andhra Pradesh and Karnataka. It is spoken as a medium of communication by a large number of communities in the Gulf, UK and many other countries of Europe, as well as Afghanistan, Thailand, Nepal, Turkey, USA, Australia, Fiji, South Africa, Zambia, Mauritius and Malawi (Junior, 2005). The eloquent Urdu poetry and Indian movies have given rapid popularity to this language. Then, the immigrant workers have spread it in all four corners of the world, and it is now one of the largest languages of the world. Rehman (2004) and Grimes (2000) consider Urdu-Hindi the second most spoken language of the world.

Urdu, Hindi, Hindustani and Hindi-Urdu are various terms used for the same languages. Linguists normally use the term Urdu-Hindi to study common features in both the languages. There are some morphological and phonological differences, but Urdu differs from Hindi mainly in its extensive Persian-Arabic borrowings along with Perso-Arabic script, while Hindi on the other hand is written in Devanagri script. Urdu has been a victim of cultural genocide at different periods of time (Azim (1975: 259) and was ruthlessly ignored by various governments. Perhaps, this is why Urdu borrows English loans more heavily than Hindi. A background of the lexical influence of English can help to understand loanword morphology.

# 1.3.2. Background of the Influence of English and the Status of the Loanwords

Lexical borrowing brings with it cultural implications, because it is not restricted to words alone, it includes cultural and social values (Hoffer 1980: 2). Borrowing normally involves conquest, cultural domination, disparity in social development, technological advancement and media etc. Lee (2004: 1) states, "Armed with technological, political, economic and cultural machines, the English language has made, and is making, a profound influence and impact on the enrichment or demise of world languages". All these factors have caused English to influence Urdu, which has borrowed a large number of its lexical items.

English is the language of science, technology, business and academic information. It is, therefore, globally thought to be a key to success. With the growth of English as an international language, there has been a rapid increase in the demand for teaching and learning English in Pakistan. It is the medium of instruction in all private schools and a compulsory subject from grade one in all public schools. It is also a popular subject of study in the universities. It is now an empowering language, and considered to have a high status and prestige as a foreign language. Rehman (998, 2005) considers, "English is a language of power and status symbol". The use of English loanwords in daily Urdu is not just a status symbol, it has become a social habit. Code-switching is the most frequent use of English loans in the media and by the elite. The reason for this is possibly the same as suggested by Takashi (1990) about Japanese that English borrowings are common because "they seem to convey a modernity and sophistication about the subject matter under discussion". The appearance of thousands of English loanwords is also due to new ideas and technologies. The borrowing of scientific and technical lexical items has increased the enrichment of Urdu also. The perspective in which these lexical items were borrowed can be understood by Baugh and Cable's (1978: 84) explanation of the power of English words that they were "so intimately associated with an object or a concept that the acceptance of the thing [involved] was the acceptance also of the word". Scientific and technical lexical items are just an aspect of English loanwords being used in Urdu. Baumgardner (1990: 60, 1998) discusses word formation in Pakistani variety of English. He hightlights some forms of English usage eg. compounding *half pants* 'shorts', affixation *de-notify* and conversion *affectee* 'someone affected by something' etc. However, the use of such words is restricted to English newspapers in Pakistan, as Baumgardner quotes.

In sum, like other world languages, Urdu is heavily influenced by English. It borrows a large number of English words, which is similar to the one noted by Carstensen (1986: 827) in some European languages. For these languages and for Urdu, English is a reservoir from which words can be taken at random. Loanwords related to almost every part of life are in common use, but crucially they are used under the morphological influence of Urdu. Morphological changes also occur both at the inflectional and derivational level. At the inflectional level, the changes are substantial and are based on the native Urdu patterns rather than those of Persian and Arabic loanwords. At the derivational level, there is an adaptation of a large number of loans, but the adaptability is limited in the derivation of words irrespective of the native Urdu or Persian and Arabic patterns. However, there is a difference of frequency in the derivation by affixation and by compounding. It seems more frequent by affixation when English loans correlate with native Urdu affixes. But the formation of compounds is more frequent with Persian and with Arabic loan constituents.

This is a general trend that is not just restricted to Urdu, but that is also noteworthy in many south Asian languages. A brief comparison of the morphological structures of some of these languages is given in the next section.

#### 1.4. English Loanword Morphology in South Asian Languages

This section presents common and contrasting features of English loanword morphology in some of the South Asian languages. However, due to space restriction, the aim is to show only a sketch of the loanwords. Therefore, the focus is on just two aspects, the pluralisation of nouns and hybrid compound verbs in which the constituents belong to different languages.

It seems very probable that the psychological attitude of the speakers of a borrowing language towards linguistic material has much to do with its receptivity to foreign words. Some languages require structural changes in the loanwords. Sapir (1921: 2) argues that the nature and extent of borrowing depends entirely on the historical facts of cultural relation. On this basis, he compares the structural changes and contrast between German and English in their loanwords from Latin and French. A similar contrast with regard to the treatment of foreign English material can be seen in South Asia, where the Indo Aryan languages sometimes allow loans to be only phonetically restructured. At other times, inflectional or derivational changes are made by affixation (or by compounding). One element specific to their word structure is gender/number marking, e.g. in Urdu /a/ (m), /i/ (f) and /e/ (pl), in verbs and adjectives, not only nouns. It is vital in loanword morphology as well. The following subsections show some patterns of English loan nouns in five languages: Urdu, Marwarı, Sindhi, Punjabi and Pushto.

# 1.4.1. Pluralisation of English Loan Nouns

Of the five sub continent languages compared in this section, the morphology of four, i.e. Urdu, Marwati, Sindhi and Punjabi is similar. Therefore, English loan nouns of one or two syllables may be phonetically different, but morphologically they are mostly the same as their original forms. For example, many loan nouns e.g. *plate*, *glass*, and *jug* are unchanged in Urdu, Punjabi, Sindhi, and Marwati. However, the morphology of Pushto differs from that of

the rest. It exhibits no additional variations. This difference is clearly visible also in the English loan nouns, e.g. *plett* is used with structural changes as *palæt*.

There are not many cases of three or more syllable loan nouns, as for example, *manager*. If and when such words are borrowed, they mostly have to be changed structurally. Table 1 below displays the comparison of the loan base forms in the five languages.

**Table. 1: English Root Nouns in South Asian Languages** 

English	Pushto	Punjabi	Sindhi	Marwarı	Urdu
1-pleit	palæt	plæt	plæt	plæt	plæt
2-glas	gılas	glas	glas	glas	glas
3-bildin	bilding	bildin	bildin	bildiŋ	bildin
4- bʊk	bʊk	bʊk	bʊk	bʊk	bʊk
5- məʃi:n	məʃi:n	məʃi:n	məʃi:n	məʃi:n	məʃi:n
6- kıtʃın	kıtʃi:n	kıt∫ın	kıtʃın	kɪtʃɪn	kɪtʃɪn
7- kəmı∫n	kəmæʃən	kəmıʃən	kəmıʃən	kəmıʃən	kəmıʃən
8- dʒəg	dʒəg	dʒəg	dʒəg	dʒəg	dʒəg
9- регрэ	pæpə-ər	pæpər	pæpər	pæpər	pæpər
10- hospitəl	əspə <u>t</u> al				

Note that Pushto shows both phonological and morphological changes, and so it is a good example of structural contrast with the other regional languages. For example, in pluralisation, it shows no gender marking and takes the same suffix *-una* in both masculine and feminine plurals. As the illustrations of this are given in the table below, Pushto is the only singled out here. In the other four languages, the pluralisation of the loan nouns is based on the alternation of vowels at the suffix position, and the loan stems are affixed with the native

plural morphemes. Therefore, no morphological difference remains between the English loan nouns and the native nouns.

Table 2a: The Masculine Plurals of English Loans in South Asian Languages

English	Pushto	Punjabi	Sindhi	Marwarı	Urdu
1. kɪtʃɪn	kɪtʃinoon <i>a</i>	kıt∫ın	kıt∫ın	kıt∫ın	kɪtʃɪn
2. kəmısn	kəmæ∫ən <u>u</u> n <i>a</i>	kəmıʃən	kəmıʃən	kəmıʃən	kəmıʃən
3. dʒəg	dʒəgun <i>a</i>	dʒəg	dʒəg	dʒəg	dʒəg
4. регрэ	pæpərun <i>a</i>	pæpər	pæpər	pæpər	pæpər
5. mænidʒər	menædʒərun <i>a</i>	mənidzər	mənedzər	menedʒər	menedʒər
6.am'mango'	<i>a</i> mun <i>a</i>	<i>a</i> m	<i>a</i> m	<i>a</i> m	am

Table 2b: The Feminine Plurals of English Loans in South Asian Languages

English	Pushto	Punjabi	Sindhi	Marwarı	Urdu
1-pleit	p <i>a</i> lætun <i>a</i>	plætã	plætũ	plætã	plætã
2-glas	gɪl <i>a</i> sun <i>a</i>	glasã	glas/glasũ	glasã	glasæ̃
3-bildin	bildinguna	bildiŋgã	bıldıŋgũ	bildiŋgã	bildingæ
4- bʊk	bʊkun <i>a</i>	bʊkã̃	bʊks/bʊkũ	bซk $ ilde{a}$	bʊkæ̃
5- məʃi:n	məʃinun <i>a</i>	mə∫inã	məʃɪnũ	məʃinã	məʃinæ̃
6. mæz 'table	mæzuna	mæzã	mæzũ	mæzã	mæzæ

The two tables exhibit some common and contrasting features in the plural morphology of some of the most commonly used English loan nouns in comparison with one native plural in the five languages. The nouns in (2a) are treated as masculine and those in (2b) are feminine. There may be morphological changes in masculine nouns as well, but the changes mainly occur in the feminine nouns. Pushto shows no gender distinction. In four of the languages, the pluralisation of the English loans is by suffixation, but the suffixation itself is no more than an alternation of vowels, which is the main element in their gender/number morphology. Phonological alternations come in many shapes and sizes, and the processes behind them are equally varied as are the kinds of factors which condition them (Davenport and Hannahs, 2005). Masculine loan nouns generally remain unchanged in their plurals, nevertheless a plural is sometimes formed by the alternation of vowels. For example, *deta* 'datum' which is treated as masculine singular is pluralised as *dete* in Urdu. But in Marwaṭī and Sindhi, it is taken as *deto* in its singular form and *deta* in its plural.

A feminine plural is based on the suffxal alternation of vowels with nasalisation. For example, a feminine nominative plural in Marwați and Punjabi is  $plæt\tilde{a}$ , but it is  $plæt\tilde{e}$  in Urdu and  $plæt\tilde{u}$  in Sindhi. Note that there is just an insertion of a nasalised vowel in the loan base noun. A feminine plural for oblique and vocative forms is suffixal with the insertion of nasalised vowel i.e.  $\tilde{o}$  and non nasalised vowel o respectively. Thus, the case morphology is also similar in these languages. Sindhi sometimes tends to use original plural patterns borrowed from English. Unlike the rest of the five languages, Pushto however discards gender distinction, and its pluralisation is also based on the consonant-vowel suffix *-una*, for both genders, rather than merely the alternation of vowels.

Generally, the pluralisation of a noun in the five languages is based on either alternation of vowels or suffixation. This is in sharp contrast to verb formation with English loans, which is done only by compounding.

# 1.4.2. Loan Verb Morphology

The verbs in these languages are very complex as compared to other syntactic categories. They inflect for tense, mood, gender and number. The verbal structure is in two forms. There are a large number of verbs already present in the basic lexical form. For example, a lexical verb pi 'drink' requires the suffix -na to form the infinitive pina 'to drink'. It takes other suffixes according to tense and aspect needs, e.g. pija 'drank' and piraha 'drinking'. In the absence of lexical verbs, nouns and adjectives (particularly loans) combine with some dummy verbs to form compound verbs. A dummy verb is a kind of tool for the verb formation in the absence of base verbs in south Asian languages. In Urdu, it is generally a redesigned form of a Persian auxiliary. Schmidt (1999: 101) explains that Urdu verbs demonstrate a very regular conjugation with the exception of five verbs ho 'be', kər 'do', de 'give', le 'take', and dza 'go'. These five verbs primarily function as main verbs and secondarily as dummy verbs (Versteegh 2001, 488) and light verbs (Butt, 1995) in complex predicates. The difference between a dummy verb and a light verb is not the issue here but, as concerns Urdu, it is explained in Chapter 4. For the present purpose, it is sufficient to note that a dummy verb is used as akind of tool to form a verbal compound in combination with a noun, e.g. N (sãs 'breath') + dV (le 'take') = V (sãs le 'breathe') or with an adjective, e.g. A (xof 'happy') + dV(ho 'be') =  $V(x \circ f)$  ho 'be happy'). It is needed when a lexical (base) verb does not exist. Versteegh (2001: 497) considers that the extensive use of dummy verbs in compound verbs is typical of Hindi-Urdu (Indo Aryan languages) generally.

The use of dummy verbs is particularly ubiquitous with loanwords for verb formation. Even if a loan verb enters Urdu, or other Indo Aryan languages in general, it ceases to function as a verb, and it has to combine with a dummy verb, which carries the inflectional information. The morphological functions of a dummy verb are the same. Not a loan lexical verb, but only a dummy verb shows any gender and number or tense and aspect changes. Thus, all the loans, whether they are nouns, adjectives or even verbs, must be integrated with dummy verbs to form compound verbs. For example, the English loan verb *provaid* 'provide' cannot function as a base verb in Urdu. Although borrowing verbs is frequent and convenient, the borrowed verbs cannot function as verbs unless they integrate with the dummy verbs. Therefore, to perform the function of a verb, the loan verb *provaid* is combined with a dummy verb *kərna* 'to do' (base form *kər*) to form a compound verb, i.e. *provaid kərna* 'to feel'. *provaid* is only used as a dysfunctional verb (used in the status of a noun). A loan verb does not function as a verb in Urdu. Its verbal capacity is taken by the dummy verbs probably due to the morphologically complex verbal systems of the Indo Aryan languages.

In light of this discussion, it can be claimed that verb construction with a loanword is in the form of a compound. More specifically, it is a hybrid compound. Various linguists have interpreted the term *hybrid* in different words. For example, Capuz (1997: 8) terms it *hybrid* or *loan blend*. The same terms are defined by Haugen (1950: 215) as those instances of lexical borrowing in which both 'importation' and 'substitution'" can be found. Kent (1999) interprets the term "hybrid" differently as a *pseudo loanword*. A hybrid combination of a native and a loan constituent is a general phenomenon in compounding. But as the Indo Aryan languages are heavily influenced by Persian-Arabic loanwords, it is also noteworthy that some dummy verbs themselves, e.g. *kərna*, are morphologically redesigned forms of Persian auxiliaries. It means that the verbal formations by them with English bases deviate from the

morphological rules of both the source language and the target language. The detailed discussion in Chapter 5 shows the situation in Urdu in this regard. Table 3 illustrates some compounds formed by dummy verbs in the five languages:

Table. 3: The Formation of Verbs with English Bases in South Asian Languages

<b>English Root</b>	Pushto	Punjabi	Sindhi	Marwarı	Urdu
1. provaid	kavəl	kərn <i>a</i>	kərə̃ţ	kərno	kərna
2. pruv	kavəl	kərn <i>a</i>	kərə̃ţ	kərno	kərna
3. əgri	kavəl	kərn <i>a</i>	kərə̃ţ	kərno	kərna
4. ərændʒ	kavəl	kərn <i>a</i>	kərə̃ţ	kərno	kərn <i>a</i>
5. eksept	kavəl	kərn <i>a</i>	kərə̃ţ	kərno	kərn <i>a</i>
6. kritisaize	kavəl	kərn <i>a</i>	kərə̃ţ	kərno	kərna
7. rısponsəbiləti:	hayəʃtəl	lễţã	wət <sup>h</sup> ə̃t	lẽçõ	len <i>a</i>
8. rispons	vərk <i>a</i> vəl	d̃ẽτã	dejõr	deçõ	den <i>a</i>
9. hæpi	ſo	hõţã	thijət	hõţõ	hona
10. æŋgri	ſo	hõţã	<u>t</u> hıjə̈r	hõţõ	hona

The patterns of verb formations are the same in all the five languages. The Indo Aryan languages require every English loan to be attached with a device for allowing verbal marking. This device is a dummy verb which forms a verbal structure. The dummy verbs, e.g. Urdu *kərna* 'do' (1-6), for transitive and *hona* 'to be' for intransitives and passive (9-10), construct infinitives e.g. *provaid kərna* 'to provide' (1) and *æŋgri hona* 'to be angry' (10). Similarly, for gender/number marking or tense/aspect morphology, it is only the dummy verb that undergoes morphological changes. The loan constituent, whatever the category, does not undergo any morphological changes. The only difference between the dummy verbs with regard to their sources is their (morpho) phonological structures. The dummy verbs used in

the four languages are similar to each other and take a nasalised syllable or an alveolar nasal + a vowel. On the other hand, Pushto differs again and mostly takes a lateral ending (1-8). There is no other significant difference, and therefore no further discussion is necessary at this point.

# 1.4.3. Loanword Morphology in South Asian Languages: A General View

The relevance of the foregoing discussion has been to show briefly two important aspects of loanword morphology in the five important Indo Aryan languages, i.e. the pluralisation of English loan nouns and the compound verbal structures formed by the English loanwords and the native dummy verbs. Generally (with the exception of Pushto), their gender/number morphology is based on the alternation of vowels. However, only the loan nouns treated as feminine are generally affected by the morphological changes. Their plural morphology is formed by the alternation of vowels. The vowels are nasalised. The nouns treated as masculine remain mostly unchanged. The gender marking of English loans is only seen in four of the languages; there is no gender distinction in loanwords in Pushto. Also, for pluralisation all loans undergo a change by suffixation and the loan nouns take the native plural marker -una e.g. glasuna 'glasses'.

Verb formation with English loanwords in all five languages is achieved by compounding with dummy verbs. Even if a loan is borrowed in its basic verb form, it ceases to function as a verb. There are no morphological changes in the English loan constituent, and the dummy verb constituent takes the verbal capacity and gives the inflectional information. The combination thus forms a hybrid compound, and the formation is an interesting phenomenon in itself. The formation thus deviates from source language and target language rules.

Almost all the dummy verbs in the five languages are similar in morphological structures, the only significant difference being that the dummy verbs of four of the languages are based

either on nasal or nasalised vowels, while those of Pushto are mainly based on the occurrence of a lateral. There are various dummy verbs for transitive and intransitive verb formations and differ from each other according to their semantic expressions.

# 1.5. Summary of the Chapter

This chapter has briefly introduced the discussion which will be gone into in greater depth in the following chapters. The study is generally concerned with the adaptation of loanwords from Persian, Arabic and English. The loanwords from the former two languages are fully integrated and have formed certain morphological patterns in Urdu. Thus various morphological patterns borrowed functions parallel to native Urdu morphological patterns. In this perspective, the study focuses on the English loanwords whether they follow native Urdu morphology or Persian and Arabic loanword morphology.

For this, Urdu morphology itself requires due attention, in order to contextualise the adaptation of English loans. This cannot be done without paying proper attention to the three major sources of words in Urdu, i.e. native Urdu, Persian and Arabic. However, no literature is available in this respect. Therefore, much of the present work is descriptive and deals with the characteristics of Urdu morphological structures. Moreover, there is no theory available which specifically focuses on the adaptation of loanwords. So, there is also need for good, careful descriptive work before serious theoretical analysis can be carried out in future works. Thus, the discussion in general is descriptive and brings out features of the adaptation of loans from Persian, Arabic and in particular English, which are more recent and very important.

Due to its more recent status and its importance, the chapter has also given a brief background of the status of English in Pakistan and a sketch of English loanword adaptation in five South

Asian languages. The structural patterns of English loanwords may be different in one way or the other, the loanwords are generally similar. However, the major point of debate to which we will be returning is the adaptation of English loans with relevance to Urdu morphology.

In order to understand English loanword adaptation, it is significant to have an idea of Urdu morphology, which is an amalgamation of native Urdu, Persian and Arabic morphological structures. Therefore, their features require attention in inflectional morphology and derivational morphology. The next three chapters focus on inflectional morphology, derivation by affixation and derivation by compounding. However, regarding inflectional morphology, only the morphology of the Urdu nouns is discussed, due to space restriction, as the English loan noun is the major component in morphological adaptation.

\*

# Chapter 2

# The Morphology of the Urdu Noun

#### 2.1. Introduction

As introduced in the last chapter, the morphology of the Urdu noun is of relevance to the three sources of vocabulary i.e. native Urdu, Persian and Arabic. Native Urdu nouns behave in one way, the nouns borrowed from Persian behave another way and the nouns borrowed from Arabic behave in yet another way. From this perspective, the first point to be noted in this chapter is gender marking. This is because the major points of focus are based on the gender distinction in Urdu. There is then 1) the plural morphology and 2) the morphology associated with case. Although the loans generally follow their source language forms, there may be native Urdu gender and number marking, or vice versa.

Native Urdu distinguishes two genders (masculine and feminine), two numbers (singular and plural) and three cases i.e. nominative (also called direct case), oblique, and vocative. Most nouns are marked for gender, number and case. There are a tiny number of unmarked nouns, which cannot be changed structurally e.g.  $ga\tilde{o}$  'village/villages' (m) and  $pa\tilde{o}$  'foot/feet' (m). Both are invariable and exhibit no morphological changes, although they are treated as masculine and can be used as singular and plural. With respect to morphological changes in the Urdu noun, Moizuddin (1989: 20) defines some criteria: 1) there are declinable and indeclinable nouns, which can and cannot be changed structurally; 2) many structural changes are correlated with gender and number agreement and 3) many other changes are based on a noun's relation to case markers. For example, the singular noun lapka 'boy' is pluralised as lapke 'boys', which is also used as the singular oblique form. The plural oblique form is  $lapk\tilde{o}$ . The oblique forms whether singular or plural are always followed by a case marker e.g.

ergative *ne* as larke *ne* (singular) or  $lark\tilde{o}$  *ne* (plural). These three types of inflections show various forms of morphological structures in Urdu nouns.

Hardie (2003: 35) states that, "Urdu inflection is based on suffixation; the suffixes are fusional, consisting overwhelmingly of a single syllable, or even a single vowel, that may mark multiple features e.g. gender, number and case are marked on nouns". The gender of a noun is generally indicated by the final vowel -a for masculine, which is the basic gender, and -i for feminine, which is derived by substituting -i for -a. They are the major gender markers, but there are some others as well:

Masculine Noun Feminine Noun

1a. beta 'son' beti 'daughter'

b. bhəngi 'toilet cleaner' bhəngən

c. ũt 'camel' (all other (m) suffixes) ũt**ni** 

As the examples show, there seems to be a correlation between both the genders. If a masculine noun ends in -a e.g. beta 'son' (1a), the feminine form is derived by substituting -i for -a e.g. beti 'daughter'. Most native masculine nouns end with -a, but certain others end with -i as in bhangi 'toilet cleaner' (1b), in which case the feminine is formed by substituting or affixing -an i.e. bhangan 'toilet cleaner'. Whether or not all other masculine nouns take any suffix, e.g. bhut 'ghost', the feminine forms are derived by the substitution or affixation of -ni i.e. bhutni. The feminine gender thus shows more morphological alternants.

The two gender markers are applicable to both animate and inanimate nouns. Most inanimate nouns take the same markers. For example, *sona* 'gold' is masculine and *tfandi* 'silver' is feminine. They are also often used to indicate the size i.e. masculine for bigger e.g. *tf* 'ura

'large knife' and feminine for smaller e.g.  $tf^huri$  'small knife'. Rizvi (2007: 73) states, "There is no general rule for gender classification for inanimate nouns. Usually huge, heavy, powerful, dominant and big objects are masculine, while small, weak and light are feminine". In this context, a large number of plants and inanimate objects are generally assigned grammatical gender and must be either masculine or feminine.

There are also deviations from expectations. For example, the feminine of the noun *hathi* 'elephant' should be \*hathən, but this is ungrammatical. The correct form is həthni. There are various similar examples of not only native Urdu but also of loan nouns. Therefore, the focus on gender and number marking is specially in connection with the claim made by Ijaz and Moin (2003: 56) that there is a well-defined set of rules for gender and number changes, which take place by a series of processes, at the morphological and phonological level. Although there are certain rules and general phenomena, evidence is also available against their claim. There are identifiable rules or patterns, but also exceptions. Each morphological pattern shows deviation from the rules.

The deviations are particularly visible in the pluralisation of the nouns. The changes in the native nouns are presented first, as this helps in understanding the loan noun morphology.

#### 2.2.1. Pluralisation of Native Nouns

The pluralisation of a noun is determined by the gender of that noun. If masculine and feminine nouns end in -a and -i, they take the plural markers -e and  $-ij\tilde{a}$  respectively. However, there are also other plural patterns correlated with gender, as given below:

# 2. Nouns and noun pluralisation

	Masculine	Masc. Plural	Feminine	Fem. Plural
a.	ləţk <b>a</b> 'boy'	lə <b>r</b> k <b>e</b>	ləţki 'girl'	ləţkɪj <i>ã</i>

b. bhəng**i** 'toilet cleaner' bhəng**i** bhəng**ən** bəngənẽ

c. mor 'peacock' mor mor $\mathbf{ni}$ j $\tilde{a}$ 

(2a) describes the most general patterns of masculine and feminine plurals with Urdu native nouns. However, a large number of masculine nouns, irrespective of endings e.g. *bhəŋgi* 'toilet cleaner/toilet cleaners' (2b) and *mor* 'peacock' (2c), remain unchanged in their plurals. This is due to the fact that *bhəŋgi* doesn't end in -a and *mor* has no vowel ending. Therefore, regular pluralisation of masculine nouns affects only those nouns ending in -a.

In contrast, feminine nouns take plural markers with respect to their endings. Many feminine nouns ending in -ni e.g. morni 'peahen' (2c) or həthni 'elephant' (f) also take the plural marker -jā i.e. mornijā/həthnijā, which is the same as in (2a). All other feminine nouns e.g. behn 'sister' take the plural marker -ē e.g. behnē 'sisters'. In general, the plural rule for both genders depends on the ending of a noun. The sketch of native plurals can be described as:

# Masculine Nouns Feminine Nouns

3a) 
$$-a$$
 (sg)  $\rightarrow -e$  (pl) c)  $-i$  (sg)  $\rightarrow -ij\tilde{a}$  (pl)

b) all other endings  $\rightarrow$  pl. unchanged. d)  $-ni \rightarrow -ij\tilde{a}$  (pl)

e) all other endings (sg)  $\rightarrow -\tilde{e}$  (pl)

In short, the plural form depends on the gender of the noun, as specific pluralisation strategies are associated with the gender of the noun. Kashif (2004: 91) notes some general features in the inanimate plurals, in connection with gender markers, all metals e.g. *sona* 'gold' and many food items e.g. *dehi* 'yogurt' are always treated as singular.

The gender and number phenomena described above show some regularity. There are also some deviations from the rules, which are discussed in the following subsection.

#### 2.2.2. Irregularities and Deviations in Native Urdu Nouns

Although there are not many examples, native Urdu nouns do show some deviation from each of the regular patterns particularly feminine forms i.e. -i, -ən and -ni respectively. As noted in the irregular feminine form həthni, the same occurs with some other feminine forms. For example, the feminine of masculine singular lohar 'blacksmith' should be \*loharni, but it is ungrammatical and the correct form is loharən.

Moreover, there are deviations from plural norms. Generally, masculine plural ends with -e and feminine plural with  $-ij\tilde{a}$  and  $-\tilde{e}$ . The examples in (4) illustrate deviation in gender number markingr:

Noun (m)	Plural	Feminine (f)	Plural
4a) bəkr <b>a</b> 'goat'	bəkre	bəkr <b>i</b>	bəkr <b>ıj</b> <i>ã</i>
b) kvtt <b>a</b> 'dog'	kʊ <b>tte</b>	kutti /kuttıja	kʊṭṭɪjã/ kʊṭṭɪjaē
c) tsira 'cock-sparrow'	tſīŗe	tsīri/tsīrja 'sparrow'	$\int \!$
d) dakija 'postman'	dakje	-	-

As shown in (4a), and noted above, the masculine -a and feminine -i are the major gender markers, while -e and  $-ij\tilde{a}$  form their plurals. But (4b) and (4c) show that there are sometimes deviations from the set patterns. Even if masculine ends in -a, its feminine marker can also be -ija rather than the regular marker -i. The feminine plural  $-ija\tilde{e}$  also takes the form accordingly, besides  $-ij\tilde{a}$ . Despite the different backgrounds of the two nouns i.e.  $k\tilde{e}_{i}tta$  comes from Praket and tfita comes from Sanskrit, their gender and number markers show the same variations. The changes in the feminine singular/plural marking are the same in both the nouns. The deviations -ija (singular) and  $-ija\tilde{e}$  (plural) do not however exist in large numbers. They are also used optionally.

The feminine singular marker -*ija* may be optional, but for some masculine nouns e.g. *daknja* 'postman' (4d), it is mandatory. This ending of the masculine singular is different from its primary form -*a*. It is possible in feminine nouns, but it is unusual in masculine nouns. The masculine noun does not have a feminine counterpart, which is even more unusual. This is not an exceptional case. Any other masculine nouns e.g. *tfərnja* 'mad' and *kalnja* 'someone blackish in complexion', with this suffix, do not have a feminine counterpart. However, whether masculine singular ends in -*a* or -*nja*, the plural marking of both is the substitution with -*e*. This again shows that there are some complexities, and a few masculine nouns ending with -*nja* may be termed lexical exceptions.

The variations in feminine nouns are generally greater than those in masculine nouns. A masculine noun takes the plural marker -e e.g. bete 'sons' only if it ends with the vowel -a. Otherwise, it remains unchanged (4a-4c). Although the masculine singular in (4d) is different from those of (4a-4c), it takes the plural marker -e. Therefore, this is unlike the feminine plural markers (4b & 4c) that change according to the changes of feminine gender markers. There are more feminine gender markers i.e. -i, -ija, -an and -ni, than the only masculine one i.e. -a. Feminine nouns also take two plural markers i.e.  $-ij\tilde{a}$  and  $-\tilde{e}$  as compared to masculine nouns, which usually remain unchanged except for taking -e.

In short, the gender and number morphology of native Urdu nouns frequently shows irregularities. Although there are rules, the simple gender or number relation is often difficult to define due to frequent lexical exceptions in the morphological formations. While there are certain rules, there are also important exceptions.

This is also evident in Persian and Arabic loan nouns. For example, in Persian, nouns generally show no gender, but its loans in Urdu are obligatorily marked for gender. Therefore, there is likely to be a mismatch between the behaviour of nouns in the two languages.

# 2.3. Gender and Number Morphology of Persian Loan Nouns

This section discusses the gender and number morphology of the Persian loans and their deviations from both the native nouns and the Persian loans. Persian does not have gender morphology, and so the Persian nouns are assigned gender for Urdu. In addition, most loan nouns take the native Urdu plural markers. Therefore, it is imperative to focus first on these two points so that deviation from native Urdu morphology can be understood. Then, it is also significant to note that Urdu borrows more plural markers from Persian than from Arabic, though generally most of them are of lower frequency. Only two of them i.e. -an and - at form a considerable number of plurals, which is discussed later in the section.

#### 2.3.1. Persian Loans with Native Urdu Gender and Number Markers

Persian does not show grammatical gender. Megerdoomian (2000: 4) states that there is no gender distinction in Persian, but there are several plural morphemes some of which are loans from Arabic. This means that a Persian loan noun must be assigned gender, which is typically based on an equivalent native Urdu noun or in some cases on an Arabic equivalent. For example, a loan e.g. ədakar 'actor' becomes ədakara(h) 'actress' (f) in Urdu, although the noun is the same for both genders in Persian. The feminine marker is in fact influenced by the Arabic feminine marker e.g. farra(h) 'poetess'. A brief sketch below illustrates the gender marking in Persian loans:

Masculine Noun

Feminine Noun

5a.  $bat \int a(h)$  'child' (m)

bətsi 'child' (f)

b. ədakar (generally rhotic ending)

ədakara(h) 'actress'

c. dərzi 'tailor'

dərzən

d. all other nouns e.g. fer

ſer**ni** 'lion'

In many loan nouns e.g. batfa(h) 'child' (5a), the final [-h] is deleted. Therefore, they are treated the same as the native masculine nouns ending with -a. Keeping the correspondence between masculine and feminine, all masculine nouns with a final -a have the feminine marker -i e.g. batfi (f) in Persian loans also. Thus, the gender morphology of the loans adopts the native patterns.

However, a problem arises when some loan nouns with the same suffix -a(h) e.g.  $\partial dakara(h)$  'actress' (5b) are treated as feminine. This is in fact due to the Arabic influence on both languages, because these nouns are treated as feminine in Arabic from which such loans in Urdu e.g. farra(h) 'poetess' are also treated as feminine. They may nevertheless be borrowed either directly or via Persian. Rizvi (2007: 73) points out that -a(h) is used for mostly feminine nouns in Urdu. This means that some Persian loans are also assigned a feminine gender ending in -a. Thus, semantics override what appears to be a grammatical marker. However, these feminine loan nouns are not numerous.

A similar contrast can be seen in the gender of masculine loan nouns, which end with -i e.g. dərzi 'tailor' (5c). These loans are feminised with -ən e.g. dərzən. Most other masculine nouns with no specific ending, generally -r e.g. fer 'lion', take the feminine marker -ni e.g. ferni 'lioness. Both the nouns in (5c & 5d) behave like native nouns, as in (1) seen above and show that the loans are fully integrated in Urdu morphology.

Most of the loan nouns appearing with the native plural markers show this general tendency. Detailed discussion of native plural markers has been presented in the previous sections, so no further discussion is necessary at this stage. There are however many Persian loan plural markers, to which we now turn.

# 2.3.2. Persian Loan Plural Markers

Although most Persian loans are completely integrated with native Urdu plural markers, the morphology of many others is not restricted to this. Many loan nouns, treated as masculine, take Persian plural markers, but those treated as feminine are almost all devoid of the loan plural marking. There are five Persian loan plural markers, all are used for masculine nouns. Two of the loan plural markers -an and -at are frequent. Three others i.e.-gan, -ha and -dʒat cannot be fused with many bases. They may attach to the native Urdu nouns e.g. barha 'many times' and even to Arabic loan nouns e.g. məsaladʒat 'spices'. However, they are not very productive. Table 2 illustrates some examples:

**Table 1 Plural Markers in Persian Loans** 

No.	Masculine Noun	Pl. Marker	Fem: Noun	Plural
1.	fərzənd 'son'	fərzənd <b>an</b>	doxtər 'daughter'	dʊxt̪əran
2.	məqal <b>a(h)</b> 'research paper'	məqal <b>a</b> ţ		
3.	selab zəda(h) 'sb flood affected'	selabzədgan		
4.	həzar 'a thousand'	həz <i>a</i> r <b>h</b> a		
5.	məsala 'spice'	məsala <b>dza</b> ţ		

Of the the first two major plural markers, -at may be the result of Arabic influence on both Persian and Urdu. It is a plural marker for the feminine nouns in Arabic, but it doesn't show any gender distinction in Persian. In Urdu, -at is a feminine plural marker for Arabic loans e.g. talibat 'students' and a masculine plural marker for Persian loans e.g. bayat 'gardens'.

In contrast to -at, -an is mainly used with masculine loans e.g. fərzəndan 'sons', but there are some examples of feminine loans as well e.g. doxtəran 'daughters'. In semantics, it is parallel to -ha in Persian; while -ha is the most productive and can be attached to most nouns, -an is generally used with animate nouns in formal language, but can also pluralise some inanimate nouns (Kahnemuyipour, 2000). For example, some plurals e.g. doxtəran 'daughters' and doxtərha are possible. In contrast, Urdu discards any difference, and -an is commonly used, as compared to -ha, which is restricted in Urdu plural marking. Therefore, the plurals like \*doxtərha are ungrammatical in Urdu, although they may be normal plurals in Persian.

However, -ha is distinctive in Urdu in the sense that it attaches to nouns and adjectives and its plurals are mainly used as adjectives and adverbs. For example, in the phrases, barha məvaqe 'many occasions' and həzarha vədzuhat 'thousands of reasons', -ha functions as the plural

marker of adjectives rather than those of nouns. Thus, -ha generally forms plural adjectives and adverbs by attaching itself to some native Urdu bases e.g. bar 'times'.

The examples of plural adjectives by -ha are a few, in spoken Urdu, and so are grouped with the less frequent Persian loan plural markers -ha, -gan and -dzat. Nonetheless, the latter two form plural nouns. The case of -gan is worthy to be noted, as its affixation causes phonological changes in the coda position of the bases. Any loan noun or adjective ending with the suffix -da(h), is often pluralised with -gan e.g.  $b \rightarrow n da(h)$  'servant of God'  $\rightarrow$ bəndəgan. After the affixation of -gan, the base bənda(h) changes into bəndə. Katamba (1993: 89) in discussing English divides affixes into two groups, neutral and non neutral. Neutral affixes have no phonological effect on the base to which they attach themselves e.g. home/homeless, in which -less is neutral, as it has no phonological effect on the base. In contrast, there are also some affixes that affect the base phonologically e.g. morpheme  $\rightarrow$ morphemic. The suffix -ic is non-neutral, because it provokes stress shift in the base: in morpheme the stress falls on the first syllable, in morphemic the stress falls on the second syllable. Therefore, -ic is termed non neutral affix. In this context, -gan is the only non neutral Persian loan suffix. The baseword is modified, and the suffix -a(h) is substituted by schwa -aalong with the affixation of -gan. This is however different from nouns such as farzand 'son' which end in dental and require only affixation of -an for pluralisation with no change in the basewords. This is also different from the nouns which end in -a(h) e.g.  $meva(h) \rightarrow meve$  'dry fruits' and generally have the native plural markers -e for masculine and  $-\tilde{e}$  for feminine. The difference of -gan in the two cases shows that it requires an ending of only -da(h) in the bases. Despite the changes, -gan forms a regular pattern of pluralisation.

As compared to -gan, the use of -dzat is even less frequent. Some loans ending with -a(h) take this plural marker, e.g. meva(h) 'dry fruits'  $\rightarrow mevadzat$ . However, as noted above -dzat is substituted by -at or native -e. Most loans generally have the native Urdu plural marker. Therefore, this is the rarest of the three less frequent loan plural markers even in written Urdu.

Both meva(h) and maqala(h) have the same endings and take the native plural marker. But the loan plural markers, which they take, are different i.e. mevadzat and maqalat. Similar usual plural pattern have been noted in the case of -ha, which attach itself only to adjectives to form plural adjectives rather than forming plural nouns.

This shows that there are frequent lexical exceptions in the loan plural patterns also. There are some deviations from Persian plural patterns and native plural patterns. This is discussed in detail in the following subsection:

#### 2.3.3. Deviation from the Native Urdu and the Persian Loan Patterns

As noted, Persian loan nouns adopt the native genders, but they take both the native and Persian plural forms. There is often deviation from the set patterns in both the processes. Some loan nouns generally end with -a(h) e.g. botfa(h) 'child' (m), and so with relevance to the native masculine marker -a, they are unarguably treated as masculine. However, some feminine loan nouns e.g. odakara(h) also end with -a(h). As noted above, Persian loans are also feminised following the Arabic patterns<sup>1</sup>. Thus, interestingly, the two Persian loans botfa(h) 'child' (m) and odakara(h) 'actress' (f) have the same origin but the gender markers are from two different sources i.e. native and Arabic.

\_

<sup>&</sup>lt;sup>1</sup>. The dictionary published by the Centre for Research in Urdu Language Processing, Islamabad. (CRULP)

The deviation in gender marking from both the native and the Persian rules also shows the deviation from native plural marking or vice versa. A loan feminine noun mostly takes the native plural marker  $-\tilde{e}$  e.g. yza (h) 'meal/food'/  $yza\tilde{e}$ , but this is not always correct. Although rare, some feminine loans take the Persian plural marker -an e.g. dextaran. Two major points should be noted. First, this is a deviation from the native pluralisation of the Persian loan nouns treated as feminine. The native plural \*dextare is ungrammatical, although  $-\tilde{e}$  attaches to most other feminine nouns. Secondly, as -an is mainly used as a masculine plural marker, its affixation with a feminine noun also shows deviation from usual plural marking. It is not an unusual phenomenon in many languages, but it is unusual in Urdu.

However the examples of -an attaching to both genders are only a few. In contrast, there are various examples of -at which are very often used for both the genders. For example, nəvadrat 'valuables' is masculine, but dəramədat 'imports' is feminine. It is in fact the result of the Arabic influence on both languages, due to which -at generally differentiates the gender between Persian and Arabic loans. It is a masculine plural marker for Persian loans and feminine plural marker for Arabic loans.

However, many other nouns show deviation from only the usual masculine plurals, as the ones ending with -a(h), e.g. meva(h), generally take the native plural marker -e, i.e. meve, but the original Persian plural marker -dzat, e.g. mevadzat is also used, although only sometimes. Sometimes, it alternates with the Arabic loan plural marker -at e.g. mevadzat or mevadzat (research papers'. This also shows variations.

pluralisation of an adjective is possible in Urdu, and with relevance to the ending, it takes the loan plural marker -gan i.e. selabzədgan 'flood affected (people)', although it is a hybrid adjective. The plural also represents both the genders. This is unlike native Urdu where an adjective whether singular or plural shows gender distinction. Thus, both the base adjective and the plural form show deviation from the rule. Generally, Persian loan nouns show various inflections often deviating from both the original Persian and the native Urdu. There is a case against Naim's (1999: 60) claim that Persian and Arabic plurals are never inflected in Urdu. This is to be discussed in the sections on Arabic loan morphology. All above examples show deviations from the rules of the source language and the target language.

In sum, most loan nouns follow the native Urdu morphology. Persian shows no gender distinction and the loans are treated as either masculine or feminine with the markers -a and -i respectively. However, some loan nouns, treated as feminine, also take the Arabic feminine marker -a(h), which is in harmony with the masculine gender marker after the deletion of aspiration. It is thus often used as a common gender marker for both masculine e.g. bətʃa(h) 'child' and feminine e.g. ədakara(h), and shows no morphological difference. Many loans treated as masculine remain unchanged in plural forms, the same as native Urdu plurals. Some loan nouns nevertheless do take Persian plural markers. Two loan plural markers i.e. -an and -at are important. Although -an is mainly related to the masculine gender, it attaches to certain feminine nouns as well. The other -at is more frequent for both the genders. Thus, used for both the genders, these two plural markers show that a large number of loans deviate from native Urdu morphological rules. This is in fact due to the influence of the Arabic loan nouns in both languages, e.g. the Arabic feminine marker on Persian loans in Urdu.

There are a large number of Arabic loan nouns, but their gender and number marking is not necessarily by native Urdu or Arabic loan suffixes. Certain Arabic loan plurals are based on vowel changes, thus called broken plurals. They often take Urdu or Arabic plural markers.

# 2.4. Morphological Changes in Arabic Loan Nouns

This section discusses the gender and number morphology of Arabic loans in Urdu and their deviation from both native Urdu and Arabic patterns. Although deviation from source language form is less relevant, it may be helpful to understand the morphological changes. Citing Wegener (2004), Winter (2008: 168) terms it peripheral target language form, if there is a deviation from the rules of source language and those of target language. The adaptation is particularly interesting when Arabic loan plural markers attach to Arabic loan plurals based on vowel changes. This shows that these adaptations are more complex than the ones seen in the Persian loans. All Arabic loan nouns are used with three possibilities: 1) most loan nouns adopt the native Urdu gender and number markers. 2) both the loan singular and plural forms are borrowed and integrated, and this often occurs 3) some of the loan plurals, based on vowel changes, are also affixed with native Urdu or even loan plural markers.

The Arabic treatment of masculine gender as the basic gender already exists in Urdu, and the feminine nouns are derived forms. However, the gender treatment of a noun is not necessarily the same. For example, ktab 'book' and korsi 'chair' are masculine and foms 'sun' and buhera(h) 'lake/sea' are feminine in Arabic, but the former loans are feminine and the latter ones are masculine in Urdu. The only Arabic loan gender marker is feminine -a(h) e.g. taliba 'female student'. Rizvi (2007: 73) also points out that -a(h) is mostly used for feminine nouns. It is however not used in its original form. In Arabic, as Ryding (2005: 22) points out, it is presumed that a feminine noun ends with an underlying /t/ e.g. qabiilah(t) 'tribe', but in Urdu,

there is no such presumption at all. Feminine loan nouns simply end with -a(h), with even the deletion of aspiration. It generally attaches to the masculine loans ending with nasal /m/ or rhotic /r/ e.g. masculine faur 'poet' becomes faura(h) 'poetess'. The nouns are similar to the native Urdu nouns ending with rhotic /m/, /r/, but they are feminised with -ni e.g. lohar 'blacksmith' becomes loharni. Although -a(h) attaches to Persian loans also, it is nevertheless limited to a few Persian and Arabic loan nouns, which are however frequent in spoken and written Urdu. Most singular loan nouns appear with the native gender markers.

However, considerable number of loan plurals are formed with the Arabic plural markers -in and -at. Linguists e.g. Hafez (1996: 393) and Ryding (2005: 84) describe three forms of Arabic plurals i.e. dual plurals, sound plurals and broken plurals. Hafez (1996) states, "there are two plural inflections. One of them is suffixal and regular, traditionally termed 'sound plural' with the two realizations -in for masculine and -at for feminine...The alternative is the 'broken plural' inflection, usually leading to insertion, deletion, and vowel change, and often an indication of maximal integration of a word into the language." There is no place of the Arabic dual plural in Urdu, but both the loan sound plural and the broken plurals are very common. Arabic terms of sound plural and broken plural are also irrelevant, and the use of these terms is only to differentiate the two loan plural forms. The loan sound plurals have entered Urdu with both the suffixes -in e.g. mahrin 'experts' (m) and -at e.g. talibat 'students' (f). However, for any specific nouns, only rarely are both the gender forms borrowed. It is either masculine e.g. mahrin or feminine e.g. talibat is in general usage. The latter are however more common than the former. Frequently, the two loan plural forms in parallel use with the native plural forms e.g. talibat 'students' (f) and talibae respectively.

Arabic plurals are also formed by vowel change e.g. *qəlb* 'heart' is pluralised as *qʊlub*, which is also used in the same way in Urdu. However, not necessarily all forms of a loan noun are used. Besides, a loan broken plural may also undergo a nativisation process and take the native Urdu or even Arabic plural markers. The following table illustrates the patterns:

Table 2. Arabic Loan Plurals Used in Urdu

No.	Masculine Nouns	Plural	Feminine Nouns	Plural
1	mahır 'expert'	mahır <b>in</b>	taliba(h) 'student'	<u>t</u> alıb <b>a</b> t
2	same 'listener'	same <b>in</b>	baqi 'remaining'	baqia <u>t</u>
3	mərz 'disease'	əmraz	qıst 'instalment'	∂qs <b>a</b> <u>t</u>
4	nəfil 'additional prayer'	nə <b>va</b> fil	xatun 'woman'	xəva <u>t</u> in
5	vəkil 'lawyer'	vokla	ərət 'woman'	ərə <u>t</u> ẽ
6	məqsəd 'aim/purpose'	тәq <b>а</b> ѕід	mədʒlis 'assembly'	mədzalıs
7	q∂lb 'heart'	q <b>v</b> l <b>u</b> b	sətər (*sətər)'line'	sə <u>t</u> ur

Among the seven loan forms, the first two represent the simple plurals formed by the affixation and are termed sound plurals, and the rest are various loan patterns of the so called broken plurals. Both the two Arabic loan plural markers -in (m) and -at (f) attach to the loanwords, which are basically adjectives but also used as nouns. The feminine plural marker is nevertheless more nativised and shows more morphological changes.

### 2.4.1. Nativisation of Arabic Sound Plurals

The feminine plurals are formed by simply affixing -at e.g. məsturat 'women', but they are not always so simple. Sometimes, a base noun ending with -jəṭ undergoes an alternation of vowels, and the schwa -ə- is substituted by the long -a- for the pluralisation:

6a. 
$$xasijat$$
 'quality'  $\rightarrow xasijat$ , (N)

b.  $kefija\underline{t}$  'state/condition'  $\rightarrow keifja\underline{t}$  (N)

c. ajat 'verse'  $\rightarrow ajat$  (N)

d) baqi 'remaining'  $\rightarrow baqijat$  'remains' (A)

The three feminine plurals (6a-6c) are as usual, although the singular forms e.g. xasijat 'quality' have endings other than the normal loan Arabic feminine marker -a(h). Saying \*xasijat is ungrammatical. Ryding (2005: 22) points out the concept of underlying /t/ in Arabic. There is nothing like this in Urdu. However, as the examples show, it is pronounced in some loanwords, and so it is in sharp contrast to the original forms in the source language. This marker -at derives feminine plurals from certain loan adjectives as well e.g. baqi 'remaining'  $\rightarrow baqijat$  'remains' (6d). It is not unusual but generally it is the function of the masculine plural marker -in. There are various examples of this, as shown below:

7a. same 'listener' (A)  $\rightarrow$  same**in** 'listeners' (N)

b. hazir 'present' (A)  $\rightarrow hazir$  'spectators/those who are present' (N)

c. nazir 'watcher/seer' (A/N)  $\rightarrow nazirin$  'audience' (N)

As examples (7a-7c) show, masculine plurals are generally formed by adjectives. An adjective being used as a noun is not native Urdu. It is only an Arabic loan form. However, it is only plural adjective being used as a noun. A singular adjective remains an adjective in Urdu, unlike that in Arabic. Moreover, plurals are only used as nouns rather than as adjectives. The process of pluralisation is simply by the suffixation of -in or -at. Some bases e.g. nazir 'watcher/seer' (7c) are mainly adjectives but may be used as nouns. The gender and number morphology of adjectives is a feature of Urdu, but gender distinction in a plural

adjective is not traditional. This is also in contrast to Persian loan adjectives which show no gender distinction.

Many of the loan sound plurals are parallel to their nativised counterparts. There may or may not be changes after nativisation, but changes are mostly in feminine plurals, as shown below:

<b>Masculine Loans</b>	Nativised Singular Lo	oan Plural Form	Nativised Plural
8a) talıb 'seeker/studer	nt' none	<u>t</u> alıb <b>an</b>	<u>t</u> əlb <i>a</i>
b) nasīr 'Publisher'	none	naʃır <b>in</b>	nasir (unchanged)
c) mahir 'experts'	none	m <i>a</i> hır <b>in</b>	mahır (unchanged)
Feminine Loans			
d) talıba(ht) 'student	' <u>t</u> alıba( <b>h</b> )	<u>t</u> alıb <b>a</b> t	<u>t</u> alıb <b>a</b> ẽ
e) naʃira( <b>ht</b> ) 'Publish	ner' $na \int ira(\mathbf{h})$	*naʃır <b>aţ</b>	na∫ır <b>aẽ</b>

The nativised forms of the masculine loans generally remain unchanged, but there is a possibility of morphological changes. The masculine plural *talıban* (8a) is unusual because the loan plural marker is generally -*in*. The loan may have entered Urdu via Persian, as -*an* is a Persian plural marker used for the loans treated as masculine. The nativised plural *təlba*(*h*) (8a) is also in sharp contrast to the Urdu pattern, which is generally unchanged (8b & 8c). This shows its deviation from both the source language and target language forms.

The nativised forms of feminine loan nouns e.g. nafira(h) (8e) often undergo morphological changes, but there may be certain deviations from the normal changes. Its nativised plural e.g.  $nafira\tilde{e}$  is possible, although a female publisher is untraditional. Its loan plural \*nafirate\* is

ungrammatical, but another feminine plural *talıbat* 'female student' in (8d) is grammatical. This also shows that there are certain lexical exceptions in the nativisation.

In sum, most Arabic loan nouns must adopt native plural forms. The nativisation may however vary depending on the borrowed patterns. The two plural markers i.e. -*in* and -*at* may attach to a large number of adjectives, but it is mainly the masculine plural marker that does the job. The nativised masculine plurals generally remain unchanged and most changes take place in the feminine nativised plurals, which are preferred to the loan ones.

The two loan plural markers are just one aspect of Arabic loan pluralisation, and they are in limited number. However, they are in daily usage. A large number of loan plurals are based on vowel changes. They are formed the same way as in Arabic depending on the modification of base. However, they are simply 'borrowed' forms. Many masculine loan nouns adopt this plural pattern, whereas feminine nouns take native plural markers.

## 2.4.2. Nativisation of Arabic Broken Plurals

Arabic plurals based on vowel changes are termed broken plurals. They are the most challenging plurals, as they are complicated and follow various patterns. They primarily involve internal modification of the base by vowel change e.g. nofs 'soul'  $\rightarrow nofus$ . Booij (2005) terms the modification of base *revowelling*. He states that the inflection in broken plural involves vowel change. Radcliffe (1990: 113) notes 27 patterns of broken plurals. Urdu has borrowed hardly some patterns. Although used the same as the original forms in the source language, broken plurals are merely borrowed forms in Urdu. Therefore, the complexity of their structure is irrelevant. Without any discussion, Moizuddin (1989: 40) terms them "examples of transplanted Arabic morphology". This pluralisation is in sharp

contrast to the native Urdu pluralisation, also used parallel to it. For example, the masculine *nafs* 'self conscious' remains unchanged in the native form and has a parallel use.

These plurals are examples of synchronic borrowing, and are used widely, but many of them have diachronically adapted themselves. An interesting element of the adaptation is that some of the non Arabic loans e.g. Turkish *xəvaţin* 'women' also adopt the broken plural forms. In addition, although rare, some loan nouns are normally affixed with plural markers in Arabic, but take the broken plural forms in Urdu. This is a unique feature in the whole pluralisation process, and therefore it is the focus of this section. Abdallah (2006) points out various patterns of broken plurals in Arabic from which Urdu has borrowed many forms. The following table illustrates some major patterns and their nativised Urdu counterparts:

Table 3: Major Patterns of the Loan Broken Plurals and their Native Counterparts

MasculineNoun	Loan/Nat. Pl	Feminine Noun	Loan/Native: Pl
1.dʒohər 'quality'	dʒə <b>v</b> ahır/dʒohər	xatun 'woman'	xəvatin / *xatune
2. hakım 'rulers'	hʊkam/ hakım		
3. mərz 'disease'	əmraz / mərz	yərz 'interest'	əγraz / γərz <b>ẽ</b>
4. molk 'country'	məm <b>a</b> lık / mʊlk	mənzıl'destination'	mən <b>a</b> zıl/ mənzıl <b>ẽ</b>
5. ʃək 'doubt'	∫ʊkuk. / ʃək	rəsəm 'tradition'	i)rəsum (ii)rəsəm <b>ẽ</b> (iii) rəsum <b>at</b>
6. vəkil 'lawyer'	v <b>ʊ</b> kl <i>a</i> / vəkil		

The table shows the six major patterns. For each noun given, there are at least two plurals. The one on the left is the loan broken plural, while the one on the right is the nativised form. Firstly, an interesting element to see is the Turkish feminine noun *xatun* 'woman' (1) that only adopts the broken plural form *xavatin* rather than the native form. The other loans have nativised counterparts as well, but they have not integrated with the native Urdu plural marker

-ē. Therefore, the nativised plural \*xatunē is ungrammatical. This shows the possibility that some non Arabic loanwords may also be pluralised the same as Arabic loan broken plurals, whether or not they take nativised plural forms. Moreover, though the Arabic singular counterpart ərət 'woman' is more common, it only takes the native plural marker i.e. ərətē rather than forming the broken plural \*ərat, which is ungrammatical. It is an interesting element that Urdu adopts a non Arabic noun in broken plural form rather than plural noun from the original source. Their semantics is the only difference between the two singular nouns, and the Turkish xatun expresses more formality and respect. Arabic pluralisation with a Turkish loan noun may be because Arabic has also influenced Turkish.

Evidence is also available from the Persian masculine loan noun dzohar 'quality/precious stone'  $\rightarrow dzavahir$  (1). Its native plural form remains unchanged. Both loan plurals are influenced by Arabic, so their formation may have occurred in their respective languages. However, the two are identical in the plural formation and have the common infix -ava- in combination with quite different bases coming from different sources. The same infix -ava- in both non Arabic loans suggests that their formation occurred within Urdu. Nevertheless, they are not a native pattern, but signal the nativisation of broken plurals as a pattern emerging.

This is further witnessed in *hakam* 'ruler'  $\rightarrow hv kam$  (2) in the table. It shows a broken plural, which it is not used in Arabic. Besides Turkish and Persian examples, this is concrete evidence that there is a formation of broken plurals within Urdu, although it is Arabic feature.

In sum, the patterns of the so called Arabic broken plurals are complex and the most challenging types of loans. However, some bases of sound plurals and non Arabic nouns show that broken plural as a plural form may be surfacing in Urdu. Two different bases taking the same infix -əva- also show the complexity. This is further explicitly seen in some loan broken

plurals taking Arabic plural markers i.e. -*in* or -*at*. Thus, sometimes there may be three plurals i.e. the loan broken plural, the native plural, and the plural deviating from both forms.

## 2.4.3. Affixation of Arabic Plural Markers with Loan Broken Plurals

Although broken plurals are frequent, the speakers generally use an adapted plural. The loanword integration with frequent deviation from the rules is interesting. Most importantly the loan broken plurals take the loan plural markers -in or  $-a\underline{t}$ , and so they are generally used as sound plurals e.g.  $v \rightarrow d a a b$  'reason'  $\rightarrow v \rightarrow d a a b$ . This illustrates two points: 1) while functioning like native affixes, -in or  $-a\underline{t}$  adopt the native status. 2) the loan broken plurals adapt into the peripheral nativised plurals, because -in and  $-a\underline{t}$  are not the original native plural markers. The two tables give examples of the phenomenon:

Table 4a. Feminine Nouns

Noun	Broken Pl.	Peripheral Nativised Pl.	Native Plural	
1. rəqəm 'amount'	rəq <b>u</b> m	rəq <b>u</b> m <b>a</b> <u>t</u>	rəqəm <b>ẽ</b>	
2. rəsəm 'custom'	rəs <b>u</b> m	rəs <b>u</b> m <b>a</b> ţ	rəsəm <b>ẽ</b>	
3. vədʒa(h) 'reason'	vədz <b>u</b> h	vədʒ <b>u</b> h <b>a<u>t</u></b>	vədʒa(h) <b>ẽ</b>	
4.təqrib 'ceremony'	təqarib	təqrıb <b>a</b> t	<u>t</u> əqrıb <b>ẽ</b>	

**Table 4b. Masculine Nouns** 

Noun	Broken Pl.	Peripheral Nativised Pl.	Native Plural
1. rʊkən 'member'	ərkan	ərakin	rʊkən
2. məʃrɪq 'east'	mə∫ <b>a</b> rıq	məʃrɪq <b>en</b>	məʃrɪq
3. dʒohər 'ruby' (Persian	dʒə <b>va</b> hır	dʒə <b>va</b> hır <b>a<u>t</u></b>	dʒohər
4. ləqəb	əlqab	əlqab <b>at</b>	ləqəb
5. lazım (adj) 'necessary'	ləvazım	l <b>əv</b> azım <b>a</b> ţ	-

There are loan broken plurals, the native Urdu plurals and the peripheral nativised plurals. The Persian masculine loan *d3ohor* 'quality/ruby' (3) adopts a broken plural form *d3ovahur*, which further takes the plural marker -at i.e. *d3ovahurat* (3). The plural marker -at attaches to feminine loans from Arabic, but attaches to masculine loans from Persian. The affixation of -at with the broken plural form certainly shows the nativisation of the two loan plural markers. This sort of pluralisation seems frequent both in masculine (3,4,5) and feminine nouns (1,2,3). Many adjectives, e.g. (5) in (4b), also undergo a similar adaptation. Thus, the rule deviation is not just seen in the recipient language but also in the donor language.

Some affixations e.g. təqribat 'ceremonies' (f) (4) and məfriqen 'east and west' (m) (2) occur with singular nouns rather than broken plurals. They are thus seen in two directions, one with the broken plurals and the other with their bases. Both are unusual in the source language and in the target language. This means that there are certainly complexities in the adaptation process and deviation from the set patterns which need to be discussed in detail.

## 2.5. Deviation from Morphological Patterns in General

In addition to the broken plurals of non Arabic loan nouns, a few distinctive factors have been noted. Some loans normally take plural markers in Arabic but they may adopt broken plural forms in Urdu, although this is rare. Moreover, Arabic loan plural markers -in and -at attach to the loan broken plurals. In some cases, they attach to their singular bases instead. Further, some loan adjectives taking broken plural forms also undergo the affixation. Thus, though Arabic loans generally follow the rules of the two languages, they also often deviate from both. Consider the three loans and their plurals:

The first two (9a & 9b) are masculine nouns and the third (9c) is an adjective, but their plurals are also treated as masculine plurals. Each of the three examples has at least two plurals of which the first is the Arabic loan and the second (and the third) is the adapted form. Although all singular forms are identical, morphologically, their plurals are different and deviate from both the language patterns. The plurals (i) and (iii) follow Arabic and native Urdu patterns respectively, but (ii) deviates from both. In (9a-i), there is normal Arabic loan sound plural, while (9a-ii) shows a broken plural form, not used in Arabic. In (9b), the pattern of pluralisation is a reversed from the one in (9a). The normal Arabic plural is broken plural (9b-ii) and the peripheral nativised plural is sound plural (9b-ii). The adjective in (9c-ii) shows that its broken plural itself takes a plural marker. The newly formed plural adjective *lovazmat* 'essentials' thus shows a deviation from Arabic rules while at the same time it does not follow Urdu rules for pluralisation. There are two contrasting elements. Unlike Persian adjectives, it takes gender and number marking, which is a feature of Urdu. Like Persian masculine plurals, it takes -at for pluralisation. All the integrations of loans show a variety of changes and deviations from the rules.

The patterns of change are evidence that there are certain rules and patterns but there is also deviation seen in almost every pattern. The examples also illustrate that a loan noun (or adjective) may have more than one plural. The adaptation may be partial and full. For example, *hokam* (9a-ii) and *xadmin* (9b-ii) are partially adapted, because the integration is not exactly according to the native pattern. The examples above show the three forms of morphological integration, as defined by Winter (2008: 158). The first plural in each is according to the non native pattern. The second is peripherally integrated plural, and the third

follows a target language pattern. Therefore, it can be claimed that Urdu morphology is not restricted to native Urdu patterns. Rather, there are elements of Persian and Arabic morphology as well. There is often amalgamation of the three morphological structures.

Sometimes, the amalgamation is to the extent that it is difficult to decide between the native and the loans. This is because many Arabic loan nouns have entered Urdu via Persian, and undergo significant morphological changes. These changes are not restricted to either of the two origins. The root word *xəbər* 'news' for example, is an Arabic loan and entered Urdu via Persian. The morphological patterns are different from those of the origins. This is compared in the modified examples from Riaz, (2007: 5):

10a. xəbər 'news' (Urdu, Persian and Arabic root form)

b. əxbar 'newspaper' (Urdu and Persian, but broken plural of 'news' in Arabic)

c. əxbarat 'newspapers' (Urdu and Persian)

d. xəbr**e** 'news' (plural) (native Urdu)

The tri-consonantal Arabic loan base entered Urdu via Persian and undergoes various morphological changes in Persian and then in Urdu. The Arabic broken plural  $\partial x bar$  is used as a singular noun with different semantics, 'newspaper' in Urdu and Persian. After the affixation of  $\partial x dat$ , its plural  $\partial x \partial x dat$  is also common in both languages, but it is alien in the Arabic lexicon. The native Urdu plural  $\partial x \partial x dat$  is strange in both Arabic and Persian. All the changes make it difficult to differentiate between the origins of the forms.

The morphological changes in the Arabic loans, in short, show a variety of features. Urdu has borrowed three major points i.e. the feminine gender marker -a(h), the sound plural markers -in (m) and  $-a\underline{t}$  (f) and various patterns of broken plurals. The deletion of the final [-h] leaves

the feminine marker -a, which is homophonous with the native masculine marker -a. However, it is limited to very few Arabic and Persian loans. The loan nouns are pluralised in the Arabic manner, but often take the native plural markers, which is the nativisation of the loans. In the nativisation process, there are two important points. Some non Arabic loans and the bases of sound plurals adopt the broken plural forms. Moreover, some Arabic loan broken plurals undergo further morphological changes and are also affixed with the Arabic sound plural markers -in and -at.

All of these changes show that the loans often deviate from the native rules. However, whatever the source of the nouns, the morphological changes according to case mainly follow the native Urdu rules, although there is a possibility that a loan noun e.g. axbarage 'newspapers' may remain mostly unchanged in various case forms.

## 2.6. Morphological Changes of Nouns Correlated with Case

A grammatical word form is a form of word that may express a semantic relationship with the other constituents of a sentence. Thus, Urdu word forms a 'come', ana 'to come' and aja 'came' are all grammatical forms of the base verb a 'come', as they express semantic relationships with the other constituents (e.g. subject) in a sentence. Aronoff (1996) explains that a grammatical word is a lexeme in a particular syntactic context, where it is provided with morpho-syntactic features (like case and number). One of the word forms is the form of a noun based on case. This section discusses the singular and plural morphology of the Urdu noun in relation to case. In this regard, the morphological structures with relevance to the three cases i.e. nominative, oblique and vocative are the major focus. McGregor (1972: 1) treats vocative as a special form of the oblique case, while in his division of the Urdu nouns, Moizuddin (1989: 31) terms them the three variants.

The Urdu nominative (also called direct case) is phonologically null, as it does not bear any case marker. The oblique and vocative forms are always followed by various case markers. The two forms of a singular masculine noun are identical to nominative plural, but are different if a noun is plural. The oblique plural ends with a nasalized vowel, while vocative plural is not nasalized. There is no other difference between the two. In fact the vocative form is rare and used only in poetry.

A noun is incomprehensible unless the gender and number morphemes -a/i/e are affixed to it. Consider the stem form lark 'boy/girl' that cannot be understood without gender and number markers. The difference between the three forms of the masculine noun larka 'boy' is shown by Mohanan (1994: 80) in her table:

Table 5: Case forms of native masculine noun (ending in -a)

Function	Singular	Plural
Nominative	ləţk <b>a</b> 'boy'	ləţke 'boys'
Oblique	ləţke 'boy'	ləţk <b>õ</b> 'boys'
Vocative	ləţke 'boy'	ləţk <b>o</b> 'boy <b>s</b> '

(Source. T. Mohanan)

Table 5 summarises the structural changes of a noun in relation with a case. The singular oblique and the vocative forms i.e. *logke* are identical to its plural nominative form *logke*. The stem forms of masculine nouns ending in -a are always inflected when used as non nominative arguments. This is also shown in the oblique and vocative forms of the plural. The difference between both is that of nasalization. If a masculine noun ends in something other than -a, no changes occur in the singular nouns with relevance to case:

Table 6: Case forms of native masculine noun (all other endings)

Function	Singular	Plural
Nominative	nokər 'servant'	nokər 'servants'
Oblique	nokər 'servant'	nokər <b>õ</b> 'servant <b>s</b> '
Vocative	nokər 'servant'	nokəro 'servants'

The changes in Table 6 show some differences from those seen in the previous table. The singular nominative, oblique and vocative forms are identical, if a masculine noun ends in other than -a, e.g.  $nok \sigma r$  'servant'. The nominative plural also remains unchanged, and so it is identical to the three singular forms. However, the oblique and the vocative plurals end in  $-\tilde{o}$  and -o respectively. They are identical to the ones seen in Table 5.

The three singular forms of a feminine noun are also the same. The structural changes of a feminine noun are usually more interesting and different from those of masculine nouns. Consider the feminine noun larki 'girl':

Table 7: Case forms of native feminine noun (ending in -i)

Function	Singular	Plural
Nominative	ləţki 'girl'	ləţkıjã 'girl <b>s</b> '
Oblique	ləţki 'girl'	lərkıjõ 'girl <b>s</b> '
Vocative	ləţki 'girl'	ləţkıjo 'girl <b>s</b> '

Compare Table 7 with Table 5, as they show the changes in the masculine noun  $l \partial r k a$  and feminine  $l \partial r k i$  respectively. The feminine singular, in sharp contrast to the masculine singular, remains unchanged in its nominative, oblique and vocative forms, which are the same as those of the other noun  $nok \partial r$  in Table 6. Moreover, the three forms of the plural nouns are also notable. The nominative plural is different from those of the masculine nouns in the two previous tables but interestingly the oblique and the vocative plurals are the same i.e.  $-\tilde{o}$  and

-o. The two case forms remain the same with all nouns, which can also be witnessed in Table 8, which shows the case forms of a feminine noun with zero marking:

Table 8: Case forms of native feminine noun (all other endings)

Function	Singular	Plural
Nominative	behn 'sister'	behne 'sisters'
Oblique	behn 'sister'	behnő 'sisters'
Vocative	behn 'sister'	behno 'sisters'

The changes in Table 8 represent all other native Urdu feminine nouns with an ending different from -i (or -ni). The feminine singular noun remains unchanged in the three case forms. However, the plural nominative, oblique and vocative forms show different endings. The nominative plurals end with  $-\tilde{e}$ , while the oblique and vocative plurals are the same as those of all other nouns, noted in the three tables, i.e.  $-\tilde{o}$  and -o respectively. Unlike masculine nominative plurals, which are identical to the masculine oblique and vocative singulars in Tables 5 & 6, the feminine nominative plurals in Tables 7 & 8 are different from the feminine oblique and vocative singulars. This shows that the feminine plural reflects case differences but there is no structural change in the feminine singular noun based on case.

The case morphology generally shows gender distinction. Masculine nominative plural is the same as the masculine oblique and vocative singular whatever the ending is. However, the oblique and vocative forms of masculine plurals are different from each other and from those of masculine singular nouns. The case morphology of feminine singular nouns remains unchanged irrespective of any ending. However, the nominative form of feminine plurals is different from the oblique and vocative singular nouns. The oblique and vocative forms of the feminine plurals are also different from those of singular nouns.

The case morphology of both genders remains unchanged in singular forms, except that a masculine noun ends in -a. Moreover, if masculine and feminine nouns end in -a and -i, there is a morphological difference between the nominative plurals of and feminine nouns i.e. lagke 'boys' and lagkaja 'girls' respectively. But there is no difference in their oblique and vocative plurals, which end in  $-\tilde{o}$  and -o respectively. The two latter forms are the same in all nouns.

The case morphology of a loan noun generally follows the native morphology, but it may also adopt a case without any major changes regarding gender distinction, as noted below:

Table 9: Case forms with a Persian masculine noun

Function	Singular	Plural	
Nominative	mehman 'guest'	mehman <b>an</b>	
Oblique	mehman 'guest'	mehman <b>an</b>	
Vocative	mehman 'guest'	mehman <b>an</b>	

Table 10: Case forms with an Arabic feminine noun

Table 10: Case forms with an Arabic feminine noun				
Function	Singular	Plural		
Nominative	sətər 'line'	sətur 'lines'		
Oblique	sətər 'line'	sətur 'lines'		
Vocative	sətər 'line'	sətur 'lines'		

It is noteworthy that there is no gender distinction, and the case morphology remains in a similar pattern. The distinction of the Persian masculine noun *mehman* 'guest' and the Arabic feminine noun *sətər* 'line' does not make any difference, and singular and plural loan nouns undergo no structural changes in the three case forms, if they do not adopt native morphology.

In short, the case morphology of a singular noun is generally the same for the three cases except that a masculine noun ends in -a. The oblique and vocative plurals are also the same, in

both the genders, with the difference of nasalization. However, the nominative plurals have differences depending on the endings and on the gender. Gender distinction again disappears in a loan noun. All three forms of case are the same for singular and plural nouns respectively. The loan forms may also be used, but the case morphology is mainly based on native patterns.

## 2.7. Summary of the Chapter

This chapter has discussed the gender and number morphology of the Urdu nouns with respect to the three sources i.e. native Urdu, Persian and Arabic. The gender markers of the native Urdu nouns are generally -a and -i. Masculine is the basic gender, and feminine gender is derived with relevance to the masculine noun. If a masculine noun ends in -a, the feminine form ends in -i. The masculine nouns other than -a are not many but they do exist. If a masculine noun ends in -i, the feminine form ends in -a. Any other masculine ending takes the feminine form -ni. A native masculine noun ending in -a takes the native plural marker -e. All other masculine nouns are unchanged in plurals irrespective of their endings. All feminine nouns ending in -i are affixed with the plural marker  $-j\tilde{a}$ , and -i is substituted with the short vowel -i. All other feminine nouns take the plural marker  $-\tilde{e}$ .

The Persian loan nouns have no gender marker of their origin. They are treated as masculine in most cases. However, the gender morphology of some feminine nouns is influenced by the Arabic feminine marker -a(h). Persian loans take various plural markers i.e. -an, -at, -gan, -ha, -dzat, but only the first two are frequently used with masculine nouns. The loan nouns take the plural patterns of their origin, but they are mainly adapted in the native Urdu morphology. For example, the masculine nouns ending in other than -a e.g. adakar 'actor/actors' do not bring plural inflection at all. This could be attributed to the conformity to

the plural system of the target language. Moreover, some Persian loans e.g. ədakara(h) 'actress' (f) show the Arabic gender, but they form their plural in native manner ədakaraē.

The entrance of Arabic feminine marker -a(h) in Urdu is due to some Arabic feminine loan nouns. However, gender treatment of most Arabic loans follows native rules. It is also noteworthy that many nouns e.g.  $k\sigma rsi$  'chair' are masculine in Arabic, but they are feminine in Urdu. The loans have brought only the feminine gender marker -a(h), but two Arabic loan plurals i.e. the sound plurals ending in -in and -at and the broken plurals based on the alternation of vowels. Although they are no more than borrowed forms in Urdu, some base nouns of the sound plurals and some non Arabic nouns show the emergence of broken plurals within Urdu. Moreover, many loan plurals deviate from both the source language and the target language patterns, -in and -at attach to even loan broken plural forms.

This is also due to the fact that many Arabic loans entered Urdu via Persian. Sometimes, two similar nouns e.g. *lohar* 'black smith' and *fair* 'poet' adopt different plural patterns e.g. *lohar* and *fora*. This shows that Urdu nouns have some gender and number rules, but there is often divergence from the rules. Although some loan gender/number patterns exist, the loans often adopt native forms. Such integration causes more than one plural forms, which sometimes follow neither native rules nor loan plural rules. So, they are peripheral target language forms.

There are variations in the gender and number formations. This is because loanwords may take native affixes or loan affixes may also sometimes attach to native bases. This is also frequent in derivational morphology, because it is also based on the three sources of words.

# Chapter 3

## **Derivational Affixes in Urdu**

### 3.1. Introduction

One of the major word formation processes is affixation. The three sources words i.e. native Urdu and Arabic also function in Urdu derivational affixes. Ahmed (2008: 2) states that Urdu has also borrowed a large number of lexical items and many derivational affixes from Persian and Arabic. Therefore, the language of origin plays a vital role in derivational morphology as well. There are various patterns from each source and many variations within each pattern. This chapter focuses on Urdu derivational affixation with reference to these three sources.

The interactions between bases and derivational affixes allow us to study derivation not only from a structural point of view but also from semantic and phonological ones. Although a discussion on structural issues is the main concern, some phonological and semantic issues with relevance to morphological structures are also included. However, phonological issues are only relevant in connection with Arabic loanword phonology, which has also shared in word formations, e.g. yonda 'scoundrel', a product of two distinctive sounds i.e. the voiced velar fricative / y / from Arabic and the voiced retroflex stop /d/ from native Urdu.

The derivation of most words is rule-governed, but there are also words whose affixations deviate from the rules in differing degrees. A derivation of word within a derivational pattern may be grammatical, but it may be ungrammatical for another word in the same pattern. For example, applying a rule, a noun *satf* 'truth' can change into an adjective and then a noun again. But its antonym *dzut* 'lie' does not follow the same rule, as shown below:

$$1a. \ \mathsf{sət} \mathsf{\int} \quad \mathsf{`truth'} \quad (\mathsf{N}) \to \quad \mathsf{sət} \mathsf{\int} \pmb{a} \quad \mathsf{`truthful'} \ (\mathsf{A}) \quad \to \mathsf{sət} \mathsf{\int} \pmb{a} \mathbf{i} \quad \mathsf{`truth'} \qquad (\mathsf{N})$$

b. dʒut 'lie' 
$$(N) \rightarrow d$$
ʒhut $\boldsymbol{a}$  'liar'  $(A) \rightarrow *d$ ʒut $\boldsymbol{a}\boldsymbol{i}$   $(N)$ 

Note that, the noun sat/ai (1a) is grammatical in its second stage of derivation. But unlike it, the other noun \*dzutai (1b) is ungrammatical. There are various similar examples, as Urdu derives a large number of words from Persian and Arabic loans. Derivation forms all major categories e.g. nouns, verbs, adjectives, adverbs and even pronouns through both prefixes and suffixes. Although there are fewer suffixes than prefixes, most words are derived by means of suffixation. There are a number of derivational patterns and many variations within each pattern. Many affixes are productive, and can attach to a variety of words to derive new formations. Some affixes are ancestral i.e. derived from latter stages of Sanskrit and Prakrit. For example, -a and -va create transitive and causative verb roots from intransitive verb roots e.g.  $h\tilde{a}s$  (i):  $h\tilde{a}s$  'laugh' +  $a \to h\tilde{a}sa$  'make (somebody) laugh' and (ii)  $h\tilde{a}s$  'laugh' +  $va \to h\tilde{a}sva$  'get (somebody) laughed'. A causative shows the possibility of a semantic expression.

Aronoff (1976) expresses his doubt about the assumption that morphemes are minimal meaning-bearing units. He states that a morpheme e.g. -mit in permit, submit, and remit, has no sense in isolation. It expresses meaning only after its affixation. Following him Kashif (2004: 91) claims that most Urdu affixes have no meaning of their own and cannot be understood in isolation. However, the same is not true of the affixes borrowed from Persian and Arabic e.g. na 'no' + ehl 'capable'  $\rightarrow naehl$  'incapable'. There are various similar affixes. Kashif is nevertheless right about the native Urdu affixes which do not generally express semantics in isolation. The semantics is discussed only with relevance to structural issues.

Of the various affixes from each source, it is only necessary to discuss some of the most important ones in order to show the various derivational patterns. The native Urdu affixes derive most new words (even) with loanwords, and therefore the discussion starts with them.

### 3.2.1. Native Urdu Affixes

There are few native Urdu affixes, e.g. -a, -i and the negative markers  $\partial$ ,  $\partial n$ . As they form the base of Urdu morphology, they demand particular attention. The formers are vowel based and homonyms of native Urdu gender markers, but also derive new words with the bases of various source languages. They do not show gender morphology with loan adjectives. Derivational patterns by them seem to be systematic and rule based.

There are also some less frequent affixes. Although, in general, native Urdu affixes do not express any meaning, it is not seen in these less frequent ones, as illustrated below:

**Table 1a: Native Urdu Affixes** 

Pref/Suff	Word	Derivation	Word	Derivation
1a/-i	dz <sup>h</sup> ut 'lie' (N)	dz <sup>h</sup> ut <b>a/i</b> 'liar' (A/N)	mehnət (N) 'diligence'	mehn <u>t</u> i 'diligent' (A/N
2i	xof 'happy'(A	xσʃi 'happiness' (N	bora'bad' (A	borai 'evil' (N)
31ја	dak 'mail' (N)	dak <b>ıja</b> 'postman' (N)	kəbar (N) 'wreckage'	kəbaţ <b>ıja</b> 'ragpicker'
4. <i>ə</i>	mər 'die' (V)	əmər (A) 'immortal'	tut (V) 'break'	ətut (A) 'unbreakable'
5. ən 'not'	pərh 'read' (V)	<ul><li>anpəγh illiterate'</li><li>(N)</li></ul>	dʒan 'know' (V)	əndʒan 'stranger' (N)
6. tso 'four'	paja 'foot' (N)	tfopaja 'four footed'animal' (N	rah 'path' (N)	tforaha 'crossroad' (N

The suffixes -*a* and -*i* are quite distinctive even among the most productive affixes. Plag (1999) defines morphological productivity as "the property of a given word formation process to be used to derive a new word in a systematic fashion." This systematic fashion is observed in these affixes, and so they are quite productive to derive new words. They attach to nouns and mainly derive adjectives. The derivation expresses gender, if the base noun is native. If the base noun is a loan, only -*i* attaches, and the derived adjective expresses no gender.

```
2. N + -a/-i \rightarrow A/(N) (Gender distinction with native noun)

a. sət \int 'truth' (native) \rightarrow sət \int a (m) / sət \int i (f) 'true/ truthful person' (A/N)

b. d3^{fi}ut 'lie' (native) \rightarrow d3^{fi}uta (m) / d3^{fi}uti (f) 'lying/liar' (A/N)

c. mehnət 'hard work' (loan) \rightarrow mehnti (N/A) 'diligent'

d. d3^{fi}uti 'heaven' (loan) \rightarrow d3^{fi}uti (N/A) 'the one rewarded with Heaven'
```

The nouns (2a & 2b) take either -a or -i and show gender marking. The newly derived words are typically adjectives e.g. sətʃa kam 'true work' (2a) and dʒʰuti xatun 'a lying woman' (2b). But they may also be used as nouns e.g. ek dʒʰuta 'a liar'. The ones in (2c & 2d) are loans and so take only -i without specifying any gender. Both the derived adjectives can be interpreted as either masculine or feminine, although the borrowed nouns i.e. mehnət 'hard work' and dʒənnət 'heaven' are treated as feminine. Most of the base nouns that -i attaches to, are common nouns and abstract nouns and are treated as feminine. The native and the loan nouns differ in that the derivations from the native nouns are mainly adjectives but may also be used as nouns. The derivations from the loans are only adjectives and can rarely be used as nouns.

Morphology sometimes interacts with phonology. In the process of derivation, there may also be some minor structural changes. For example, -o- is deleted and the original coda becomes

an onset after the change of category from *mehnət* to *mehnţi* (2c). This is because the final-i allows the -t- to function as an onset, which it cannot do when word-final. This is also because the nasal /n/ is pre-aspirated [ $^h$ n] without requiring any vowel. The structural changes illustrate what was said earlier that the affixations sometimes deviate from expected patterns of morphology, which is seen in almost every pattern of word formation.

The suffix -i is more comprehensive in its role. It also transforms a large number of adjectives into nouns, which is quite a contrast to the previous function. It often attaches to a native or loan adjective without any distinction. Consider the examples below:

3. A + - 
$$i \rightarrow N$$

a. 
$$x \circ j$$
 'happy' (A)  $\to x \circ j i$  'happiness' (N) (Persian)

c. 
$$bvra$$
 'bad' (A)  $\rightarrow bvra$ i 'evil' (N) (Native)

d. 
$$\operatorname{ət} \int a$$
 'good' (A)  $\to \operatorname{ət} \int a\mathbf{i}$  'goodness' (N) (Native)

Note that the derivation from adjective to noun is possible with both native and loan bases. If the base is a loanword, the suffixation of -i is enough without differentiating the source language, as in (3a & 3b). But a native adjective shows gender marking, and so -i is only possible with masculine bases, as in (3c & 3d). The affixation of -i with a feminine base means doubling the final -i e.g. \*b\vartit{o}ri\vartit{i}\$ and so it is ungrammatical, although a word-final long vowel may be accepted in other instances. This shows that there are two suffixes -i, with different effects and that they both have the same pronunciation is in a way accidental. However, the affixation of -i to derive noun from adjective is possible only if the base is

originally an adjective. It is generallay ungrammatical, if the base is already derived from a noun as in (3e), although there may be exceptional cases.

This discussion has shown that -i is a productive suffix, and derives both nouns and adjectives. It derives adjectives, from native nouns, showing feminine gender. It is thus parallel to -a, which derives adjectives showing masculine gender. It sometimes requires the base noun to undergo structural changes in the coda. Bauer (2003: 70) observes, "a morphological process can be said to be more or less productive according to the number of new words which it is used to form." As the two affixes can conveniently aattach to native and loan bases, it can be said that both -a and -i are productive affixes.

A parallel of -*a* is -*ija*, which derives masculine agentive nouns and has no feminine counterpart. This thing distinguishes it from -*a*. Moreover, it is very limited in word formation as compared to -*a* and -*i*, which seem to derive most words. Unlike -*a*, it is distinctive in that it may derive new words with loan bases as well. The examples below illustrate this:

The suffix -*ija* derives only agentive masculine nouns from nouns e.g. *dakija* 'postman' in (4a). The derivations are very small in number and have no feminine counterpart, but they are the same in form as some feminine nouns e.g. *kottija* 'bitch'. As discussed in the previous chapter, the feminine nouns are base nouns and not derivations. Therefore, the two are different. The affixation of -*ija* with loanwords e.g. in (4c) is also possible. It is

ungrammatical to attach a feminine marker e.g. \*dakijai or treat the masculine derivation also as feminine. Thus, it also shows irregularity and it becomes evident that Urdu morphological patterns are complex and often seem to deviate from the rules, as all the three suffixes also show deviations from their patterns in one way or the other.

There are no other important suffixes whereas among the prefixes the negative markers  $\partial$  and  $\partial n$  are noteworthy. Although not very productive, they do form some new words. Both  $\partial$  and  $\partial n$  seem to function in exactly the same way. They mainly derive adjectives from verbs and the derivation is an antonym variant i.e. it is from a different category. The derivation from  $\partial n$  may also be used as an adverb. Consider the following examples:

- 5.  $\ni$  + V (base form)  $\rightarrow$  A(dj)
- a.  $\mathfrak{d}$  + mər 'die'  $\rightarrow$   $\rightarrow$   $\rightarrow$  mər 'immortal' (A)
- b.  $\mathfrak{d}$  + tut 'break'  $\rightarrow$   $\mathfrak{d}$ tut 'unbreakable' (A)
- 6.  $\operatorname{an} + V$  (base form)  $\rightarrow$  A (may be used as Adv or Noun)
- a. n + part' 'read/study'  $\rightarrow$  anpart' 'illiterate/uneducated' (A/N)
- b.  $\operatorname{an} + \underline{t} \operatorname{hak}$  'exhaust'  $\rightarrow$   $\operatorname{anthalk}$  'untiring/untiringly' (A/Adv)

Both  $\partial$  and  $\partial n$  attach to verbs but are different in derivation. They form deverbal adjectives, but the formations with the latter may be used as nouns or adverbs as in (6). Both are restricted to some specific root verbs and cannot replace each other e.g. \* $\partial t^h \partial k$  or \* $\partial t^h \partial$ 

In sum, the main native Urdu affixes -a, -i, -ija and the prefixal negative markers  $\partial$ ,  $\partial n$  are only few, but they derive various new category words. Although the prefixes playing the role of negative markers are important, the suffixes have the key role in the derivation. They attach

to nouns and adjectives and derive all major category words. These affixes are fewer than loan counterparts, but they derive nouns and adjectives from Persian and Arabic loan bases also. Gender morphology with loan adjectives is irrelevant. This means that there may be various exceptions particularly in case of loanwords. The irregularity is nevertheless not in the sense of being 'unpredictable'. The affixes certainly show some rules, patterns and a system.

It is also sensible that the major native affixes do not show semantics in isolation, although not about the less frequent ones e.g. *tfo* 'four'. Moreover, Persian and Arabic loan affixes also deny this. Persian affixes are the most of all and generally express semantics in isolation.

### 3.3. Persian Loan Affixes

Naim (1999) states that most of the commonly used words in Urdu are of Persian and Arabic origin, otherwise these languages have exerted little or no influence on grammatical structure. He terms these words Perso-Arabic, as many Arabic loans entered via Persian, which makes extensive word building, by derivational agglutination. There is a large number of Persian loan affixes. They are productive in that they may attach to words other than those of Persian itself. However, they derive fewer words than do the native affixes. Suffixes predominate in Persian itself, but prefixes are the dominant loans in Urdu. The majority are negative markers and express from absolute to partial negation. Some negatives e.g. na 'no/not' derive mostly adjectives from nouns and nouns from nouns. Although suffixes are not very frequent, some e.g. -agi, -ana and -amez interestingly derive more words with non Persian bases. Thus, they show complete integration, and seem more native affixes than Persian loans.

This section discusses only a few loan affixes, giving a glimpse into their integration with words from various sources. Some parallel examples with non Persian bases are given below:

Table 2: Persian Loan Affixes in Urdu

Prefix/Suffix	Word	Derivation	Word	Derivation
1. na 'no/not'	bina 'clear-sighted'	nabina 'blind'	mərd	<b>na</b> mərd
	(N/A	(N	'man'(N	'impotent man' (N)
2.be 'without	qərar (N) 'rest'	begerar (A) 'in	hisab (N)	behisab (A)
		panic'	calculation'	'countless'
3. kədʒ	fehem 'sense' (N)	kəd3fehem	rəf <u>t</u> ar	kəd3rəftar 'slow'
'little'		'credulous' (A)	'speed' (N)	(A)
4. kəm	ılm (N) knowledge	kəmilm 'less	əqəl 'wit'	<b>'kəm</b> əqəl' 'stupid'
'little'		educated' (A)	(N)	(A)
5. pəs <u>t</u>	hımmət 'courage' (N)	pəst himmət	fıtrət	pəstfitrət 'mean
'little/mean'		'coward' (A)	'nature' (N)	nature' (A)
6. no 'new'	dʒəvan (N) 'of	nodzevan (N)	dolət (n)	nogolatia (N)
	mature age'	'young man'	'wealth'	'newly rich'
7. por 'full	dzos 'excitment' (N)	pordzof (A)	məyəz (N)	<b>pσr</b> məγəz (A
of'		'excited'	'mind/brain'	'thoughtful'
8. bala	xana (N) 'house'	bala xana (N)	dəst (N)	baladəst (Adj)
'above/up'		'upstairs'	'hand'	'strong'
9. ba	əndaza (N) guess	<b>ba</b> əndaza (Adv)	vədzud (N)	<b>ba</b> vədʒud (Adv)
'having'		with guess	existence	'despite'
10. zud (sth	əsər 'effect' (N)	<b>zud</b> əsər	həzəm (N)	<b>zud</b> həzəm
that does sth)		'effective' (A)	'digestion'	'digestive' (A)

11(ə)gi	zında 'alive' (A)	zındəgi 'life' (N)	hævan 'animal' (N)	hævan <b>gi</b> 'animal' (N)
12 <i>a</i> n <i>a</i>	mərd 'man' (N)	mərd <b>an</b> a	bətsa 'child'	bətʃk <b>ana</b> 'childish'
		'manly'(A/Adv)	(N)	(A)
13 <i>a</i> mez	ſikajət'complaint' (N)	ſīkajə <b>tamez</b>	rəŋ 'colour'	rəŋ amez 'dyed' (A)
'full of'		'plaintive' (A)	(N)	
14∫ʊd̪ <i>a</i>	gom 'lose' (V)	gʊm <b>ʃʊd̪a</b> 'the	xʊʃk	xบริkร์บ <u>ส</u> a 'dried' (A)
'done'		lost one' (A)	'dry'(A)	
15dar	ızzət (N) 'respect'	ızzə <b>tdar</b>	vəfa (N)	vəfa <b>dar</b> (A) 'loyal'
'showing'		(A) 'respectable'	'loyalty'	
16ıstan	qəbər 'grave' (N)	qəbr <b>ʊst̪an</b>	pak (n)	Pak <b>ıstan</b> 'Pakistan'
'place/land'		'graveyard' (N)	'pure'	(N)

Of the sixteen affixes, ten are prefixes (1-10) of which five (1-5) are negative markers. The first two *na* 'no/not', and *be* 'without' express absolute negation and the others i.e. *kədʒ* 'little', *kəm* 'little' and *pəst* 'little/mean' express partial negation. In general, *na* and *kəm* are the most frequent both in spoken and written Urdu.

Used as a noun in rare cases, na has a variety of functions as a negative marker. Functionally, it is the same as the native negatives  $\boldsymbol{\partial}$  and  $\boldsymbol{\partial} n$  and Arabic la, but there are major structural differences. Unlike native  $\boldsymbol{\partial}$  and  $\boldsymbol{\partial} n$  which attach to verbs only and yield mainly adjectives, Persian na attaches to adjectives, nouns and even verbs and derives adjectives and nouns  $(na + A/N/(V) \rightarrow A/N)$ . Moreover, its affixation with nouns may not necessarily change the

category of the derivation. For example, it derives a noun *naafna* 'stranger' with a noun *afna* 'familiar person', but its affixation with another noun e.g. *omid* 'hope' forms an adjective i.e. *naomid* 'hopeless'. Such patterns of changes with the same category of words in two different directions justify irregularity. There are nevertheless some observations. When *na* affixes animate common nouns e.g. *afna* 'familiar person', it doesn't change the category. When it interlinks with abstract nouns e.g. *omid* 'hope', it transforms the base nouns into adjectives.

Although a few examples are found, the affixation of *na* is possible with Arabic loan nouns e.g. *namehram* '(for woman) a non blood relative with whom marriage is allowed' and adjectives e.g. *namaqul* 'inept'. It does not change the category but shows some complex formations. As seen in the examples, some formations show no changes in the bases, but in some others the affixation of *na* requires the deletion of final vowel in the stems. Some base adjectives, e.g. *razi* 'consented', are prefixed with *na* to derive an antonym *naraz* 'angry', which must lose the final vowel. The derivation is ungrammatical \**narazi* 'angry', if the deletion does not occur. The derived words keep two syllables *na.raz*, and the third -*i* is deleted. Nonetheless, this is not the general feature in the affixation of *na* with Arabic loans. It only shows complexities in the affixations.

Unlike the Persian and Arabic bases, the native Urdu stems are base verb froms e.g. somod3h 'understand', although there are very few of them. The formations with them are nouns or adjectives  $(na + V \rightarrow A/N)$  e.g. nasomod3h 'credulous'. However, the use of na in this formation type is restricted to few derivations, because most verbs are affixed with native Urdu native negative markers o and on, which convert the verbs into nouns and adjectives. Nevertheless, the affixation of na strengthens the perception that some patterns, in each case, are complex in rules and often show deviations.

The alternations shown by *na* are in the majority of all Persian loan prefixes. They seem even more complex by some suffixes, which cause morphological changes in the bases. Although most suffixes derive adjectives, some derivations may be nouns. *-gi* needs particular attention, as it attaches to both adjectives and nouns but only derives nouns. Non Persian base nouns sometimes show no morphological changes. Consider the following:

8a. 
$$zinda(h)$$
 'alive'  $+-gi \rightarrow zindagi$  'life'  $(A + -gi \rightarrow N)$  (Persian)

b.  $darinda(h)$  'beast'  $+-gi \rightarrow darindagi$  'beastliness'  $(N + -gi \rightarrow N)$  (Persian)

c.  $\int arminda(h)$  'shameful'  $+-gi \rightarrow farmindagi$  'shamefulness'  $(A + -gi \rightarrow N)$  (Arabic)

d.  $\int arminda(h)$  'animal'  $+-gi \rightarrow hevangi$  'wildness/beastliness'  $(N + -gi \rightarrow N)$  (Arabic)

The four examples show the affixation of -gi with both nouns and adjectives, but the derivations are only nouns. The basewords in (8a-8b) are Persian loans, but those in (8c-8d) are Arabic loans. The affixation generally causes some phonological alternations in the basewords, although (8d) shows no change. The coda vowel -a(h) is substituted by the schwa  $-\partial$ - along with the affixation of -gi for the derivation. The Persian loan suffix -gi is a non neutral suffux, according to Katamba's (1993: 89) definition, as it causes structural changes in the adjectives and nouns ending with -a(h). Merely the affixation of -gi to Persian or Arabic bases is not enough to transform them into nouns. Therefore, \*z - mdagi 'life' and \* $\int a - mm dagi$  'shamefulness' are ungrammatical. The final vowel must be replaced with -a-, and then the morpheme -gi is attached to the basewords, e.g. z - mdagi 'life' (8a) or  $\int a - mm dagi$  (8c). This shows no distinction of the source language, and the derivation refers to the nativisation.

The affixation rule applies the same way. However, it may cause morphological differences in the derived nouns, as seen in Arabic *hevangi* (8d) and the Persian *dərındəgi* (8d). Like the adjective, the Arabic noun is not itself a derived form. Moreover, it does not end with -a(h),

and so its closed coda only takes the suffix -gi. Further, -gi also shows that Arabic loans are the convenient bases for the Persian affixes which can attach to the bases of a variety of source languages. Ending with -da(h) is the major cause of the changes by the Persian -gi. The affixation with two base endings also refers to the previous argument that although Urdu derivations follow some patterns they may be complex and may cut across linguistic components. The affixes are thus completely integrated and nativised.

Some other suffixes e.g. -ana, -amez and -dar also show some interesting nativisation, because they mostly attach to non Persian bases. Among them, -ana is particularly flexible in the affixation and can freely attach to a base of any of the major source languages discussed. The following example illustrates the process properly:

9a. mərd 'man' 
$$+ -ana \rightarrow mərd$$
 'manly'  $(N + -ana \rightarrow A)$  (Persian)  
b. radzput (a surname)  $+ -ana \rightarrow radz$ put 'like a radzput'  $(N + -ana \rightarrow A)$  (Native)  
c. zalım 'tyrant'  $+ -ana \rightarrow z$ alım 'tyrant'  $(A/N + -ana \rightarrow A)$  (Arabic)  
d. fəlsəfi 'philosopher'  $+ -ana \rightarrow f$ əlsəfij **ana** 'philosophical'  $(N + -ana \rightarrow A)$  (English)

The four derivations by Persian -ana show its affixation with the base nouns of four source languages. The affixation with a Persian base (9a) is not an unusual matter. Although on rare occasions it can also attach to native Urdu bases as in (9b). This is also expected to be a result of the diachronic process. However, its derivations with Arabic loans (9c) and particularly English loans (9d) are surprising and show the nativisation of Persian affixes.

Despite being large in number and having the capacity to attach to a variety of bases, Persian loan affixes derive fewer words than those from native ones. There are more loan prefixes than suffixes, and most of them are negative markers. Some of the suffixes e.g. -ana show

various derivations with even non Persian bases. This again refers to the complexities in Urdu derivational morphology. Although the derivation follows some patterns and rules, the rules are complex and may cut across linguistic components. Therefore, Naim (1999) rightly claims that there is frequent derivational agglutination and the formations deviate from Urdu and Persian patterns. The loan affixes are so completely integrated that they seem more native affixes than Persian loans. Various such derivational patterns with many variations are seen.

This is also because some Persian loan affixes easily attach to Arabic loan bases. A large number of Arabic loanwords can be seen in Urdu though Arabic affixes are the rarest of the three source languages. Moreover, the interaction of morphology and phonology seen in (8) may be an exceptional case in Persian loans, whilst it is often seen in Arabic loans.

## 3.4. Derivation in Arabic Loans by Affixation and by Phonological Changes

Sound change in loanwords is a normal process. It becomes interesting, when it affects the formation of new words. The derivational morphology of Arabic loans in Urdu is not restricted to affixation and compounding. Rather, it shows that the modification of base by internal vowel change, e.g. a noun *qətəl* 'murder', can change into another noun *qatıl* 'killer'. Thus, there is often an interaction of morphology and phonology. Affixation also shows this interaction many times. For example, when used word medially, the Arabic loan definite article *vol/əl* follows the source language rule, and the lateral /l/ is deleted, which results in the doubling of the following consonant e.g. *kttabv(l)nnur* 'the book of light'. However, the Arabic rule is discarded word initially e.g. *əlnur* 'the light'. Kaisse (2005) concludes that "the relation between word formation and phonology is complex." Arabic loanword derivation also shows various complex features in the interaction between morphology and phonology. Although the interaction is not the issue here, and the main concern is morphology, Arabic

loanwords often show this interaction. So, it becomes necessary to see in the end of this chapter some major features of Arabic loanword phonology. Some key interesting features of this include rise and fall of Arabic phonemes.

There are four major points discussed in this chapter: 1) derivation by Arabic loan affixes 2) derivation by the modification of Arabic loan bases (by phonological changes in them) 3) the nativisation of loan morphemes in a manner that they affix even with many words derived by phonological changes and 4) loanword phonology. This section includes some discussion on the neutralisation of many Arabic loan phonemes (losing their phonetic value) and their rise and reversal from native Urdu words.

An interaction between morphology and phonology may cause inflectional and derivational changes. Derivational changes can be in two ways i.e. by affixation and by modification of base. Some affixes, e.g. the Arabic loan definite article  $\partial l/\partial l$ , also show this interaction.

## 3.4.1.1. Derivation by Affixation

Although there are a large number of Arabic loanwords, there are not as many derivational affixes, as those of Persian. But they are used very frequently. Most of them are negative markers. Structurally and functionally, the negatives are different from native and Persian negative markers. They are also different from each other. For example, la attaches to nouns and forms both nouns and adjectives. It refers to absolute negation. bila also attaches to nouns, but it forms adverbs. It refers to a lack of something. Among other loan affixes are the adverbial suffix  $-\partial n$  and the definite article  $\partial l/\partial l$ , which is structurally similar to its appearance in Arabic but functionally different in Urdu. This section describes only the structures of a few affixes. It shows in particular that  $\partial l/\partial l$  connects two independent nouns into a single

whole. It also partly supports Naim's (1999) claim that  $\partial l/\partial l$  is mostly found in nominal and adjectival compounds, e.g.  $dar \underline{\sigma} l \sigma l u m$  'educational institute' in which the second constituent is made definite. Consider the table below illustrating some important affixes:

Table 3: Arabic Loan Affixes in Urdu

Pref/Suff	Word	Derivation	Word	Derivation
1. la 'not'	sani 'second'	lasani (A)	∫əur	lasəur (N)
	(A)	'incomparable'	'consciousness'	'unconsciousness'
2.bila	zərurə <u>t</u> (N)	bılazərurə <u>t</u> Adv)	dʒəvaz (N)	bıladʒəvaz (Adv)
'without'	'necessity'	'unnecessarily'	'reason'	'unreasonably'
3. yær	zəruri (A)	<b>yær</b> zəruri (A)	hazır (A)	<b>yær</b> hazır (A)
'without'	'necessary'	'unnecessary'	'present'	'absent'
4ən	ıttfaq (N)	ıttfaqən (Adv)	əmanət 'trust'	əmanə <u>t</u> ən (Adv)
	'coincidence'	'coincidentally'		'by trust'
5.zi 'having	fəur 'wit' (N)	zisəur 'wise' (N)	ruh (N) 'soul'	ziruh 'animate' (A
6. อไ/ชไ	Nurv(I) ddin			

There are three loan negative markers i.e. la 'no/not', bila 'without' and yar 'stranger/without', the definite article  $\partial l/\partial l$  and a few other suffixes e.g. adverbial  $-\partial n$ . The negative yar is among the most frequent affixes in spoken and written Urdu. It derives adjectives from adjectives and nouns from nouns, as shown below:

10a. zəruri 'necessary' (A) → **yær**zəruri '**un**necessary' (A)

b. mərd 'man'  $(N) \rightarrow \gamma \alpha r m$ ərd '(for woman in a religious sense) non relative man' (N)

In deriving adjectives from adjectives and nouns from nouns, *yær* shows some interesting structural and semantic features in comparison with the Persian *na*. Unlike *na*, it forms only antonyms. Like *na*, it can also construct a noun from a noun, but there are not many examples. Both can attach to Persian or Arabic nouns but their affixation with native nouns is rarely seen. In rare cases, both have interchangeability e.g. *yærməsavi/ naməsavi* 'unbalanced'. Sometimes, the interchanging may also cause semantic change e.g. in two synonyms i.e. *admi* 'man' (from Hebrew via Persian), and *mərd* 'man' (Persian).

```
11a. admi 'human/man' (N) \rightarrow yæradmi 'stranger' (N)
```

- b. marg 'man' (N)  $\rightarrow$  yærmarg '(for woman in a religious sense) non relative' (N)
- c. admi 'human/man' (N)  $\rightarrow *na$ -admi 'stranger' (N)
- d. marg 'man' (N)  $\rightarrow$  **n**amarg 'impotent' (A/N)

As (11a & 11b) show this, *yær* attaches to both *admi* and *mərd* and forms antonyms with two different meanings. *na* is ungrammatical with *admi* as in (11c) and affixing with *mərd* 'man', it shows a quite a different meaning as in (11d). Comparing (11b) and (11d) shows the interesting combination of two negative markers with the same noun. The first derives a noun, as in (11b), and the other an adjective that is also used as a noun, as in (11d).

The affixation of both with the three nouns i.e. *məsavi* 'balanced', *mərd* 'man' and *admi* 'human/man' shows three features. First, they may be interchangeable with no semantic change, e.g. *yærməsavi/naməsavi*. Second, their interchanging may also cause semantic changes, e.g. (11b & 11d), and third their interchange ability is limited and not possible in every case, e.g. (11a & 11c). Although both function as native negative markers and share few factors in common, the differences are dominant. The three different results of the same process show that there are some rules, assumptions and possibilities while at the same time

there are frequent deviations from the rules. These deviations exist because both negative markers are from two different sources. As noted in the previous section, na is wide and comprehensive. It attaches to nouns, adjectives and some verbs. Moreover, it forms both nouns and adjectives. By contrast, yar is limited. It mainly derives adjectives from adjectives, but sometimes it derives adjectives also from nouns, whether bases may be Persian.

Like *yær*, other Arabic negative markers *la* 'no/not' and *bıla* 'without' can also attach to Persian and native bases. However, they are not interchangeable with the native or Persian loan affixes. They are comparatively more restricted in use, but they show structural variations. *la* is more frequent, and attaches to nouns and adjectives and derives antonyms:

12a. four 'consciousness' (N) (Arabic) 
$$\rightarrow la$$
 four 'unconsciousness' (N)

b. sani 'second' (A) (Arabic) 
$$\rightarrow la$$
sani 'incomparable' (A)

c. pərva 'care' (N) (Persian) 
$$\rightarrow$$
 lapərva 'careless' (A)

d. zərurət 'need' (N) (Arabic) 
$$\rightarrow$$
 **bıl**azərurət 'unnecessarily' (Adv)

e. tſuk 'mistake' (N) (Native) 
$$\rightarrow$$
 **bıla**tſuk 'unmistakingly' (Adv)

f. 
$$t \tilde{u}$$
-v-t  $\hat{u}$ 

la attaches to both nouns and adjectives and forms the same categories as in (12a) and (12b). But (12c) shows that the derived category may be different from that of bases, although in some cases. For example, the Persian base parva is noun but the derivation laparva is adjective. As seen, la can affix with Persian or native Urdu bases as well. In general, it shows fewer word formations than yar. It can also be observed in bila 'without' (12d), which is the least frequent in word formations in all negative markers. It attaches to a noun but forms an adverb (12d). Like some other Arabic negatives, it can also conveniently attach to native and Persian bases e.g. bilatfuk (12e) and bilatfuraction variations are categories as in (12a) and <math>bilatfuraction variations as in (12b).

Thus, generally seen, Arabic negative markers are prefixes and function like native affixes. They are only a few and restricted in function. However, they show a variety of affixation with native and Persian bases. Moreover, like other negatives, they show some interchange ability with Persian affixes.

Other than negative markers, Arabic loan affixes also include the definite article  $\partial l/\partial l$ . Used mostly as a prefix or an infix, it has mostly lost its original status. It occurs mainly at a prefix position, and may attach to noun and adjective. As an infix, it can connect two nouns to form a proper noun e.g.  $\int \partial m/\sigma(l) ddin$  'the light of the religion'. The two constituents  $\int \partial m/\sigma(l) ddin$  'sun' and din 'light' may seem to be two separate nouns, but the proper noun functions as a single unit by virtue of some morpho-phonological features shown by  $\sigma l$ . The lateral l/ is deleted and the following consonant is doubled. As a result, the so called definite article becomes a binding force that connects the two separate nouns into a single whole, which is only taken as single name, not a compound. There is no first or second name in it, which means that both the definite article and the second constituent 'din', have been completely incorporated. With such names as this, a pserson can be addressed as famf but not as \*v(l)ddin, which is ungrammatical. This shows that  $\partial l/\partial l$  has lost its status as a definite article  $\partial l$ . However, this is partially true, and  $\partial l/\partial l$  is found in nominal and adjectival compounds e.g.  $\underline{dar}\underline{\sigma l}$  olum 'educational institute', in which the second constituent is made definite. The semantic role of the definite article is not the issue of concern here. As an infix, it may form proper nouns as single units or compounds. It has no suffixing role.

There are nevertheless some important loan suffixes, e.g. the adverbial -ən. It is is distinctive among loan affixes in that it does not convey any meaning at all. It derives adverbs from nouns e.g. ittfaqən 'coincidentally'. But its affixation with nouns is restricted to mainly

dental, nasal and uvular stops at coda position i.e.  $/\mathbf{t}/$ ,  $/\mathbf{q}/$ ,  $/\mathbf{n}/$ ,  $/\mathbf{m}/$  and  $/\mathbf{q}/$  respectively. It cannot attach to any noun with an open coda, as shown below:

```
13a. Irada(h) 'intention' + ən → *Iradaən
b. Irada(h) 'intention' + ən → *Iradən
c. Irada(h) 'intention' + ən → Iradatən 'intentionally'
```

Example (13) exhibits morphophonological harmony of -ən with the base nouns. Many base nouns e.g. tradga(h) 'intention' end with an open coda, which is the result of the deletion of final aspiration in the noun tradga(h). As noted in chapter 2, the deletion of the final aspiration is a common feature in Arabic loans. However, in this case, only affixation of -ən causes ungrammaticality, as shown in (13a). Its affixation requires dental, nasal and uvular stops at coda position, as said above. There are two options here then, either the deletion of final vowel -a or its affixation along with one of the consonant required. In the former case, the derivation is ungrammatical again as shown in (13b). It in fact requires an insertion of voiceless dental /t/, which becomes necessary along with its affixation. Why a voiceless dental /t/ should necessarily be inserted is only because many Arabic nouns treated as feminine originally end with an underlying voiceless dental /t/, but in Urdu the dental sound is not pronounced at all. This is quite contrasting situation. The dental /t/ and the aspiration /h/ are deleted and not pronounced in many nouns, but in the transformation of noun to adverb the dental is reinserted after the affixation of the adverbial -ən, which causes the reinsertion and the modification of base noun.

This formation of adverbs by  $-\partial n$  with a noun is parallel to compounding by ablative/instrumental marker se with a noun e.g. ntifaq se 'coincidentally'. Although both can

sometimes interchange with each other, there is a major difference in that  $-\partial n$  is limited in deriving adverbs, and it attaches to mostly nouns with certain coda consonants:

```
14a. Ittifaq 'chance' + ən → Ittfaqən 'coincidentally' /Ittifaq se 'coincidentally'
b. dʒəldi 'quickness' + ən → * dʒəldiən / dʒəldi 'quickness' + se → dʒəldi se
c. əmanət 'trust' + ən → əmanətən 'with/by trust'/*əmanət se
```

Both the instrumental *se* and adverbial suffix *-ən* can sometimes interchange, and both can substitute for each other, as in (14a). But it is not always possible, as shown the ungrammaticality of *-ən* with an open coda in (14b). There are also certain places where the affixation of *-ən* is possible but that of the ablative *se* is ungrammatical, as in (14c), due to a semantic constraint. *əmanəţ se* is morphologically correct, but there is no semantic expression. *əmanəţ* is culturally taken as something kept (for a certain period) by somebody to return it to its owner at a proper time. As *se* describes only manner of action, it cannot give the correct semantic expression, while *-ən* can. Thus, *əmanəṭən* means something kept as a trust to be returned to its owner. In general it can be seen that *-ən* is sometimes interchangeable with *se*, but at other times one of them shows ungrammaticality.

In every pattern, Urdu affixation shows some rules, though there are also deviations from the rules. As seen,  $-\partial n$  cannot affix with an open coda base but this rule is irrelevant if a noun ends with -a(h). Although the final aspiration is deleted, it allows dental  $/\underline{t}$ / along with the affixation. This shows that there sometimes its affixation also requires modification of base. A large number of Arabic loans do not require affixes for derivation. Rather, they undergo some modification in their base form.

# 3.4.1.2. Derivation by Modification of Base (Phonological Changes)

The Arabic loan affixes discussed derive only a limited number of formations. Most loanword derivations are based on the modification of base. Bauer (2003) states that where affixes are not used for creating new words, the most widely attested method is to make some kind of phonological change to base. This involves discontinuous affixes and discontinuous bases.

Arabic derives various categories of words by making phonological changes in tri consonantal bases e.g. zlm. The derivations are however no more than loans, as the process of deriving paradigms from trilateral roots does not occur in Urdu. In addition, the loan derivations are restricted to some patterns and each loan pattern contains only a few derived words. They are used in the same way as the original forms. For example, a noun e.g. zolm 'tyranny' can become an adjective i.e. zalm 'tyrant' or mozlum 'tyrannised'. The three forms are known and commonly used, but many other derivations are uncommon. Moreover, each loan derivation from a tri consonantal root may undergo a nativisation process by some affixation. Thus, by this affixation, there are some native Urdu derivations, not part of Arabic derivation by the modification of base or phonological changes. The three loan derivations are thus nativised, as zolmi 'cruel person', zalmana 'tyrannical' or mozlumana 'tyrannised'. There are thus both the processes, the modification of the base first and then the affixation. The latter is both by native and Persian affixes. The Arabic patterns of derivation by modification of base and their nativisation by the affixation go together. Co-occurrence of affixation and the modification of base is a common phenomenon and is termed transfixes, by Bauer (2003).

All derivations, whether loanwords or nativised forms, derive two major categories of words i.e. nouns and adjectives. Table 6 below shows only some loanword patterns of derivations. It gives a brief explanation of all loan patterns and then nativised derivational forms in Urdu.

Table 4: Some Arabic derivational Patterns in Urdu

Derivation	Root $1 = \dot{z}lm$	<b>Root 2</b> = \$\int 9 \int 9	Root 3= ðlt	Root 4= Slt	Root 5= ṣḍq
a.	zvlm (N)	Isq (N)	zıllə <u>t</u> (N)	ıllə <u>t</u> (N)	sıdq (N)
	'tyranny'	'adoration'	'disgrace'	'illness'	'truth'
b.	zalım (N/A	asiq (N/A 'lover'	zəlil (A)	əlil (A) 'sick'	sədaqət (N)
	'tyrant'		'disgraced'		'truth'
c.	məzl <b>u</b> m (N/A)	maʃ <b>u</b> q (N/A)	zəlalə <u>t</u> (N)	əlalət (N)	təsdiq (N)
	'tyrannised'	'beloved'	'baseness'	'sickness'	'verification'
d.			təzlil (N)	talil (N)	mʊsəd̪dɪqa
			'disgrace'	'cause'	'verified' (A)

Note: All emphatic sounds in brackets are neutralised and spoken as: z/z, ð/z, s/s & S/vowel.

The concept of triconsonantal root, as known in Arabic, is itself absolutely unknown in Urdu. The derivational patterns in Arabic are pronounced with Arabic emphatic sound, but an emphatic sound in the loanwords in Urdu must be modified to be pronounceable. Therefore, the derivations from triconsonantal roots are shown with native sounds. In addition to the modification of base, there is sometimes an insertion of a consonant-vowel prefix in almost every root pattern e.g.  $m\partial$  in  $m\partial zlum$  'tyranised'. Generally a prefix is formed from  $m\partial zlum$  in combination with a vowel. Almost every root pattern contains at least one derivation with such prefixes. This prefixation is used in the same way as in the source language.

The patterns of derivations are generally used the same as in Arabic, and so they are no more than borrowed forms. Arabic may have various derivations from a root pattern, but they are not necessarily used in Urdu. For example, a derivation <u>təssəddəv</u>q 'alms giving' (N) from the root <u>sdq</u> is not used with the original meaning, but it is used as a proper noun. There are various similar loan patterns. Moreover, sometimes Arabic derivations by modification of base take native Urdu, Persian or even Arabic affixes to form nativised derivations. For

example, the noun  $z\varpi lm$  'tyranny' can be converted into an agentive noun  $z\varpi lmi$  'cruel man' by a native suffix -i. However, it may be expected. But it is interesting to see when a loan suffix from one source attaches to a loanword from another source.

# 3.4.1.3. Affixation with the Derivations by Modification of Bases

Arabic loanwords are often adaptable enough to take at least a few specific affixes from not just native Urdu, e.g. -*i* but also from Persian, e.g. -*ana* and -*amez* and even from Arabic, e.g. -*ijat*. In the process of this nativisation, -*ijat* is different from its original function in Arabic.

The affixation of the morphemes from the three sources with particularly the derivations by modification of bases is a major interesting phenomenon and requires attention. The table below illustrates some nativised derivations:

Table 5: Arabic loan bases and derivations, by modification of bases, with Urdu Affixes

Affixes	<b>Derivation A</b>	<b>Derivation B</b>	<b>Derivation C</b>	<b>Derivation D</b>
1i (Nat)	zolm <b>i</b> 'cruel	asiqi 'love' (N)		təsdiqi 'verified' (A)
	man'(A/N			
2 <i>a</i> n <i>a</i>	zalım <b>ana</b> 'oppressive'	asiq <b>ana</b> 'loving'	zəlil <b>ana</b>	
(Per)				
	məzlum <b>ana</b> 'like an	masuq <b>ana</b> 'like		
	oppressed' (A)	beloved'		
3amez			zıllətamez (A)	
(Per				
			zəl <i>a</i> lə <b>tamez-</b> A	
4. <b>-</b> ∫ʊd̪a				təsdiq <b>ʃʊda</b> 'verified'(A
(Per)				

5ıjə <u>t</u>	məzlum <b>ıjət</b> 'the state		
(Ar)	of oppressed' (N)'		
6ıj <i>a</i> (h)		Ifq $ija(h)$ 'of love	
(Ar)			
7a(h)			mʊsəddɪqqa(h)'verified'
(Ar)			

The table shows the possibility of various affixes, from the three sources, to attach to the Arabic loanword derivations by phonological changes in bases. There may be others, but the affixes mostly attached are native -i, Persian -ana, foda and -amez, and Arabic -ijot, -a(h)/-ija(h). However, these affixes attach only to a few loanwords which are adaptable enough. They cannot freely attach to any of the loanwords. Most Arabic loans are adaptable with the native Urdu affix -i and the Persian -ana, which also seem to be the most productive with them. The native -i has a variety of formations e.g. zolmi (N/A), afiqi (N) and təsdiqi (A), as in (1), although it attaches only to nouns. The formations verify its productivity and the adaptability of the Arabic loans.

The adaptability is also often seen with Persian affixes which generally form adjectives. There may be even more formations with the Persian -ana. However, there are not a variety of formations. Arabic qualitative nouns, often treated as adjectives, e.g. afiq 'lover' as in (2), can be converted into adjectives i.e. afiqana. The base nouns describe the characteristics of a person, but their derivations modify the action. Only some derivations by the phonological changes in the bases show their adaptability with some affixes, but many others do not. For example, zəlilana 'base' is possible, but \*zıllətana is ungrammatical, although the formation with -amez instead may be grammatical zıllətanez 'disgraceful'.

The use of Arabic affixes here is also not the same as the way they are used in Arabic loanwords. For example, -a(h) is generally used as a loan feminine marker in Urdu, but presently it functions as a derivational affix and derives nativised adjectives from adjectives as in (7). This sort of affixation sometimes also causes semantic differences. These Arabic derivations taking the Arabic affixes -a(h), -ija(h) and -ijat are though even less frequent than those with Persian and native affixes, they derive formations that are unknown in Arabic morphologically or semantically. For example, the loan mosaddiq is originally used as a derived Arabic noun/adjective in Arabic, but it is used as a male proper noun in Urdu. The affixation of -a(h) converts it into an adjective i.e. mosaddiqa(h) 'verified', which is also not used in Arabic. The former is not common in Urdu, unlike the latter that is frequent. Of the other Arabic affixes, -ijat forms the nativised nouns e.g. masumijat 'innocence' in combination with the loan derived adjectives i.e. masum 'innocent'. The other -ija(h) derives adjectives from nouns ifajaja(h) 'of love'.

These nativised Urdu formations show some common features in the nativisation process. First, they are only possible when some Arabic loan derivations are adaptable enough to adjust with native and the loan affixes in Urdu. However, not all derivations are adaptable and the affixes cannot attach to them freely. The changes by the affixes show some rules, but the patterns are sensitive to both structure and semantics. There are various other factors involved. It cannot be predicted which affixes attach to which loanwords. For example, -i attaches to an abstract noun, e.g. zolm 'tyranny' and forms mainly noun and adjective zolmi 'cruel man', but with other abstract nouns, e.g. təsdiq 'verification' it may form only an adjective, e.g. təsdiqi 'verified'. Sometimes, the semantics of the (Persian) affixes also affect the formation. For example, -ana describes the way something is done, but -foga describes the

completion of a job. Therefore, both attach only to the loans relevant to their semantics. From this perspective, formations like \*zalimfoda and \*təsdiqana are ungrammatical.

Although the affixation with the Arabic loan derivations by the phonological changes in bases is not on a large scale, the formations neither conform to the source language rule nor to the target language rule. Therefore, they are on peripheral target language patterns. The affixation with these derivations shows some irregularities. Thus, they deviate from the general native Urdu and all the loan patterns. There are certainly some rules in native Urdu, but the deviations cannot be ignored particularly when the loanword morphology interlinks with the native morphology. This is because rules from the three sources of words function together.

The three sources have formed some patterns of derivations which sometimes accommodate each other's affixes but which at other times do not. Arabic loanwords, particularly the derivations by phonological changes, show adaptability, and so they take affixes from the various sources. As Dressler (2003: 39) points out, Wurzel (1984) considers that adaptability shows fitting properties of loanwords with the native affixes, and so their integration into native morphological systems occurs. Arabic loanwords are well-suited in the Urdu derivational system, as they have fitting properties. The derivational affixes from the three sources are productive enough to accommodate them without requiring any major changes, i.e. deletion or substitution of a vowel. This shows that Arabic bases can be extended by the addition of new items (the Urdu derivational affixes in general), and therefore seem to have been fully integrated into the Urdu derivational process.

Arabic loanword derivations in sum are in two forms, by affixation and by modification of base. Both loan types are affected by phonological changes e.g. neutralisation or rise and fall of Arabic phonemes in Urdu. It is necessary to discuss some phonological changes also, as

they often interact with morphological changes. Arabic loan affixes are however less affected by them. They are not as many as Persian and the native Urdu affixes, and the derivations from them are also fewer, they show a variety of derivations. Most commonly used affixes are the negative markers  $y \alpha r$ ,  $la \ bila$ , the definite article  $\partial l/\partial l$  and the adverbial  $-\partial n$ .

There are only a few patterns of loan derivation by the phonological changes in bases, but they are the same as in the original language. However, they are more affected by nativisation than the derivations by affixation. They take various suffixes without any distinction of the three sources in focus i.e. native Urdu, Persian and Arabic. The affixes from these sources can attach to the derivations by phonological changes. This shows that Arabic loan derivations by the phonological changes in the bases are adaptable and that some affixes from the three sources are productive. This is why interaction of both occurs and further words are derived.

Although the patterns of derivation by phonological changes are the same in Urdu, with some changes in the Arabic loanwords phonology, some Arabic phonemes function in Urdu the same as the native Urdu phonemes. Therefore, it is imperative that there be a brief glance at Arabic loanword phonology.

## 3.4.2. Arabic Loanword Phonology in Urdu

Urdu has various phonemes that are also available in many other languages spoken in the world. Many of the segments found in Urdu are loans and to a degree Urdu thus shares a segment inventory with the languages it borrows from, i.e. in this instance Arabic. As Watson (2002: 1) points out, Semitic languages (including Arabic) are marked by a limited vocalic system and rich consonantal system e.g. uvular, velar fricatives, pharyngeal, and pharyngealized (emphatic) sounds. Urdu has borrowed a large number of words in whom some consonant sounds e.g. uvular and velar fricatives have been retained while many others,

e.g. pharyngealized (emphatic) sounds, have lost their phonetic value and are substituted by the native sounds. Despite the sound changes, all loanwords can be recognised both by the mutual segment inventory and by common words in the two lexicons. Interestingly, children learning the Quran get special training for the correct pronunciation of words, which they ignore when speaking Urdu. This is because the same is the case with religious scholars using original Arabic words in speaking Arabic, but using the same as loans in Urdu. Such as they correctly pronunce emphatic dentals in Arabic, but they ignore this while speaking Urdu.

Naim (1999) notes, most Arabic loans are easily recognisable due to dentals and emphatic dentals. They contrast in Arabic, but there is no difference perceived in Urdu relative to these sounds. For example, words containing  $\frac{1}{2}/t$  are treated the same as those with  $\frac{1}{2}/t$ . Similarly, other words containing  $\frac{1}{2}/t$  and  $\frac{1}{2}/t$  are treated like those with  $\frac{1}{2}/t$  with  $\frac{1}{2}/t$  with  $\frac{1}{2}/t$  with the following vowel. This shows the phonological adaptation of the words, which are different from how they originally were in Arabic.

# 3.4.2.1. Consonantal Integration

Some Arabic loan phonemes are accommodated in the Urdu sound system. There are many sound segments, which do not show changes and are adopted without any modification. However, many times the loanwords undergo important alternations of sounds and the foreign sound segments are replaced with more 'native essences'. Hardie (2003: 34) observes that Urdu's consonant inventory is somewhat different from that of Arabic; distinctions that existed in Arabic are neutralised (the sounds lose phonetic value) disregarding the position whether word initially, medially or finally. The neutralisation of the loan sounds is mainly seen in the dentals and emphatic dentals, as illustrated in the following table:

Table 6: Arabic Loanword Phonology in Urdu: The Change of Emphatic Sounds

Phonetic Change	Arabic Origin	Urdu Adaptation	Gloss
1. /d/ → /z/	ħa <b>d</b> ır	hazır	present
$2. \ /\dot{z}/ \rightarrow /z/$	<b>ż</b> alım	<b>z</b> alım	tyrant
$3. \ /\delta/ \rightarrow /z/$	ðıkər	zıkər	mentioning
$4. /\theta/ \rightarrow /s/$	varı <b>0</b>	varirs	heir
5. /ṣ/ → /s/	xaş	Xas	special
6. /t/ → /t/	ṭəl <i>a</i> q	təl <i>a</i> q	divorce
7. /ħ/ → /h/	ħadır	<b>h</b> azır	present
$8. / \$ / \rightarrow \text{ following vowel}$	<b>C</b> ılm	ılm	knowledge
	fa <b>\$</b> al	fa'al	Active

The most important point is to see the substitution of Arabic emphatic dentals by native sounds. The native Urdu alveolar fricatives /z/ and /s/ substitute for two or more loan emphatic sounds. The Arabic dental  $/\dot{d}/$  is neutralised and substituted by the alveolar voiced fricative /z/, as shown in (1). Therefore, words like  $ha\dot{d}ur$  'present' are pronounced as hazur. The alveolar fricative /z/ also substitutes for voiced strident fricative  $/\dot{z}/$  e.g.  $\dot{z}alum$  'tyrant' (2) and dental fricative  $/\dot{\partial}/$  e.g.  $\dot{\partial}ukar$  'mention' (3). Thus, the two words are pronounced as zalum and zukar respectively. The voiceless alveolar fricative /s/ also substitutes for two loan phonemes i.e. voiceless dental fricative  $/\theta/$  and the emphatic sounds /s/. For example, words like  $varu\theta$  'heir' (4) and vas 'special' (5) are pronounced as varus and vas respectively. The last emphatic dental sound /t/ as in vas (6) is also changed with the voiceless dental stop /t/. The word is now pronounced as vas (6) is also changed with the voiceless dental

The Arabic voiceless pharyngeal fricative  $/\hbar/$  (7) does not exist in Urdu. It is pronounced as aspiration /h/. Thus, there is no phonological difference between /h/ and  $\hbar$  in loans like hadi 'leader/guide' and  $\hbar adj$  'pilgrim'. Both words begin with /h/.

Finally, Arabic voiced pharyngeal fricative /S/ (8) is often considered a controversial subject. It is mostly deleted and the loanwords are pronounced with the following vowel. Hardie (2003: 34) observes that f' is not found or pronounced as a glottal stop, as zero, as  $\bar{a}$  or a, depending on various factors including its environment. The Urdu pronunciation of most loanwords shows that it is deleted and loans, e.g. Somer 'age' are pronounced with the following vowels /a/, /v/, /ı/, /ə/ and /v/, instead of, i.e omer in this case. Paradis and LaCharité (2001: 258) state, "The non-availability of the Pharyngeal node also explains systematic deletion of the pharyngeal and laryngeal gutturals in Arabic loanwords in French." The same non-availability of the pharyngeal node also results in systematic deletion of the pharyngeal gutturals e.g. /S/ in Arabic loanwords in Urdu. Even if /S/ is accepted as the glottal stop, the presence of glottal stop in Urdu is itself an issue of argument. Some writers, e.g. Hussain (2004: 3 & 5), (Nawaz 2002: 227) and Shahid (2002) agree that it is controversial whether glottal stop exists in Urdu or not. Shahid (2002: 6) comes to the conclusion that no rule completely describes the existence or removal of the glottal stop, while Hussain admits that its existence is a controversial matter and a point of open debate<sup>2</sup>. It is generally neutralised, does lose its phonetic value and replaced by following vowel.

Neutralisation however does not mean that all the phonemes have lost their phonetic values. Some Arabic sounds, e.g. velar fricatives /x/ and  $/\gamma/$  are a permanent part of the Urdu phonetic inventory. They substitute for native phonemes in native words. Moreover, they also create

.

<sup>&</sup>lt;sup>2</sup> Email correspondence (November, 2007)

new lexical items in combination with native phonemes. However, as recently seen, there seems a reversal of the sounds and some of them are confused with native phonemes.

#### 3.4.2.2. Rise and Fall of Arabic Loan Phonemes

Although some Arabic phonemes are neutralised, and they lose their phonetic value, many others act like native phonemes and are commonly used in native Urdu words e.g. tarxa 'reject humbly' instead of tarka. The recent data nevertheless show that some of them are substituted by native sounds. Thus, there appears to be a rise and fall of Arabic loan phonemes, which is the focus of this section. The rise of Arabic sounds here means that they act as if they were the original native sounds. Some Arabic loan phonemes substitute for native Urdu phonemes, e.g. the Sanskrit origin word tarka 'branch' becomes tarka in Urdu. The aspirated velar stop is thus substituted by the velar fricative i.e. tarka in Urdu. The Although Hindi and Urdu are both treated as the same language by linguists, some lexical, morphological and phonemic differences however exist. Looking at this view, tarka becoming tarka does not seem to be merely hypercorrect pronunciation. Both voiceless and voiced velar fricatives tarka and tarka in fact tarke the status of normal Urdu phonemes and function the same as original native phonemes. They do not merely cause the substitution but also the creation of new lexical items e.g. tarka 'firecracker'. Thus, their role is distinctive.

In recent years after the rapid growth of electronic media in the Indian subcontinent, it is observed that some loan phonemes sometimes undergo a reverse process. Although on rare occasions words like *xali* 'empty', in normal Urdu, are pronounced as  $k^hali$ . This is a reverse process  $/x/ \rightarrow /k^h/$ , however it may be small. It is also possible that  $k^hali$  existed all along, but

was restricted to certain regional/social varieties, and is now being used more widely. In such a case, the variation between /x/ and  $/k^h/$  from the start cannot be denied.

However, certain other examples show that a process of reversal of sounds is going on. Phonetic reversal is the process of reversing the phonemes of a word. Wohlin (2010: 38) notes this in Indo-European and Germanic languages, from plosives to fricatives to plosives again.

# 3.4.2.2.1. Substitution and Creation Processes

The substitution and creation as well as the reversal are interesting and are a contrast to each other, and therefore some discussion is necessary. Consider the proess in the ctable below:

Table 7: The Substitution of Native Phonemes by Arabic Phonemes Urdu

Change	Example	Word origin	Word used in Urdu
$1. k \to x$	∫əla <b>k</b> a	Sanskrit	səlax 'rod / iron bar' (N)
$2. k \rightarrow x$	tın <b>k</b> rək <i>a</i>	Sanskrit	təxna 'ankle' (N)
$3. k^h \rightarrow x$	∫akʰ	Skrt/(via)Pers.	∫ax 'branch' (N)
$4. \text{ k} \rightarrow \text{x}$	bətək	(via Persian)	bətəx 'duck/swan' (N)
$5. \text{ k} \rightarrow \text{x}$	tərk <i>a</i>	Hindi	tərxa 'reject humbly' (V)
$6. \text{ k} \rightarrow \text{x}$	tıt <b>k</b> ari	Hindi	tıtxari 'clacking of tongue' (N)
7. /x/	pəta <b>x</b> a	Urdu	pəţaxa 'firecracker'
$8. k \rightarrow x$	t∫əţəkni	Hindi	tsətəxni 'bolt to lock the door' (N)
$9. \text{ k} \rightarrow \text{x}$	t∫əţək	Hindi	tsətəx 'crackle' (V)
$10.k \rightarrow x$	pətək	Hindi	pətəx 'throw down violently' (V)
$11.\int \to x$	∫ʊ∫k	Sanskrit/Ostai?	xυʃk 'dry' (A)
$12.k \rightarrow q$	<b>k</b> əl <i>a</i>	Sanskrit	qəlabaz 'Juggler'(N)

13. none	<b>d</b> a <b>k</b>	Prakrit	dak 'mail' (N)
14. none	kəsək	Prakrit	kəsək 'pain or anxiety' (N)
15. none	la <b>k</b> <sup>h</sup>	Prakrit	lakh '100,000' (N)

12: The word kəla, common in Hindi, is used in Urdu as qəlabaz with only the Persian suffix baz.

The sound changes are historical and mostly occur in nouns and then in base verbs and adjectives. Although there are other changes also, one is stable i.e. velar stop /k/ or sometimes, aspirated velar stop /k<sup>h/</sup> is substituted by velar fricative /x/. It remains consistent on the same pattern. In some rare examples, post-alveolar fricative /ʃ/ is also substituted by velar fricative /x/ i.e.  $\int \rightarrow x$  (11) or by alveolar fricative  $\int \rightarrow s$  (1), but it is unimportant. Various other changes do not seem to show a general rule because the changes are not with relevance to, or affected by, the neighbouring segments. Sometimes, there are also alternations of vowels along with the substitution of phonemes but none are consistent, which may be called the cause of the native-loan sound swap. The only morphemes consistently appearing in some cases are tfat / pat at the prefix position in Hindi origin words (8-10).

-

<sup>&</sup>lt;sup>3</sup>. A dictionary produced by CRULP (Centre for Research in Urdu Language Processing)

Thus, the origin of a word does not affect the substitution process much. A few words seem to have entered Urdu via Persian, i.e. Sanskrit  $\rightarrow$  Persian  $\rightarrow$  Urdu, but the changes are the same, e.g. fax (3) and solax (1). This is interesting, as Sanskrit words entered Urdu either directly or via Hindi. It may be claimed that the one in (3) is a simple loan from Persian, or from Arabic via Persian, but the other in (1) shows substitution of  $far{k}$  by  $far{k}$  in the native words.

There is no transformation of sounds except for a few, e.g.  $b \rightarrow k \rightarrow b \rightarrow k \rightarrow k$  'duck / swan' in (4). The example exhibits two contrasting processes. The Arabic emphatic dental /t/ is substituted by the Urdu dental /t/ word medially. Although it is Urdu sound, the velar stop /k/ is also substituted by the loan velar fricative /x/. Similar changes are also noteworthy in (11), in which /x/ substitutes for the native post-alveolar fricative /ʃ/ i.e.  $f \circ f k \rightarrow x \circ f k$  'dry', this time, but the velar stop remains unchanged. Moreover, /ʃ/ is substituted by /x/ word initially but is retained word medially. However, such contrasting processes are rarely seen.

Some words, e.g. *qəlabaz* 'Juggler' (12), show phonological shuffling, while also showing a mixture of morpho-phonemic processes. Their examples are however rare. The baseword *qəla* 'trick' is a phonological reshaping of Sanskrit *kəla*. Although both *qəla* and *kəla* are obsolete, the derivation *qəla<u>baz</u>* is commonly used. It shows the substitution of native velar stop /k/ by the Arabic uvular stop /q/, but also shows lexical creation by a suffix *-baz* (from Persian *baxtən*), which is nevertheless not the point of discussion at this stage. Like various other examples, no specific rule can be derived here.

Shapiro (2003: 261), terms the substitution process *hypercorrect pronunciations*. However, he does not discuss in detail and only notes that in some instances they "are heard in which Hindi words are pronounced with loan phonemes, as though the items were foreign borrowings e.g. *fir* 'then' instead of *phir*." Urdu perspective is quite different from this. Arabic /q/, /x/ and /y/

may originally be loans. Nevertheless, they are part of the Urdu phonetic inventory, whether or not they are present in Hindi. This is evident in some words, e.g. pətaxa 'firecracker' (7). The example shows that two contrasting sounds i.e. the native retroflex stop /t/ and the loan velar fricative /x/ create a new lexical item. However, the creation in general is not on a large scale. It is even less by /x/ and /t/, but more by the voiced velar fricative /y/ in association with the native retroflex stops /t/, /d/ and /t/:

Word	Origin		Use in Urdu
15a. yət yət	(Urdu)	-	yət yət 'successive gulps' (N)
b. yʊrrahəţ	(Urdu)	-	yorrahət 'sound made by a beast in anger' (N)
с. үәрәр	(Urdu)	-	yərəp 'a sound made by drowning/sinking' (N)
d. <b>y</b> ʊn <b>d</b> <i>a</i> (	Sanskrit ???) <sup>4</sup>	-	งูซท <b>d</b> a 'scoundrel' (N)

All creations by the combination are nouns. Although the number of creations by each retroflex individually may be very small, the loan fricative actively participates in the creation. It always appears word initially. Most formations are with /t, but there are also examples with flap /t/ e.g.  $y \rightarrow t \rightarrow t$  a sound made by drowning/sinking' (15c). The formation by /t/ and /t/ in  $y \rightarrow t \rightarrow t$  are also examples with flap /t/ e.g.  $y \rightarrow t \rightarrow t$  as a creation or a substitution. The expression of a distinctive meaning in this creation is rare, but most creations contain similar semantic expressions and refer to the names of a specific sound e.g.  $y \rightarrow t \rightarrow t$  'successive gulps' or the factors producing these sounds e.g.  $p \rightarrow t \rightarrow t \rightarrow t$ 

The examples illustrate that some Arabic loan phonemes do exist and function like normal native phonemes in the Urdu phonetic inventory. However, the substitution of native

-

<sup>&</sup>lt;sup>4</sup>. This is according to CRULP's spelling. I assume, /γ/ must be either /g/ or /k/ here, as /γ/ is not a Sanskrit sound. If this is the case, it must be a native Urdu word, I have pointed out this to CRULP. If the sound is really /γ/, the word must be a lexical creation within Urdu.

phonemes is far more frequent than the creation of new words. In either case, the loan phonemes show their presence.

Although Urdu and Hindi are considered to be the same language by linguists, they exhibit lexical, morphological and phonemic differences. Hindi is highly Sanskritised, and Arabic phonemes in a large number of loans are replaced by native Hindi phonemes. So, words like *yolam* 'slave' and *xadım* 'servant' are used as *golam* and *khadım*. Therefore, from Hindi perspective, Shapiro (2003: 261) may be right in his argument, as the loan phonemes are not used even in Arabic loans. But the phonemes are part of the Urdu phonetic inventory. They are commonly used. They not only substitute for native phonemes in native words but also create new lexical items. It is not merely hypercorrect pronunciation, but a higher function instead. It should be termed a diachronic renovation of Urdu lexicon.

However, in recent years, some surprising observations reveal that there is a reversal of the loan phonemes, although on a very small scale. Arabic loan sounds i.e. /q/, /x/ and  $/\gamma/$  are substituted by the native sounds /k/,  $/k^h/$  and /g/ even in Arabic loanwords. The loans, e.g. xali 'empty' are pronounced as  $k^hali$ . This is interesting and does require attention, because this is in sharp contrast to what we have noted, namely, that Arabic sounds entered the Urdu phonetic inventory as a result of diachronic process.

# 3.4.2.2.2. Reversal of Arabic Phonemes: q/k, x/kh and γ/g

Generally, Urdu tends to accommodate loanwords, following some adaptation rules like other recipient languages. The heavy influx of some loan types has caused the Arabic phonemes, e.g. /q/, /x/ and  $/\gamma/$ , to enter the Urdu phonetic inventory. However, as noted in recent years, some people are confused when using /k/,  $/k^h/$  and /g/ for /q/, /x/ and  $/\gamma/$  respectively and cannot differentiate between them. There may be some other factors, but they are particularly

influenced by a flood of Indian movies and TV serials. People's confusion between uvular and velar stops /q/ and /k/ seems very easy. Although different from each other, both the sounds are very close in their place of articulation. There is also a reversal of the velar fricatives /x/ and / $\gamma$ /. They seem to not only be substituted by the native velar stops /k/, /k<sup>h</sup>/ and /g/ in the native words they entered initially but also in the Arabic loans they originally belong to. Consider the table illustrating some examples:

**Table 8: The Reversal of Arabic Loan Phonemes** 

<b>Sound Change</b>	Word Origin	1	Urd. Adaptation	Glossary
$1.  q \to k$	qatıl (N)	(Arabic)	<b>k</b> atıl	killer
$2. q \rightarrow k$	mə <b>q</b> sə <u>d</u> (N)	(Arabic)	mə <b>k</b> sə <u>d</u>	aim
$3. q \rightarrow k$	rız <b>q</b> (N)	(Arabic)	rīzk	subsistence
$4. x \rightarrow k^{h}$	tʃətəx (V)	(Native)	tʃətəkʰ	crackle
$5. x \rightarrow k^h$	xali (A)	(Arabic)	k <sup>h</sup> ali	empty
$6. \gamma \rightarrow g$	yət yət (N)	(Native)	gət gət	straight gulps
$7. \gamma \rightarrow g$	baγ (N)	(Arabic)	bag	garden

Note the three nouns (1-3) showing  $/q/ \rightarrow /k/$ . There is no difference in the appearance of /q/ whether word initially, medially or finally. Although their place of articulation is different, the two sounds are close to each other, and so some people are confused by them. The words, e.g. qatil 'killer', are pronounced with /k/as katil.

The case of  $/\gamma$ / and /x/ is important, as they previously substituted for native phonemes and on a certain scale are also used along with native Urdu phoneme in newly created lexical items. Unlike the case of their entrance, there is no difference in their appearance in any position, for the reversal. Voiceless velar fricative /x/ is particularly notable. It diachronically substituted

for the native velar stop /k/ in some native words, as in  $t \int \partial t \partial x$  'crackle' (9) in Table 4. In some words, it also replaced /kh/ and /ʃ/. However, in the reverse process, it is now exchanged with the native aspirated voiceless stop /kh/ only, as in (4) Table 5. Moreover, /x/ is replaced by /kh/ even in the Arabic loans e.g. (5), which is however a natural process. The voiced velar fricative / $\gamma$ / is a little different. It diachronically created some words e.g.  $\gamma \partial t \gamma \partial t$  'successive gulps', as in (10). However, the same words are now pronounced with voiced velar stop /g/ as in  $g\partial t g\partial t$ . Moreover, Arabic loans are also used with /g/ as  $b\partial t \partial t$  'garden' (7) in Table 8.

It is often observed in the conversation of some people on TV street shows. It is also noted in the communication between my wife and daughter of 14. There is nevertheless no evidence available in literature, as the reversal is not on a large scale. Note that the Arabic phonemes are confused with the native ones, and their reversal from Urdu is noteworthy.

Wohlin (2010: 38) notes a similar reversal of sounds in the Indo-European and Germanic languages, e.g. from plosives to fricatives to plosives again. He states that a difference of phonemes for the same words between neighbouring ethnic groups could be interpreted as a manifestation of a polarizing force, whose counter force is the reciprocal loans. He terms counter force "the self-preservation" of a language and sees a historical correlation between the development of phonemes and the social divergence-convergence. The geographical spread of humans happens together with the spread of phonemes from plosives to fricatives, voiceless or front phonemes, which then turn to the original phonemes, when some groups turn back towards the areas of their language. Citing Grimm's (1827) and Verner's (1875)

\_

<sup>&</sup>lt;sup>5</sup>. A sound law, first formulated by German philologist Jakob Grimm in 1827, relating German consonants in other Indo-European languages. It states a chain reaction: aspirated voice stops become regular voiced stops. voiced stops in turn become voiceless stops, and voiceless stops become fricatives.

<sup>&</sup>lt;sup>6</sup>. Karl Verner (1875), Danish philologist, explained a category of seeming exceptions to Grimm's Law and observed that voiceless fricatives in Germanic became voiced if the preceding syllable was unstressed.

laws, he states that linguists talk about the circular pathway in the displacement from plosives to fricatives to plosives again in the development of Indo-European → Germanic languages; perhaps this is just one example of how more original forms of words may reappear.

Of the two possible causes, Wohlin (2010: 38) points out, i.e. the neighbouring communities and historical development, the former (via electronic media) seems to influence the reversal of the Arabic loan phonemes in Urdu. However, the extent of the sound shift that he sees in the Indo-European and Germanic language families cannot be observed here. It is limited to a few specific cases and the shifting back is also not the same, as shown below:

Loan	Example	Reversal	Example	Glossary
16a. k $\rightarrow$ x (Substitution	on) tʃətək / tʃətəx	$x \to k^{\text{h}}$	tʃəţə $\mathbf{k}^{ ext{h}}$	crackle
b. $\rightarrow \gamma$ (Creation)	γəţ γəţ	$\gamma \to g$	<b>g</b> ət <b>g</b> ət	successive gulps
c. $k \rightarrow q$ (Substitution	n) <b>q</b> əla <i>baz</i>	$\mathbf{q} \rightarrow \mathbf{k}$	<b>k</b> əla <i>baz</i>	juggler

There is some difference in shifting back shown in (16a), although (16a-16b) seem the same. The loan voiceless velar fricative /x/ substituted for the native voiceless velar stop /k/ in the past generally but the shift back shows its substitution by the aspirated voiceless velar stop /kh/. Therefore, there is an obvious divergence in the reversal process. Secondly, the voiced velar fricative / $\gamma$ / created some lexical items but is replaced by the native voiced velar stop /g/ in the same words, as in (16b) and in Table 5. But, this may be expected as a natural process. Similar to this, (16c) shows the loan uvular stop /g/ substituted by the native velar stop /k/.

Although there are some differences, the shifting back of the phonemes in Arabic loanwords is noteworthy. Therefore, to an extent, the data seem to support Wohlin's (2010: 31) claim regarding Uralic languages that most sound shifts occur between phoneme types within the

group with the same articulation site: Velar plosives  $\leftrightarrow$  velar fricatives, alveolar plosives  $\leftrightarrow$  alveolar fricatives; the sound shifts within a group of tongue sounds appear as natural process.

From the Urdu perspective, some differences are noted. There is a shifting back to native velar stops from loan velar fricatives, but it shows aspiration in the voiceless stop i.e.  $x \to k^h$ . Moreover, there is a difference in groups of sounds shifting /q/ to /k/, the former is uvular and the latter is velar, in contrast to Wohlin's observation 'phoneme types within the group'.

A third difference is the cause of the rise and fall of Arabic loan sounds. Wohlin's (2010: 38) hypothesis suggests that the geographical spread of humans is the major cause of the rise of loan phonemes in Germanic and Indo European languages; when some groups turn back towards the areas of their language, the shift back also occurs and the loan sounds are substituted by the original sounds. Although he agrees that various examples may contradict this hypothesis, he emphasises that it could be valid even if it is only one out of 10 ratios. In contrast, it was the geographical spread of Arabs, instead, which caused the introduction of Arabic loan sounds in native Urdu phonetic inventory. In this context, the grounds of the shift back also differ from Wohlin's (2010: 38) hypothesis. It is not the return of native speakers of Urdu, but the strong influence of media instead.

The hypothesis suggests that the different social and geographic conditions involved in the reversals of the loan sounds in some languages could also be involved in the reversal within a word. Reversal within a word conveys that there is sometimes an alternation of two neighbouring consonants which swap with each other. He states that reversal of phonemes adjacent to or away from each other, called inter version or metathesis, is a sound change interpreted in terms of switching directions. A slip of the tongue in children's language e.g. Swedish  $br\ddot{a}da$  'board'  $\rightarrow b\ddot{a}rda$  and correspondingly English  $broad \rightarrow board$  could be good

examples. Although not very frequent, some Arabic loans show similar metathesis i.e. the consonants swapping may be adjacent or a vowel may be in-between them, as shown below.

Example (17a) shows a simple metathesis. The fricative /f/ and the lateral /l/ have diachronically swapped positions within the Arabic loan. The word was actually spoken as  $q \sigma f l i$  initially but after the swapping of the two sounds the word is now pronounced as q u l f i. Thus the word index has changed from  $C_1 V_1 C_2 C_3 V_2$  to  $C_1 V_1 C_3 C_2 V_2$ .

Example (17b) is different and shows a morpho-phonemic process. The metathesis is the result of the affixation of the Persian negative marker na, and the consonants /z/ and /r/ are not swapped diachronically. Moreover, there is a vowel in-between, and it also changes from /e/ to /i/, although the position remains intact. This confirms Wohlin's (2010: 30) observation that the consonants swapping may be adjacent or distant from each other. The word index thus also shows extreme changes from  $C_1V_1C_2V_2C_3$  to  $C_1V_1C_2V_2C_4V_3C_3$ . The reversal of the sounds and the change of the word index is however associated with morpho-phonology and the examples like those following a negative are rare.

The reversal is nevertheless not the dominant feature in standard Urdu. Arabic phonemes are an important part of the Urdu phonetic inventory and distinctive from Hindi phonemes used in the same words. Like the substitution of the native phonemes and the creation of new words they undergo various other phonological processes e.g. assimilation, elision, gemination.

#### 3.4.2.3. Assimilation, Elision and Gemination

A sound is assimilated by adopting some feature(s) of an adjacent sound. Elision is simply the deletion of sound due to a certain process, while gemination may occur as a result of elision. Although the three processes are frequent in Arabic loanwords, they are restricted to a few specific loans. Therefore, the discussion in this section is restricted to some major points.

The place assimilation rule for nasals (found in many languages like English, Spanish and Urdu with a little variation) states that a nasal has the same place of articulation as the following consonant (Shah, 2002). In Arabic loans in Urdu, assimilation is generally seen in the nasal /n/ and the lateral /l/ sounds. It is influenced by some following consonants. For example, when the nasal /n/ precedes bilabial voiced stop /b/, it gets labialised and becomes /m/ i.e. /n/  $\rightarrow$  [m]/ \_b e.g.  $\alpha$ nb $\alpha$ r 'a rich perfume'  $\rightarrow$   $\alpha$ mb $\alpha$ r. The alveolar nasal /n/ loses its phonetic value, as the place of its articulation is bilabial due to the following bilabial voiced stop. Therefore, the loan is pronounced differently.

In contrast, elision is the complete omission of a sound due to the effect of a following consonant. This is often seen in the Arabic loan definite article  $\partial l/\partial l$ , when the lateral /l/disappears, while the following consonant is geminated e.g.  $nur\sigma(l)ddin$  'the light of the religion'. The article generally occupies word initial and medial positions and functions in connection with some consonants divided into two groups based on whether or not /l/ is geminated with them. Although the Arabic distinction in terms of the 'solar letters' /t/l, /t/l

not. Since the article  $\partial l/\partial l$  ends in a coronal consonant, it loses phonetic value with these sounds. Following the Arabic rule, the lateral /l/ is elided in Urdu also, when followed by solar letters, e.g.  $nur\sigma(l)ddin$  'the light of the religion' and is not otherwise e.g.  $nur\sigma(l)ddin$  'the light of the book'. However, in many other loanwords, the lateral /l/ in the Arabic definite article shows phonological differences with respect to the position of its appearance in the words and compounds. Unlike in Arabic, there is a restriction of position in Urdu and the elision and gemination occur only word medially. The process does not occur word initially, although the following consonant may be a solar letter. For example, the Arabic rule does not apply when any noun begins with the Arabic loan definite article. Consider the following:

18a. kɪt̪abʊ[1]**n**nur

b. əlnur

c. \*ə[1]nnur

The application of the Arabic rule is possible only when the definite article appears word medially, as in (18a). The lateral /l/ is elided as it precedes a solar sound /n/ which is geminated, and thus the word is grammatical. But the elision does not occur if the loan Arabic definite article appears word initially as in (18b), although it may be ungrammatical in Arabic. The elision the lateral /l/ and the germination of the following nasal are in fact ungrammatical in this position, as shown in (18c). This is in sharp contrast to how it is used in the source language. Elision and gemination are not features of Urdu. Therefore, a rule may be applied in one case (18a), but it is not applicable in other acse (18b- 18c).

In short, in addition to the neutralisation and rise and fall of the loan sounds, Arabic loanwords show assimilation, elision and gemination of the sounds. Both assimilation and elision are caused by the effect of the following consonants but the two rules function in

different environments and have different results. For example, the nasal /n/ is assimilated and becomes /m/ when followed by bilabial stop /b/. Elision is the deletion of a sound, e.g. lateral /l/ of the loan definite article  $\partial l/\partial l$ , if it is followed by a solar letter. The elision rule in Urdu is functional only if  $\partial l/\partial l$  appears word medially. As a result of the elision the following consonant is geminated i.e. doubled e.g.  $s\partial lah\partial [l]ddin$  (proper noun) and  $ktdab\partial [l]nnur$  'book of light'. The process of gemination is not found in native Urdu words.

However, the elision and consonantal doubling are specific with only the Arabic loan definite article, not in any native Urdu or Persian loan affixes. They are among the major aspects of the consonantal sound changes in the loanwords. Although the consonantal sounds show more important features the changes are less than those in the vocalic sounds. Sometimes, the deletion of a consonant results in vowel changes, e.g. vowel lengthening or creation of diphthongs in the loanwords. Vowel changes may occur in various places.

### 3.4.2.4. Vowel Integration in Arabic Loans

Change of vowels in the loanwords is a general phenomenon and can be seen in the Arabic loanwords as well. However, considering the morphological focus of this work, this section discusses briefly some major aspects e.g. vowel lengthening, shortening caused by the deletion of pharyngeal fricative, by the pluralisation and by derivational phenomena.

# 3.4.2.4.1. Vowel Changes Caused by the Neutralisation of Pharyngeal Fricative /5/

As noted in the beginning of Section 3.4, many of the Arabic loan sounds are neutralised and lose their phonetic value. The neutralisation of pharyngeal fricative  $/\mathcal{G}$  causes a word to be pronounced with the following vowel, which may take one of three different forms. It may lengthen, become a diphthong or retain its original form. For example, when a loanword, e.g.  $fa(\mathcal{G})al$  'active' is neutralised, the word is often spoken with a long vowel as fal, but when  $/\mathcal{G}$ 

in between two different vowels e.g.  $fa(\mathfrak{S})l$  'subject/agent' is deleted, there is the creation of a diphthong, as [al] in fall. Languages differ in the length of diphthongs. In languages with phonemically short and long vowels, diphthongs typically behave like long vowels. Waqar and Waqar (2002: 20) agree that the deletion of some consonants i.e. glottal stop /?/, palatal approximant /j/, labiodental /v/ and pharyngeal fricative / $\mathfrak{S}$ / results in the formation of diphthongs whose most underlying vowels are long. This means that the deletion correlates with the lengthening of the vowels, which are sometimes diphthongs. However, there are certain cases in which the first vowel may be long while the second is generally short if taken individually. For example, the word \*fail is ungrammatical if both the vowels are long.

Moreover, neutralisation of  $/\Omega$  does not always result in the formation of a diphthong or the lengthening of a vowel. Sometimes, a vowel may remain unchanged.

19. (
$$\S$$
) rlm  $\rightarrow$  rlm 'knowledge'

There is neutralisation of  $\fineskip \fineskip \finesk$ 

20a. (?) alim 'scholar' 
$$\rightarrow$$
 aalim

b. (f) 
$$i f q$$
 'love'  $\rightarrow i f q$ 

c.  $ma(s)_{ija\underline{t}}$  'companionship'  $\rightarrow ma_{ija\underline{t}}$  'companionship/funeral'

The lengthening of the vowel only occurs when  $/\varsigma/$  appears with /a/ irrespective of the position, whether word initially as in (20a) or word medially as in *faal*, noted above. Otherwise, no vowel changes occur word initially, as in (20b), and there is a formation of a diphthong word medially, as in (20c). The last change shows an interesting element. After the neutralisation of  $/\varsigma/$ , there is no difference left in the pronunciation of the word with that of another word having the meaning 'funeral'.

Therefore, this sort of change sometimes causes confusion. Although the examples show certain regularity, the three results of the same consonantal disappearance confirm the complexities of phonological changes in the various patterns of the loanwords. These intricacies can be observed also in the shortening vowel, explained in the next section.

### 3.4.2.4.2. Shortening of Vowels

Vowel shortening in the Arabic loans is mainly the result of the pluralisation of feminine loan nouns with native Urdu plural markers and the formation of adjectives from nouns. If a loan noun ends in -i, it is affixed with the plural marker  $-j\tilde{a}$ . But at the same time-i has to be substituted by the short front vowel -i-. Consider the following examples:

#### Nouns Plurals

21a.  $k \sigma r s i$  'chair' (f)  $\rightarrow k \sigma r s i j \tilde{a}$ 

b. hazri 'presence' (f)  $\rightarrow hazrij\tilde{a}$ 

The noun in (21a) takes the feminine plural marker  $-j\tilde{a}$  while at the same time the long vowel -i is also substituted by the short vowel -i. The morphology here is in conformity with phonology because it is not possible for long -i to precede  $-j\tilde{a}$ , a suffix with a glide on onset position. The noun hazri in (21b) is the natively derived feminine noun from the loan

noun/adjective *hazır* '(someone) present'. The feminine marker -*i* fulfils the requirements of the pluralisation, which ultimately shows the substitution by -*i*-. Naim (1999) states that Urdu long vowels are inherently long, similar to the vowels in *tall* and *lord* respectively and all long vowels are pure, as they are not pronounced with a following glide. The feminine plural marker - $j\tilde{a}$  starts with a glide and for this reason the process of pluralisation requires the short vowel -*i*- which substitutes for the feminine marker -*i*.

The same rule is applied in the formation of nouns from other nouns or adjectives, both of which end in a long -i. The adjectives are themselves nativised forms and originally derived from the Arabic loan nouns, e.g. ilm 'knowledge/information'  $\rightarrow ilmi$ . After the suffixation of -jot, the adjectives become nouns, but there is also a vowel change from long to short, as in ilmijot in this case. The derivational suffix -jot begins with a glide and therefore the rule remains the same. The formation of a noun requires the long -i to change into a short-i-.

The suffix -i also derives nouns from base adjectives e.g. mehrum 'deprived'  $\rightarrow$  mehrumi 'deprivation', which in turn can be transformed into another noun mehrumijot. The process of vowel shortening is the same. This shows that there is some regulation in the shortening of vowel, although it may not be seen in other sound changes.

The discussion on loanword phonology can be summarised as follows; many Arabic loan phonemes are neutralised but some others retain their phonetic value. Some of these latter ones, e.g. /x/,  $/\gamma/$  and /q/ not only substituted for some native phonemes diachronically but also created new lexical items in correlation with them. It cannot simply be termed hypercorrect pronunciation, as Shapiro claims about Hindi. The two processes, substitution and creation, in Urdu show that the three phonemes are part of the Urdu phonetic inventory. However, as observed recently, the loan phonemes shift back not only from the native words

they previously entered but are also substituted by native counterparts in the loanwords, which is not unnatural. The reversal of sounds shows some features different from those noted by Wohlin (2010: 31) in Uralic languages, although data in Urdu supports his observation. There is a shifting back to native velar stops from loan velar fricatives e.g.  $/x/ \rightarrow /k^h/$ , but it also shows aspiration in a voiceless stop. Moreover, the reversal of phonemes occurs in different groups of sounds, i.e. in /k/ and /q/, the former is a velar stop and the latter is a uvular stop. There is also a difference of the rise and fall of the sounds from what Wohlin notes. However, as for the reversal within a word, the data from Arabic loan phonology confirm his claim that two sounds within a word can swap even if they are distant.

# 3.5. Summary of the Chapter

The main focus of the chapter has been to describe derivation by affixation. There are more prefixes than suffixes, but they derive fewer words. Affixes from the three sources in general give various patterns, but frequent deviation from each pattern is also noteworthy. The vast majority of affixes in Urdu come from Persian, though the native affixes are productive and derive most words. Thus, there is a distinction between the number of forms and frequency of use. Native affixes, e.g.-*a* and -*i*, do not express their meaning.

Loan affixes however express various semantics and thus give evidence against his claim. Although Persian loan affixes are the most, they derive only a few words. There are nevertheless some productive affixes, e.g. *na* 'no/not' and *-ana*. The number of derivations from them is lower than that of native Urdu affixes and higher than that of Arabic affixes.

The derivational changes in Arabic loanwords are achieved both by affixation and by the modification of bases. The affixes, e.g. *la* 'no/not' and *yær* 'not/without' are few, but they are productive and derive some words. Like Persian affixes, they express their meanings as well.

But the derivations by affixation are not as many as derivation by vowel changes. The words derived by the modification of bases are large in number and in various patterns, but they are no more than loans in Urdu. Moreover, they are nativised by taking affixes from native Urdu, Persian and even from Arabic. The affixation with these derivations has some irregularities, which show that there is often deviation from the rules particularly regarding loanwords.

The intricacies in the loanword morphology from Arabic are often affected by phonological changes in the Arabic loans. Therefore, it was necessary to have an idea of Arabic loanword phonology, as Arabic phonemes function like native Urdu phonemes and also play a role in the creation of lexical items. The most important feature is the neutralisation and the rise and fall of some Arabic phonemes. Some loan phonemes not only substituted for native phonemes but also created new lexical items diachronically. Therefore, Shapiro's (2003: 261) term *hypercorrect pronunciation* is not justifiable in respect of Urdu. Arabic loan phonemes are important members of the Urdu phonetic inventory and function like native phonemes. However, a reverse process is also seen and borrowed velar fricatives  $\langle x \rangle$  and  $\langle \gamma \rangle$  are being replaced by the velar stops  $\langle x \rangle$  and  $\langle y \rangle$ . The data presented confirms Wohlin's (2010: 38) observation of reversal of sounds, but the features shown in the case of Urdu are different.

The phonological process often interacts with morphology and affects inflectional and derivational morphology. Derivational morphology, by affixation, has been discussed in this chapter. Derivation is not restricted to affixation and many words are derived by compounding also. Urdu compounds have some key features. Most compounds are formed by some distinctive compounding words, while a large number of compounds take infixes. Compound formation is discussed in the next chapter.

# Chapter 4

# **Derivation by Compounding**

#### 4.1. Introduction

Two aspects of the derivation discussed in the last chapter, with respect to the three major sources of words in Urdu, were the affixation of various morphemes to bases and the modification of bases themselves. The derivation of new words is also very frequent by compounding two words having independent lexical status. However, compounding is not necessarily made by two constituents from the same source language. There is a possibility that both constituents are from two different sources. Such compounds are called hybrid compounds. They are heavily dominant in Urdu and more frequent than normal, i.e. native + native, combinations. They form the same types of compounds and show the same features, which normal native based compounds show. Therefore, this chapter will focus on the various types of compounds rather than look at the compounds from a specific source language. These types include, endocentric, exocentric, copulative, postpositional and verbal compounds.

Although these combinations are specific with their features, e.g. an endocentric compound is a hyponym of one of its constituents and a copulative has two independent constituents, they are sometimes ambiguous and confusing. Sogaard (2004: 322) exemplifies that although the Mandarin Chinese compound *fumu* (father-mother) 'parents' is a copulative compound, it can be given the endocentric interpretation 'a mother who acts like a father'. This ambiguity, often noted in Urdu compounds, may be due to the two different source languages of the constituents. For example, when two constituents are synonymous or near synonymous e.g. hoson 'beauty' + dsomal 'beauty'  $\rightarrow hoson-o-dsomal$  '(kinds of) beauties', it is generally an endocentric compound. Nevertheless, constituents are synonyms in many other compounds e.g.  $bal\ batfe$  'family', but the formations may be other types e.g. copulative in this case. The

constituents in hosən-o-dʒəmal are both Arabic, but the constituents in bal bətʃe are from native and Persian sources. The former compound is endocentric and simpler to understand due to the same source constituents. However, the latter compound is copulative and ambiguous, as the constituents are from two sources, although they are synonymous. Thus, the change of the source language of the two constituents may cause ambiguity in the semantic of a compound.

Ambiguity also arises due to the structural nature of the Urdu compounds. Like those in Germanic languages, Urdu compounds may be left-branching (modifiers come before the head) e.g.  $k \underline{\sigma} \underline{t} \underline{\sigma} b$  'books' + mela 'fair'  $\rightarrow k \underline{\sigma} \underline{t} \underline{\sigma} b$  mela 'bookfair' whereas like those in Romance languages, they may also be right branching (the modifiers come after the head) e.g. nur 'light' +  $x \circ da$  'God'  $\rightarrow nur$ -e- $x \circ da$  'the light of God'. Of the two compounds  $k \circ do$  mela and nur-e-xoda, an infix plays key role in the formation of the latter, and former is formed by a distinctive compounding word. These are two major elements in Urdu compounds. There are three infixes, -e-, -o-, and -b-, which play various grammatical and morphological roles in the compounds, due to which Durani (2007) terms them determinative compounds. The role of -b- in some determinative compounds needs separate discussion, due to the semantic ambiguity that the compounds may be exocentric or copulative. Most compounds are formed by distinctive compounding words, each of which may derive various compounds. For example x = 0 'happy' derives a large number of formations e.g. x = 0 'smell'  $\rightarrow x = 0$  formations e.g. x = 0 'smell'  $\rightarrow x = 0$  formations e.g. x = 0 'smell'  $\rightarrow x = 0$  formations e.g. x = 0 'smell'  $\rightarrow x = 0$  formations e.g. x = 0 'smell'  $\rightarrow x = 0$  formations e.g. x = 0 'smell'  $\rightarrow x = 0$  formations e.g. x = 0 'smell'  $\rightarrow x = 0$  formations e.g. x = 0 formations e.g. x ='fragrance', x = f + n = sib 'fate'  $\rightarrow x = f = sib$  'lucky'. Sometimes, some constituents also conjoin randomly to form a compound, e.g. sot∫a 'thought' + səmdʒha 'understood' → sot∫asəmdʒha 'well planned'. Neither of the two verbs derives any other compound. Such casual compounds are not numerous, and so not the focus in this work.

By classification type and number of word formations, most compounds are endocentric and verbal compounds, but they also include exocentric and copulative compounds as well as postpositional nexuses. The formations include AN (Adjective + Noun), NA (Noun + Adjective), NN (Noun + Noun), AV (Adjective + Verb), NV (Noun + Verb) and the resulting compounds are mostly adjectives, nouns and verbs, but there are also adverbs, pronouns and postpositions. This discussion begins with some issues in the endocentric class followed by exocentric and copulative compounds.

#### 4.2.1. Endocentric Compounds

An endocentric compound is a type of compound in which one member functions as the head, the other as its modifier, and the compound is the hyponym of the head constituent. For example, the Urdu compound kotob mela 'bookfair' is an endocentric compound in which kotob 'books' is the modifier and mela 'fair' is the head, and the compound kotob mela is the hyponym of the head constituent. Endocentric compounds seem to be the most frequent kind of compounds found in Urdu. They occur primarily in NN and AN, NA and AA combinations. Consider the various configurations.

#### 1a. Noun + Noun

- i) nur 'light' + xvda 'God'  $\rightarrow$  nur-e-xvda (N) 'the light of God'
- ii) hosən 'beauty'+ dzəmal 'beauty'  $\rightarrow$  hosən-o-dzəmal '(kinds of) beauties'
- iii) kʊtʊb 'books' + mela 'fair' → kʊtʊb mela 'bookfair'

#### b. Adjective + Noun

- i) ədzib 'strange' + yərib 'poor'  $\rightarrow$  <u>ədzib</u>-o-yərib (A) 'very strange'
- ii) pak 'pure' +  $\underline{d}$ Il 'heart'  $\rightarrow$  pak  $\underline{d}$ Il (N) 'sincere/true-hearted'

### c. Noun + Adjective

- i) ədalət 'court' +  $\sigma$ zma 'supreme/great'  $\to$  <u>ədalət</u>-e- $\sigma$ zma 'supreme court' d. Adjective + Adjective
  - i) xv f 'happy' + xv rram 'very happy'  $\rightarrow xv f \mathbf{0} xv rram$  (A) 'very happy'
  - ii) həlka 'light' phʊlka 'light' → <u>həlka</u> phʊlka (A) 'very light'

#### e. Pronoun + Pronoun

i) 
$$\underline{t}vm$$
 'you' +  $xv\underline{d} \rightarrow \underline{t}vm\underline{x}v\underline{d}$  (P<sub>RO</sub>) 'yourself'

As the examples above show, there are five structural configurations, and each has some variations. For example, NN compounds contain three variations i.e. with and without the infixes -e- and -o-. The AN and AA compounds also have two variations i.e. with and without the infix -o-. Similarly, there are combinations of NA and (Pro) (Pro). Each of these fulfils various functions conveying various semantics. The focus here is mainly on NN and AA combinations with relevance to the functions of the two infixes -e- and -o-, which show interesting semantic features.

# 4.2.1.1. Infixes or Interfixes in the Compounds

The formation of new words in Urdu by compounding is seen in almost all categories of words, but the compounds taking an infix are mostly found in the endocentric class. Other classes of compounds are rarely seen with any infix. Bauer (2003) terms an infix in a compound *interfix* and describes it as follows, "A special kind of infix that appears between the two elements of a compound is an interfix. This is found in many of the Germanic languages e.g.  $tag + reise \rightarrow tag-e-reise$  'day's journey'." The use of such an interfix in Urdu is a Persian borrowing. Naim (1999) states that it is the most commonly used Persian grammatical feature in Urdu. There are two main interfixes i.e. -e- and -o- whose function is to structurally link the two constituents of a compound. Moreover, they describe the

grammatical relations between them. Bogel et al (2008) and David et al (2009) term -e-ezafah 'increase/addition' and state that theoretically it can only join Perso-Arabic loanwords, but in spoken usage it is occasionally used with words of Indic origin as well. It represents the genitive marker ka e.g. mar 'light' + xoda 'God'  $\rightarrow mar$ -e-xoda 'the light of God' and the agentive particle vala, which also expresses possession. Neither the genitive ka nor the agentive vala themselves are used in NN compounds. However, Mohanan (1994: 115) mentions some uses of vala in some N + V+ vala combinations. A third function of -e- is also to link NA constituents phonologically, in which it plays no grammatical role.

Unlike -e-, the interfix -o- is in fact a phonetic substitution of -v- that is often used to connect two nouns or two adjectives, e.g. *Ahmed-v-Hamidi*, and conveys the conjunctive sense of  $\sigma r$  'and'/'an addition to something'. Historically, -v- is a reduced and shorter form of  $\sigma r$ , like many other Sanskrit origin words, and the development is described in Paniniyan grammar<sup>7</sup>. Therefore, in compounds also, it plays the same conjunctive role and replaces the conjunction  $\sigma r$  'and', which is never used as an interfix. Spelling draws on the knowledge of sound-letter correspondences, syntactic rules, orthographic rules and on the knowledge of morphology (Carlisle 1987). Orthography has also played a part in the shifting from -v- to -o-, and thus phonology is influenced by spelling as well. The interfix -v- is written with a single letter, '9' called *vao*, which is also the representation of the vowel sound -o- in various words. In ordinary conversation, speakers generally pronounce *vao* in the compounds with an -o- sound rather than a -v- sound. Orthography in particular has influenced the way children read such compounds. As the letter *vao* is the same for both -v- and -o-, they pronounce the words with an -o- sound instead of a -v- sound. This is possibly because the -o- sound in the compounds

-

<sup>&</sup>lt;sup>7</sup>. *Panini's Grammar and Paniniyas' Tradition*- A lecture on May 04, 2011 at Uniwersytet Im. Adam. Mickiewicza, Poznan, Poland: By Professor Boris Zakharin from Moscow State University. The author of this work was present in the lecture.

is more rhythemic and more convenient to use than -v- sound. Therefore, -v- is restricted to spelling and -o- is the typical pronunciation. But, orthography-phonology correspondence is not the point of discussion here.

The presence of the two interfixes -e- and -o- in the compounds expresses some grammatical function. Durani (2007) terms such compounds determinative compounds: "there are descriptive compounds in which the modifier is used in attributive, appositional or adverbial manner, and there are also determinative compounds in which one is not an attribute to the other; it is rather related to the other in a way corresponding to one of its grammatical cases". Consider his division of compounds:

### 2a. Descriptive compounds

b. Determinative compounds

$$\operatorname{nur}(N)$$
 'light' +  $\operatorname{xvda}(N)$  'God'  $\to \operatorname{\underline{nur}}$ -e- $\operatorname{xvda}(N)$  'the light of God'

Example (2a) simply describes the configuration in which the modifier is used in an attributive, appositional or adverbial manner. Such compounds are termed descriptive compounds by him. They also include reflexive pronouns. Example (2b) <u>nur-e-xoda</u> 'the light of God' shows an interfix -e-. They are determinative compounds, as the infix -e- plays a grammatical role between the two constituents. It represents the genitive marker <u>ka/ki</u> or <u>ke</u> and expresses its semantics. It thus shows a possessive relation between the nominal elements. It is only semantically related and not a phonological reduction of the genitive marker <u>ka/ki/ke</u>. As mentioned above, the genitive has no use in compounds. Therefore, its semantics must be expressed by -e-. With the genitive marker, the grammatical relation is expressed as <u>xvda ka nur</u> 'the light of God', which is a phrase and not a compound.

Durani's (2007) explanation of the role of an interfix is limited to -e- and it is with just one semantic feature i.e. the representation of the genitive ka. The role of -e- is wider than what he claims, and there are other structures with a different function. Moreover, as noted above, there is another interfix, -o-, that represents the conjunctive particle pr 'and'. Giving the sense of plurality and emphasis, it also forms the structures which do not behave in the way Durani claims. There are five functions the two interfixes perform.

- 3. The interfixes (from 3b. to 3e.) representing other than genitive
- a. rah (N) 'path' + həq (N) 'truth'  $\rightarrow \underline{rah}$ -e-həq (N) 'the right path' (-e- represents the genitive marker ka/ki/ke)
- b. sahıb (N) 'someone/Mr.' + ılm (N) 'knowledge'→ <u>sahıb</u>-e-ılm (N) 'scholar' (-e- represents the agentive particle *vala* showing 'someone having something')
- c.  $\partial$ dialət (N) 'court' + alıja (A) 'high'  $\rightarrow$   $\partial$ dalət-e-alıja (N) 'high court' (-e- describes no semantics, but links the two constituents phonologically)
- d. əsər (N) 'influence' + rəsux (N) 'influence'  $\rightarrow$  <u>əsər</u>-o-rəsux (N) 'resources' (-o- represents the conjunctive particle  $\sigma$ r 'and')
- e. xvf (A) 'happy' + xorrom (A) 'happy'  $\rightarrow \underline{xvf}$ -o-xorrom (A) 'very happy' (The interfix -o-conveys emphasis)

Example (3) shows the structural varieties of the Urdu compounds with interfixes showing various semantic interpretations. Example (3a) is the same as example (2b). The two constituents are nouns, where  $N_1$  is a head and  $N_2$  is a modifier, and the interfix represents the genitive ka. In (3b), the two constituents are again two nouns but the interfix -e- does not represent the grammatical case. Rather, it is a morphological element representing the agentive particle vala showing possession 'someone having something'. In (3c), the two

constituents are noun and adjective, and -e- expresses no semantics but shows only phonological relations between the two constituents. Examples (3d) and (3e) show the interfix -o- in the structures of NN and AA. It represents the conjunctive particle  $\sigma$  'and'. When it appears with two nouns, as in (3d), it shows plurality but when it appears with two adjectives, as in (3e), it denotes emphasis. The examples from (3a) to (3e) are thus evidence that the two affixes -e- and -o- in fact form four structures and perform five functions. This is in sharp contrast to Durani's (2007) semantic interpretation, which is limited to the genitive case shown by -e-. This contrast thus leads to a discussion of what functions the affixes play and what semantics they convey by showing grammatical, morphological and phonological relations between the constituents.

## 4.2.1.2. Functions Performed by the Two Interfixes: -e- and -o-

As noted, (3a) describes possession, and -e- represents the grammatical case i.e. the genitive ka. Constituent  $N_1$  rah 'path' functions as a possession and  $N_2$  həq 'truth' as a possessor. Although (3b) also describes possession, the function of -e-contrasts with the one stated in (3a) and the expression of semantics is also different. Accordingly, the roles of  $N_1$  and  $N_2$  also change, as shown in (4).

```
4a. rah-e-həq (N) 'the right path' = həq ki rah 'the right path'
(-e- represents the grammatical case i.e. the genitive ka)
b. <u>sahib-e-ilm</u> (N) 'scholar' = *ilm ka sahib / ilm vala 'scholar'
(-e- doesn't represent genitive ka but the morphological particle vala)
```

Although the interfix -e- and the two nominal constituents in (4b) seem to be structurally the same as in (4a), they are functionally and semantically different. -e- does not represent the grammatical case, here, but refers to the morphological particle *vala* 'someone having

something.' Like the genitive ka, the agentive vala also shows possessive relation between the nominal elements, but the role of the two constituents is quite the reverse here.  $N_1$  sahib 'someone' now functions as the possessor and  $N_2$  ilm 'knowledge' as the possession. Thus the compound  $\underline{sahib}$ -e-ilm means ilm vala 'scholar/the man of knowledge'. Treating -e- as if it were representing the genitive ka/ki with no change of role by  $N_1$  and  $N_2$  would cause the ungrammaticality of the compound.

Among the individual particles, *vala/vali* can be understood in isolation and functions like a suffix to a noun, an adjective and an infinitive, as the examples below show.

5a. dud vala (N) 'milkman'

b. bori vali (A) 'the bad one'

c. rone vala (Infinitive) 'about to weep'.

Like the genitive marker ka/ki/ke, the agentive particle vala also shows gender/number marking with possessive meaning. Mohanan (1994: 116) describes that vala may be used in four forms i.e. as agentive marker, a relative clause marker, an aspect marker indicating immediate future and in some lexical compounds N + V + vala. However, our concern is not vala but the inetrfix -e- that is used to show its semantic in the compounds. So, it is enough to undertand that vala is lexical and phrasal, and it may mean a possessor, seller, agent or distributor (5a), or it may have the sense of 'one with'. Hardie (2004: 8) terms it occupational particle. It is used to convey the value or price (5b). It expresses a marker of the immediate future also (5c). Like the genitive ka, it is independent and has its own status. Similarly, vala does not have any role in compounds. The only relation between the interfix -e- and the particle vala is that -e- represents vala in the compounds exactly the same as it represents the

genitive *ka*. This representation is however seen in a few nominal compounds, while the representation of the genitive is broad enough.

By this representation, the interfix -e- thus performs a dual role which is morphological as well as grammatical. The roles of the two nouns are also changed. Now the question arises as to how to recognise the two roles of the possessor and the possession performed by  $N_1N_2$  with the same interfix -e- representing the genitive ka in some compounds and the particle vala in some others. It is only possible through an understanding of the wider semantics of the two nominal constituents i.e. supposing what relations they can have in the compounds or which one of them can be the possessor and which one the possession. -e- does not express any other semantic role in addition to those of the genitive marker or the agentive particle.

However, it does appear in some compounds, and its absence causes ungrammaticality. Its appearance sometimes involves phonological relations between the two constituents. The main cause of phonological relations lies behind the fact that there are some NA compounds, in which the second constituent is an adjective. Seidenberg and Gonnerman (2000: 356) consider that morphemes are discrete units, and morphology is correlated with other types of lexical information, including spelling, sound, and meaning. Effects attributed to morphology may be due to these correlated factors. Consider the examples below:

6a. \*ədalət alıja/ \*vəzir azəm

b. adalat(N) 'court' + alija(A) 'high'  $\rightarrow adalat(N)$  'high court'

c. vəzir (N) 'minister' + azəm (A) 'prime/great'→ <u>vəzir</u>-e-azəm (N) 'prime minister'

As (6a) shows, the removal of -e- from such structures causes the ungrammaticality of a compound. Examples (6b) and (6c) show that the function of -e- is wider than merely a

representation of grammatical and morphological elements. In the NA combination, -e- does not perform any possessive role because the representation of the genitive ka and the agentive vala is impossible. This means that the interfix -e- must play some other role. It is in fact a phonological correlation between the two constituents. The aim of these few lines is not to focus on all phonological aspects, but to point out a further role that -e- may play. Therefore, the discussion on this aspect is left to future researchers, as the focus here is mainly on the morphological functions of the interfixes with relevance to their semantic expressions.

The three functions of -e- present a case against Durani's (2007) claim that -e- creates a relation between the two constituents in a way corresponding to the grammatical case i.e. the genitive. In addition to the three functions of -e-, there are two roles, i.e. plurality and emphasis, played by the interfix -o-, this time representing the conjunction or 'and'. Although there are certain other structures, for the sake of space economy, what is discussed here are those with NN and AA constituents, both often being synonyms. Consider example (7):

7a. əsər (N)'influence' + rəsux (N) 'influence' → <u>əsər</u>-o-rəsux (N) 'much influence'
b. hʊsən (N) 'beauty' dʒəmal (N) 'beauty' → <u>hʊsən</u>-o-dʒəmal (N) 'all beauty'

Both the constituents are synonyms in the NN compounds, which convey plurality and emphasis. Such a formation is a collective noun, and it conveys emphasis on some collective actions. It forms agreement with a singular verb. It can be observed in the following sentence.

b. vs \*ka/ke vəsail zijada hẽ his/her.gen resources.pl.m many are '(S)he has many resources'

c.  $\[ vs \] ka \] ke$   $\[ vs \] ka$   $\[ vs \]$ 

1) 'His/Her tyranny is famous.' 2) 'His/Her tyrannies are famous.'

Sentence (8a) shows that the compound  $\underline{\partial s\partial r}$ - $\sigma$ - $r\partial sux$  is a collective noun and expresses plurality and emphasis. Therefore, it agrees with a singular verb. The genitive marker ka, agreeing with the compound, also gives evidence that it is a singular noun. One possible synonym of the compound is  $v\partial sal$  'resources' (8b) that is a plural noun agreeing plural verb.

Many other NN compounds, as in (8c), have synonymous constituents and express plurality or a collective sense of plurality. The compound in (8c) is in fact a hybrid formation of the Arabic noun zolm 'tyranny' and its Persian counterpart  $s_{ID}m$  'tyranny/tyrannies'. The plural of the Arabic zolm is m > zolm 'tyrannies', but the plural of the Persian  $s_{ID}m$  is the same. Note that apparently there is no head-modifier relation between the two constituents. However, we may see a superordinate or a subordinate in the two constituents, as in (8a), but it is not necessarily found in all NN compounds. The two constituents in (8c) do not show this, as they are from two different sources and synonyms of each other.  $N_2$   $s_{ID}m$  is equally compatible with  $N_1$  zolm in daily usage. The compound is used in both senses i.e. it shows plurality; and it also shows a collective sense of plurality. Therefore, it may agree with a singular or plural verb. Even if it is in a singular sense, it tries to convey a series of events with an emphasis.

Emphasis is particularly conveyed by the synonymous constituents of AA compounds. The role of -o- structurally remains the same i.e. connecting the two AA constituents, but it shows changes semantically. It now no longer represents the conjunctive particle or 'and', but rather conveys an emphasis in the expression, as shown by the following examples:

9a. xʊʃ (A) 'happy' + xʊrrəm (A) 'happy' → xʊʃ-o-xʊrrəm (A) 'very happy'
b. həsin (A) 'beautiful' +dʒəmil (A) 'beautiful' →həsin-o- dʒəmil (A) 'very beautiful'

Like those in (8a & 8c), the two constituents in (9a) and (9b) are synonyms. There is no head modifier relation between the constituents, and so the resulting compound may be termed a hyponym of both the constituents. However, there is a superordinate, traditionally the first constituent i.e.  $x \circ f$  'happy' (9a) and  $h \circ sin$  'beautiful' (9b). The subordinates i.e.  $x \circ rr \circ m$  'happy' and  $d \circ sin$  'beautiful' also mean the same as the superordinates, but their meaning is only an additive to the semantics in the compound.

Thus, it is clear that among the two constituents, whether NN or AA in all compounds with the interfix -o-, it is generally the second constituent which plays as a semantic additive. However, there is a positional binding. The positions cannot be reordered whether or not they have superordinate/subordinate relations. This is mainly because of the wide use of  $N_1/A_1$ , and so it is superordinate. On the other hand,  $N_2/A_2$  is secondary and in little use.

In sum, there are four major formations of endocentric compounds, i.e. NN, AA, AN and NA, showing various structures mostly taking the interfixes -e- and -o-. Both the interfixes express some grammatical relations and help to convey possession, plurality and emphasis. The discussion was restricted to the functions of the interfixes. Durani (2007) limits it to just the representation of the genitive ka by -e-, and ignores all other functions by both the interfixes -e- and -o-. The former performs the additional function of representing the agentive vala. Although both representations of the genitive ka and the agentive particle vala by -e- give the semantic expression of possession, they are different and cannot be substituted by each other.

Playing a structural role as a synthesizer of the two constituents means that -e- is purely morphological in its function. It is neither a phonetic reduction nor a phonological substitution of any morpheme, unlike -o-, which is a phonological substitution of -v- that is itself a

phonetic reduction of the conjunctive particle  $\sigma r$  'and'. The discussion on  $-\sigma$ - focuses on only two formations i.e. NN and AA (while both constituents being synonyms). The constituents do not play head-modifier roles in the traditional sense, but usually N1/A1 is superordinate according to the frequency of use. The compounds express plurality and emphasis.

Discussion of endocentric compounds was restricted to mainly the combinations of NN and AA in which the two constituents are often synonyms. The compounds are simple in the sense that they are unambiguous and thus easy to understand. Some exocentric compounds also show synonymy, but they are not as simple as endocentric compounds.

#### 4.3. Exocentric Compounds

As Lieber and Scalise (2006) agree that, exocentric compounding has increased scholars' interest especially in relation to formations like En[glish] *blue eyed*; such constructions, though exocentric, represent a challenge to morphological Lexicalism. *Blue-eyed* soul (also known as white soul or pop soul) is a media term used to describe rhythm and blues and soul music, influenced by pop music. This specific term was first used in 1960s for the performance made by white artists. Thus, although the compound seems to point to someone having blue eyes, it may also refer to a specific piece of music. The Urdu exocentric compounds also show similar semantic ambiguity. Like those in endocentric compounds, the constituents in them are often (near) synonyms, but the compounds are sometimes ambiguous. For example, the two constituents *bal* 'child' and *batfife* 'children' in *bal batfife* 'family' are near synonyms, but the compound is not a hyponym. Exocentricity arises when it is difficult to take a decision on semantic headedness. However, as Katamba (1993: 321) points out, such compounds are syntactically headed in that inflections are added to the right-hand edge of a word. A left hand constituent functioning as modifier cannot be pluralised in an NN

compound, as in a) \*birds-brain 'idiot' b) birds-brains. Therefore, compound (a) is ungrammatical and compound (b) is grammatical. Bauer (2005) notes that this is in contrast to the situation in Spanish as reported in Olsen (2000: 912), where the plural of *actor-bailarin* 'actor-dancer' is *actores-bailarines*, thus apparently confirming the dual-headedness of such constructions in Spanish.

In Urdu, however, these compounds are not dual headed, although they may be right headed or left headed. This section discusses basic characterization and the forms of exocentric compounds. Some semantic issues with relevance to structures are also the focus.

#### 4.3.1. The Structures

All Urdu exocentric compounds are descriptive in nature and, unlike endocentric compounds, contain no interfix that expresses any grammatical function. They are mostly restricted to AN, but there are also examples of NN, AA, NV and VN combinations. The constituents may be synonyms or near synonyms. Durani (2007: 12) considers that (Urdu) exocentric compounds "occur more often in adjectives rather than nouns." The adjectival formations are mostly composed of AN/AA. There are also some nominal formations, but others may be both nouns and adjectives. Like in endocentric compounds, constituents do no udergo structural changes.

10a. AN

- i) zəbər 'top/great' (A) + dəst 'hand' (N)  $\rightarrow$  zəbər dəst 'strong/vigorous' (A)
- ii) tən 'narrow' (A) + nəzər 'sight' (N)  $\rightarrow$  tən nəzər 'narrow minded' (A)
- iii) tʃərb 'sharp' (A) + zəban 'tongue'(N) → tʃərb zəban 'talkative' (A)
- iv)  $m_1t^h$  'sweet' (A) +  $t_1t^h$  or i'knife' (N)  $\rightarrow$   $m_1t^h$  it  $t_1t^h$  or i'charming words to flatter' (N)
- v) xvf 'happy' (A) + libas 'clothing' (N)  $\rightarrow$  xvf libas 'one who wears good clothes' (N)

b. NN

- i) bal 'child' (N) + bətftfe 'children'  $\rightarrow$  bal bətf-tfe 'family' (N)
- ii)  $\int ah(N)$  'king' + xərt $\int (N)$  'expense/cost'  $\rightarrow \int ah x$ ərt $\int$  'spendthrift' (N)

c. AA

- i) mota 'fat' (A) +  $\underline{t}$  əgra 'strong/fat' (A)  $\rightarrow$  mota  $\underline{t}$  əgra 'healthy' (A)
- ii) dvbla 'thin' (A) + pətla 'thin' (A) → dvbla pətla 'weak' (A)

d. NV

- i) tʃiṛi 'sparrow' (N) + mar 'kill/beat' (V) → tʃiṛi mar 'fowler/ boaster' (N)
- e. VN
- ii) hās 'laugh' (V) + mʊkh 'face' (N) → hās mʊkh 'cheerful/jolly' (A)

All the examples show various exocentric compounds, which are nevertheless not in a large number. Most formations are AN, but some examples of NN, AA, NV and VN are also seen. If the constituents are AN, the derivation may be a noun or an adjective. If both the constituents are nouns or adjectives, the derivations may remain unchanged in its category. Although there are not many examples, the combinations of NV/VN form either noun or adjective, or the compound may also be used as both an adjective and a noun. Bauer (2005) describes the Romance type of exocentric compound in English, made up of a verb and a noun which functions as the direct object of that verb; however, it is difficult to say in some cases that the first element is typically a verb or a noun. In contrast, the verb in Urdu exocentric compounds is in base form whatever the constituent position it takes. Moreover, disregarding the category of words, the right hand constituent is generally the morphological head. There is no example of left headedness in AN (the highest number among all combinations). NN, NV or VN formations are also right headed. Only AA combinations, e.g. mota togra 'healthy', (10c-i) are left headed and the modifier e.g. togra is nothing but a

semantic additive. The same relationship can be observed in other compounds e.g. *dvbla patla* 'weak' (10c-ii). These compounds are nevertheless rare.

## 4.3.2. Semantic Expressions of the Structures

The headedness is however morphological, not semantic. As Bauer (2005: 7) points out, some linguists (e.g. Plag 2003: 146) consider that, while these compounds are semantically exocentric in that a *redcap* is not a kind of *cap* nor a *birdbrain* a type of *brain*, they are nevertheless morphologically headed: in *redcap*, *red* modifies *cap*, and in *birdbrain*, *bird* modifies *brain*. This also applies to Urdu exocentric compounds. The constituents show head modifier relation morphologically, but the compounds are headless semantically, as shown:

The adjective constituent is the modifier and the noun is the head in both the compounds. But semantically both are headless. The adjective  $mt_l^{h_i}$  'sweet' in (11a) refers to something soft and alluring and the noun  $tt_l^{h_i}$  or 'knife' refers to something sharp. The compound  $mt_l^{h_i}$   $tt_l^{h_i}$  or it thus refers to charming/alluring words used to mislead somebody. Similarly, although the adjective  $tt_l^{h_i}$  in (11b) means something physically narrow, it semantically refers to the approach of a person. The noun  $t_l^{h_i}$  refers to a person who shows such an approach. The compound refers to a narrow minded person or an approach. The exocentric class contains many other compounds specifically referring to the nature of a person or his manner of speaking. The constituents are simple lexemes individually. However, in the compounds, as Bauer (2005) states, they do not denote the object to which they apparently refer. He calls the phenomenon synecdoche in which a lexeme, e.g. those in the compounds (11), is interpreted according to a well-known figure of speech. Figures of speech are productive, but some

become established as idioms or the words involved become polysemous. Thus, as he states, a syntactic (morphological) tree is so-called by a metaphor, and the tree becomes polysemous as a result of the usage. We can thus say that  $\underline{t} \partial \eta \ n \partial z \partial r$  and  $\underline{m} t^h i \ t^h \partial r i$  are established idioms.

Benczes (2004) terms such compounds "non-transparent" or "creative" compounds for metaphorical (and/or metonymical) noun-noun combinations. He states that with the help of cognitive linguistic tools, e.g. metaphor/metonymy, blending, these compounds are just as easily analysable and transparent as endocentric ones. Thus, it is a more imaginative word formation process. Urdu exocentric compounds are from various sources unlike Romance type English exocentric compounds which are only really exocentric compounds, as Bauer (2005) states. They are not restricted to a single source or some specific structure. Although most are from native Urdu, the non native origin of a constituent is also frequent. The origin of a constituent from diverse sources also gives the compounds various metaphorical meanings. For example, take tan nazar 'narrow minded' and tan dast 'poor'. Both compounds contain the native modifier ton with the meaning '(something physically) narrow'. It expresses various semantics depending on the head noun. In the former, with the Arabic head nəzər 'sight,' it refers to the approach of a person. But in the latter, with the Persian head dəst 'hand', it refers to poverty. *ton* is a distinctive compound word and, as the examples show, forms various compounds. Aronoff (1976: 45) points out that the existence of one of these (lexical items) with a particular meaning does not appear to block its existence with another unrelated meaning; this is important because lack of blocking is usually considered to be a sign of a productive process. This means that the distinctive compound words in Urdu are productive in nature, because each of them forms various compounds expressing distinctive semantics, as also in exocentric compounds. Qureshi and Akram (2008: 8) verify that word

compounding is a productive process to form new words in Urdu; one meaning of a constituent does not block its appearance with another, unrelated meaning.

The two constituents may be synonyms or near synonyms, e.g. *bal batftfe* 'family' (11b-i), disregarding any word. Keane (2001) states that Hindi (Urdu), like other Indian languages, is rich in exocentric compounds comprising two synonyms or near synonyms:

The constituents in (12a) have the same meaning literally, but semantically they refer to different objects.  $d\sigma bla$  is related to an animate object, and  $p \sigma tla$  to an inanimate object. However, it semantically functions as an additive to  $d\sigma bla$ . The compound conveys an emphasis in the meaning 'weak' for a man.  $N_2$  in (12b) is not exactly the same in meaning but refers to the plurality of  $N_1$ . This also shows the complex nature of Urdu exocentric compounds. There are variations in structures, but there are also variations in the constituents' semantic expressions. A modifying constituent may be an additive to the meaning of a head constituent (12a), it may show a singular form of the plural head (12b), and it may also change semantics depending on the source of a head.

In sum, there are not a large number of Urdu exocentric compounds, but they form various combinations, e.g. AN, NN, AA, NV and VN. They are metaphorical and creative in semantics, a feature which Bauer (2005) and Benczes (2004) note in English exocentric compounds. However, unlike the English ones which are only from Romance languages, they are not restricted to a single source of their constituents or only to one type of combination. Each constituent keeps its lexical status, and unlike some other compounds the modifier is not

incorporated and fused with the head constituent. Incorporation of a constituent literally means the merger of one constituent into the other, and losing its lexical identity. Unlike the previous two types of compounds discussed, the incorporation of a constituent is only seen in Urdu copulative compounds, which are also in small number but in various formations.

#### 4.4. Copulative Compounds

A copulative compound shows no head modifier relation but a coordination between the two constituents, e.g. *actor-manager* 'actor and manager'. The two constituents are independent, and thus the compound has two semantic heads. Urdu copulative compounds are relatively few in number as compared to endocentric and exocentric compounds. Nonetheless, they show a greater variety and more structural changes. Unlike those of endocentric and exocentric compounds, their constituents are fused, and one is completely incorporated into the other. Incorporation literally means the merger of one constituent into the other, but here it may also convey the merger of both constituents, and a compound seems to be a single word.

This section discusses copulative compounds focusing on the claims made by Koul (2008: 73) that Hindi (Urdu) copulatives are composed of semantically-related nouns; each noun behaves as an independent constituent in the sense that each may be separately inflected for gender and number, though not for a postposition. There are two points to argue: 1) the composition of copulatives is not made by nouns only 2) there is a possibility of only second constituent taking gender and number marking. It is also agued against Durani (2007) that Urdu copulatives "refer to two or more morphemes connected in a sense by a conjunction."

#### 4.4.1. The Structures

Copulative formations are both with and without interfixes. However, the interfix is only -o-. There are thus two forms of structures i.e. with the interfix -o- and without any infix. The role

of -o- in copulatives is structurally the same as noted in Section 2, as it functions as a conjunctive particle. But, there are semantic differences in its use in copulative compounds. Copulatives show structural differences from endocentric compounds. Most combinations are NN, but some of AA and NA formations are also seen, as the examples illustrated below:

13a. NN

i). 
$$ser(N)$$
 'head' +  $tadz(N)$  'crown'  $\rightarrow ser tadz$  'husband' (N)

b. AA

c. NA

i) 
$$\underline{d}$$
il 'heart'  $\int sp\tilde{a}$  'stick'  $\rightarrow \underline{d}$ ilt $\int sp$  'interesting' (A)

d. NV

i) sər 'head' + phir 'turn' 
$$\rightarrow$$
 sər phir $\boldsymbol{a}$  (m) sər phir $\boldsymbol{i}$  (f) 'mad/crazy' (N/A)

There are various formations of NN, AA, NA and NV. The resulting compounds are nouns, pronouns, adjectives, adverbs and pronouns. The formations may be with or without interfixes, which mostly appear in NN and in AA. There may be structural changes in the compounds without the interfixes, but the changes are often caused by -o- and by gender and number. Many changes are not merely by affixation, but there is a complete incorporation of a constituent, e.g. in (13c), as explained in the following subsections.

# 4.4.1.1. Functions of -o- in Copulatives

Although the functions of -o- seem to be simply an expression of collectivity, it shows some comprehensiveness not restricted to the object mentioned. Consider the following:

14 a. zəmin (N) 'earth' + asman (N) 'sky' → zəmin-o-asman (N) 'universe'
b. sejah (N/A) 'black' + səfed (N/A) 'white' → sejah-o-səfed (Pro)'(ownership) everything'
c. xas (A) 'special' + am (A) 'ordinary' → xas-o-am (Pro) 'everyone'

The appearance of -o- is limited in copulatives, as compared to that in endocentric compounds. The interfix -o- in (14a) conveys collectivity and implies all and everything in the universe. The compound is formed from the Arabic zəmin 'earth' and Persian asman 'sky'. This shows that Urdu copulatives may also be hybrid formations. Similar collectivity is also seen in (14b). The compound sejah-o-sfafed also shows emphasis on somebody's ownership of a certain thing e.g. a business with control of everything related to it, and gives the sense of absolute authority, which nobody can share. Although it expresses simple collectivity, the same as in endocentric compounds, the functions of -o- in the copulatives show different semantic expressions from those of the former. The plurality or collectivity expressed in the endocentric compounds is in a specific sense and is limited to certain factors. But the plurality or collectivity expressed in copulatives gives a comprehensive sense. Recall that in endocentric compounds, if NN constituents are synonymous, the compound is a collective noun, e.g. *asar-o-rasux* 'much influence'; it conveys emphasis on some collective actions, and it agrees with a singular verb. However, a copulatives formation may be a noun, but the two constituents are not usually synonyms, and therefore the compound may agree with a singular or plural verb. In both the formations of NN and AA, the constituents may be nouns and adjectives and the resulting compounds are generally nouns and pronouns i.e. N-o-N or A-o-A  $\rightarrow$  [P] / [N]. When the infix -o- appears with two nouns, i.e. N-o-N compounds, it shows plurality/collectivity and sometimes emphasis, but when it appears with two adjectives i.e. Ao-A compounds, it shows mainly emphasis, as in (14c). Thus, the role of the infix -o- here is

distinctive. However, many other compounds do not take any infix and even then their constituenst are incorporated.

## 4.4.1.2. Copulatives without Interfixes

Booij (1992: 14) considers regarding Dutch that, "There are copulative adjectival compounds such as *rood-wit-blauw* 'red-white- blue' and *Duits-Frans* 'German-French', but it is hard to decide whether they should be considered normal compounds with a right head, or as cases of word-internal asyndetic<sup>8</sup> coordination". Urdu copulative compounds may sometimes be termed the cases of word internal asyndetic coordination, e.g. *sər tad3* 'husband' (16a), as they need no interfix. Moreover, the two constituents often show complete incorporation without requiring any interfix. However, there may be structural differences:

16a. 
$$ser(N)$$
 'head' +  $\underline{t}adg(N)$  'crown'  $\rightarrow ser\underline{t}adg$  'husband' (N)

b. 
$$\operatorname{dil}(N)$$
 'heart' +  $\operatorname{fasp}\tilde{a}(V)$  'stick'  $\to \operatorname{diltfasp}$  'interesting' (A)

c. sər (N) 'head' + fəro
$$\int$$
 (V) 'sell'  $\rightarrow$  sər fəro $\int$  'very brave' (A)

d. sər (N) 'head' + phir (V) 'turn' 
$$\rightarrow$$
 sər phir $a^9$  (m) sər phir $i$  (f) 'mad/crazy' (N/V)

The various examples present two cases of incorporation. Unlike those in (16b-16d), the two constituents in (16a) are both nouns, and are completely incorporated into each other, and the compound is never treated as a compound but as a single lexical item. There is also a possibility that the genitive *ka* may be inserted, but the combination then no longer remains a compound. Rather, it is a phrase *sər ka tadʒ* 'husband' without any effect on the meaning. Thus, breaking apart the two constituents restores their independent lexical status, while in the compound both lose their individuality. Examples (16b-16d) show a complete incorporation

<sup>&</sup>lt;sup>8</sup> Having no conjunction.

<sup>&</sup>lt;sup>9</sup> phir → phira/ phiri (past participle form)

in which a word loses its distinctive identity and shows a merger into the other. The second constituent  $f ext{osp}$  'stick' in (16b) is the reduced form of the verbal noun  $f ext{osp} ilde{a}$  that is itself a restructured form of the Persian infinitive  $f ext{osp} ilde{a} ilde{a}$ .  $f ext{osp} ilde{a}$  is no longer used in Urdu, and due to its merger into the first constituent  $f ext{old}$  its distinctive identity cannot be seen. Although the compound  $f ext{old} ilde{a} ilde{a} ilde{a}$  literally means 'something that sticks to heart', it refers to something interesting. The compound is not treated as a compound. It is used as a single word instead. The second constituent in (16d) shows gender marking.

Katamba (1993: 286) notes that there may be a choice between using independent words and incorporating words such as object NPs, verbs or prepositions into verbs to form complex words. He considers that incorporation is affected by rules that are essentially the same as syntactic movement rules which shift around the constituent of sentences. Morphological incorporation puts one word inside another word rather than shift it to another place. Thus, there are important morphological changes. This is the case with Urdu copulative *diltfosp*. One constituent *fospã* ends up inside the word *diltfosp* that is in fact a compound. However, in contrast to what Katamba notes, a verbal constituent is incorporated into a nominal one.

This is also the case with the verbs *fərof* in (16c) and *phir* in (16d), though there is a little difference. The second constituent is a verb, and there is a noun-verb relation. Katamba (1993: 285) states that in such a relation the noun functions as an object of the verb, and once incorporated "the object NP becomes an integral part of the verb and loses its separate identity." Such incorporation, he states, creates complex words. The two compounds (16c and 16d) are also different from Katamba's approach in the sense that the incorporated constituent is the verb. Like *tfəsp* (16b), the second constituents *fərof* (16c) and *phir* (16d) are incorporated into the nominal constituents. This shows a difference from Katamba's

observation. In his case, the formation is in fact a verb, and the object NP is incorporated to be part of the verb. But in Urdu, the formation is a noun or an adjective, although the literal meaning shows that the nominal constituent functions as an object of the verb. These compounds may also show that the second constituent is incorporated.

The incorporation of the second constituent also shows differences from the observations made by Koul (2008: 73). Some important aspects noted in the Urdu copulatives are in sharp contrast to his approach to Hindi (Urdu) copulatives. His description 10 shows two weaknesses i.e. the limitation of copulatives to NN and gender/number marking to both constituents.

#### 4.4.2. Arguments against Koul and Durani

Koul (2008: 73) claims, "Hindi (Urdu) copulative compounds, also known as co-compounds, are composed of semantically-related nouns. Each noun behaves as an independent constituent in the sense that each may be separately inflected for gender and number, though not for a postposition." This examination is divisible into two parts: 1) Hindi (Urdu) copulative compounds are composed of semantically-related nouns 2) each constituent may be separately inflected for gender and number. The first part of the claim is partially true. Urdu copulatives are not necessarily composed of NN only, but also of AA, NA and NV:

The various combinations include NN, AA and NA. Although his claim of semantic relation of nouns seems true in many of the examples, e.g. in (13a.ii), (13bi & 13b.ii) where the constituents seem antonyms, it is not justified in other compounds e.g. *diltfəsp* (17b) in which

-

<sup>&</sup>lt;sup>10</sup>. Koul (2008) has just described Hindi (Urdu) copulative in a few lines. Unfortunately, he gives no data.

no direct semantic relation between the constituents is seen. So, it can be argued that a direct semantic relation is not necessary. However, a relation is seen only after the incorporation.

Secondly, Koul (2008: 73) observes that each (constituent) may be "separately inflected for gender and number." Various examples show that only the second constituent is inflected.

18a. sər phira (m) / sər phiri (f) 'mad/crazy' → sər phire / sər phiriā
b. dılt∫əsp 'interesting' → dılt∫əspi 'interest' (f) → dılt∫əspijā 'interests' (f)

Example (18a) is noun, and (18b) is an adjective, which is converted into a feminine noun after affixation of -i. The compounds show the gender and number marking in the second constituent only. There is no change in the first constituent. sər sər phıra (18a) is originally a masculine noun. It is changed into a femine noun and the plural forms by the alternation of coda vowel i.e. -i, -e, and -ıjā respectively in the second constituent. The original compound in (18b) is derived from the Persian loans 'dıl' and fəspidən 'to stick'. Both entered the Urdu lexicon at different periods of time, and so they do not show gender distinction. However, after derivation of the abstract noun, dıltfəspi, it shows the feminine gender in the second constituent. Then by the alternation of final vowel -i with -ıjā, it shows feminine plural marking in the second constituent. All the examples show gender and number marking in the second constituent only. If Koul's (2008: 73) claim were to be accepted, the first constituent in the two compounds should be inflected for gender/number: (a) \*səraphıra (b) \*dıltfəspi.

But the compounds are ungrammatical. This proves his claim to be invalid.

Durani's (2007: 14) observation is even more restricted. Interfixes in Urdu copulatives may also function like a conjunction, but certain points do not validate his claim that Urdu copulatives "refer to two or more morphemes that can be connected in a sense by a

conjunction." Although he means to convey that the two constituents are interpreted as if a conjunction were present, this interpretation is not always necessary. As noted above, the interpretation of a conjunction is possible with the interfix -o-. Consider the following:

A copulative may appear in various forms. It may show an interfix -o- functioning like a conjunction or 'and' with its full semantics, as in (19a). But example (19b) denies the first possibility and shows that the presence of an interfix is not necessary. In fact, he limits structural and semantic variety of Urdu copulatives. The two constituents are incorporated and connected without any conjunction or any interfix representing a conjunction.

Olsen (2000) states that it is unclear from traditional discussion, whether copulative compounds form an independent class of compounds or determinative compounds or if they are simply a formal and/or sub types of more basic determinative patterns. Like Urdu endocentric compounds, some copulatives show grammatical relations, and so they are determinative compounds. However, they are distinctive in various other aspects, e.g. some of them contain no interfix showing any grammatical relation. Therefore, they should be treated as an independent class. This work has tried to clarify two major structural aspects of the copulatives with and without the interfix -o- with various formations of NN, NA, AA and NV. Whether or not they take interfixes, they are distinctive in that their constituents are often completely incorporated. The interfix -o- functions like the conjunction or 'and', which has no role in the compounds. However, Urdu copulatives do not necessarily take any conjunctive interfix, which is in contrast to Durani's (2007) claim. Contra Koul (2008: 73), Urdu copulatives are not restricted to semantically related nominal constituents. They may show

gender and number distinctions but only by marking the second constituent, and this is as predicted by the right-hand head rule. The semantics of -o- is wider than that in endocentric.

More unique is the interfix -b-, whose structural role, in some determinative compounds, is similar to those of -e- and -o-, but its semantic link is seen with the second constituent. -b-functions in the manner of postpositions in compounds. These compounds cannot be semantically substituted by any postpositional phrase. This is unlike the compounds formed by -e- and -o-, which have the semantic substitutes of the phrases formed by the genitive ka, the conjunctive or 'and' and particle vala.

# 4.4.3. Structural and Grammatical Role of -b- in Some Determinative Compounds

This section discusses some general determinative compounds focussing on the features of the third infix -b-. In all such compounds, which are few as compared to others, the common element is the Persian loan noun s 
eta r 'head' in the  $N_1$  position. The second constituent in  $N_2$  position may be a native noun or a loan noun, e.g. Arabic sids da '(head down) in prayers', as shown below, and the formation is also a noun.

```
15a. sər 'head' + sıdʒda '(head down) in prayers' → sərbsıdʒda 'showing respect'
b. sər 'head' + rah 'path/way' → sərbrah 'leader'
c. sər 'head' + dəst 'hand' → sərbdəst 'bravely (lit: one's head on his hand)'
```

The infix -b- appears after one noun, and so it seems allomorphic. However, it does not have any other variant. It<sup>11</sup> seems affixed more with the second constituent morphologically and functions in the manner of postpositions to indicate *the position above/supported by/in contact with* (something). Example (15a) literally means 'one's head down in prayers',

-

<sup>&</sup>lt;sup>11</sup>. Explanation given by the dictionary from CRULP (*Centre for Research in Urdu Language Processing*, Islamabad)

though, it symbolically refers to showing respect to God, particularly, or even to a human. Similarly, (15b) literally means someone 'bringing heads (people) on the path' i.e. the leader, while (15c) literally means one 'having his head in his hands', i.e. it refers to someone doing something while putting his life in danger.

Thus, by functioning in a postpositional manner, -b- actually performs a grammatical role. Therefore, removing it from the compounds renders them ungrammatical e.g. \*sər-rah and \*sər-sıdʒda. However, it does not mean that it is a replacement of postpositions, e.g.  $m\tilde{e}$  'in' in (15a). Like the genitive ka/ki and conjunctive particle  $\sigma$  'and', postpositions do not function as affixes in the compounds; they do not have any morphological roles and only function syntactically. Therefore, their semantic meaning in the compounds is only conveyed by -b-. Moreover, unlike those of -e- and -o-, the grammatical use of -b- in compounds cannot be substituted for by postpositions in phrasal forms. Some phrases can be formed by the genitive ka/ki and the conjunctive  $\sigma$ , but there can be no phrases formed by postpositions e.g.  $p\sigma$  'on' in \*rah  $p\sigma$  sor, which gives the semantic sense of  $s\sigma$  that 'leader'. This is unlike the functions of -e- and -o-, whose substitutes, i.e. the genitive ka/ki and the conjunctive particle  $\sigma$  are commonly used in phrasal forms. Thus, the grammatical role performed by -b- is unique in this sense.

In addition to the grammatical role, -b- performs one more peculiar function not performed by -e- and -o-. The two constituents separated by -e- and -o- keep their distinctive identity, and so their combination is presumed to be a compound. In contrast, -b- helps to fuse the two constituents completely together into a word, so much so, that the compound seems to be a single entity. The compound is not taken as a compound; rather, it is treated as a single lexical item functioning as a noun, adjective or an adverb, as in (15a-15c).

There are several reasons to treat it as a single word. The first of them is morphological. The constituents are completely incorporated, and the second constituent particularly loses its lexical status due to -b-. It is this function that makes -b- distinctive and unique. As said earlier, removing -b- is not only ungrammatical but also results in a nonsense phrase e.g. \*sər-dəst. This is also unlike the compounds with -e- and -o-, e.g. sejah-o-sfəfed, which retain some meaning after the removal of the infixes.

Secondly, unlike those with -e- and -o-, the compounds with -b- are orthographically written as single word. -b- seems affixed more with the second constituent. In this status, if separated apart, e.g. \*brah form sər, it makes no sense. Only the first constituent seems to constitute a single word written.

The third reason is to do with the semantics. -b- is semantically linked in the postpositional manner with the second constituent. A formation, e.g. sərbrah (15b), literally means 'one who brings heads on the path', but it semantically refers to a leader. Generally speakers of Urdu are not aware of literal meaning of the compound and know only the surface meaning, considering the compounds as single words.

-b- is thus distinctive from the two other infixes in its grammatical role, incorporation and semantics. The Ffirst constituent s 
other r is a Persian loan and the second constituent may be native or Arabic, but the source of any constituent does not affect the function of -b-. Its use is limited to some compounds, but it plays a vital role in the incorporation. Unlike those of -e- and -o-, the compounds by -b- do not have substitutes in the form of postpositional phrases.

Urdu postpositions themselves form certain compounds and show some distinctive features, but such compounds are not numerous.

#### 4.5. Postpositional Compounds

There are only a few postpositions in Urdu, but they play a key role in some compound constructions. They include the instrumental/ablative *se*, the fused forms of the genitives i.e. *ke* and *ki*, *prof* 'towards' and *xulaf* 'against'. As discussed by Bauer (2005), a phrasal type of exocentric compound in English, e.g. *showoff*, may be a noun that is made up of a verb + a preposition/adverb/particle; there is no noun involved in *showoff*, yet the construction as a whole is a noun. He acknowledges that people are reluctant to accept this analysis, as phrasal verbs are often considered to be syntactic constructions, while nouns are single lexemes. Whether or not the verbs are lexemes, they are like listemes, in Bauer's words, the same as nouns. All morphemes, irregular forms, and idioms are listemes that must be memorized because their sounds or meanings do not conform to some general rule. Urdu postpositional compounds are different from the English phrasal type compounds in that they are neither noun formations nor do they involve verbs. Rather, they are adjectives and adverbs. Moreover, a postposition has a fixed position in such compounds. For example, the instrumental/ablative *se* always appears as the second constituent and the genitive *ke* or *ki* appears as the first constituent. Their compounds convey various semantics, as shown below:

Table 1. Postpositional Compounds

Constituent 1	Í	Constituent 2	Compounds
1. aram 'rest' (N)		se (instru/ ablative)	aram se 'peacefully' (Adv)
2. genitive ka/ki/ke		lije 'concerning'	ke lije 'for'
3. do 'two' (N)		tərəf 'side/towards'(PP)	do tərfa 'bilateral' (A/Adv)
4.məndərd3 'inserted' (A)		zel 'below' (PP)	məndərdza zel 'given below'(A)

Note: pp —postposition (3,4), I - Interfix

The most important postpositions are se and ke/ki, and therefore they are the focus here. They form a large number of compounds with nouns, adjectives and postpositions, and express various semantics. ke and ki mainly form postpositions and se mainly forms adverbs. It is noteworthy that ke and ki are homophonous to the genitives ka/ke/ki and seem the same, but in fact they are different from genitives. Genitive markers ka, ke and ki show gender and number distinction, while ke and ki do not. As Thakur (1997: 14) points out, Vajpeyi (1958: 30) is the first Hindi grammarian to differentiate the genitive case-markers, i.e. ka/ke/ki from ke which occurs with compound postpositions. He notes that l-a/l, l-i/l and l-e/l, in the genitive l-ka/ke/ki, are "agreement markers and l-ke in them is the genitive case marker"; postpositions denote different cases and, on this basis, grammarians have categorized l-ke, l-ke and l-ke is the fused form and functions like the genitive case. Thakur (1997: 39) considers that l-ke is the fused form and functions like the genitive, but it expresses no gender and number distinction. His observation is nevertheless limited to l-ke. Another fused form, l-ke, also functions in the same manner. A comparison with the genitive markers reveals an interesting picture:

20. <b>Gr</b> o	oup A.			Group B.			
a. ləkir	k <b>a</b>	sır <b>a</b>		d. ləkir	ke upp	er	
line.	gen (m)	edge (m)		line.	ke. abo	ove	
'the ed	lge of the lin	ne'		'above th	e line'		
b. ləkir	ki	ləmba-i		e. Joe ki	<u>t</u> ərah	se	
line.	gen (f).	length (f)		Joe ki	like.	instr	
'the len	igth of the 1	ine'		'like Joe'			
c. ləkir	k <b>e</b>	nı∫an <b>a<u>t</u></b>		f. Anne	ke	bare	mẽ
line.	gen (pl)	marks (pl)	Anne	ke	about	loc.(in)	
'the ma	rks of the l	ine'		ʻabout Aı	nne'		

As *ki* functions the same as *ke*, it should also be treated as the fused form of the genitive. However, it forms fewer compounds, e.g. *ki tərəf* 'towards' and *ki dʒanıb* 'towards'. The genitives *ka/ke/ki* in Group A and their fused forms *ke* and *ki* in Group B are different from

each other. The genitives ka/ki/ke in Group A show possession and gender/number distinction due to their agreement with the following nouns i.e. sira 'edge' (m), ləmba-i 'length' (f) and mfanat 'marks' (pl). The combinations are simple phrases and not taken as compounds. The fused forms ke and ki in Group B do not show possession and gender/number distinction. They are followed by an adjective, postposition and noun (in rare cases) with which they form compounds. None of the three compounds are grammatical without ke/ki, e.g. \*lakir upper. Not showing gender, number and agreement morphology means that they must be playing some other role. This is to form some kind of relation between the preceding noun and the following element, usually postposition or adjective. Without them, the position of a noun cannot be specified e.g. \*Joe tarah se (20e).

There are two major differences between the genitives ka/ke/ki and their fused forms ke/ki. Genitives show gender and number morphology in agreement with the following noun. They are not followed by any other category word. The fused forms ke and ki do not show possession, gender and number morphology. They are followed by adjectives, postpositions, and nouns (in rare cases) with which they form compounds as shown below:

- 21a.  $ke + bare 'about' + m\tilde{e} (locative) \rightarrow ke bare m\tilde{e} 'about'$ 
  - b. ki + terah 'like' + se (instrumental)  $\rightarrow ki terah$  se 'like'

Both ke and ki combine with adjectives in (21a) and postposition (21b). The combinations with postpositions also involve the ablative/instrumental se and the locative  $m\tilde{e}$ . The major point to explain is that genitives involve the relation of NP with an NP, but the fused forms generally make some relation between the preceding noun and the following element, usually a postposition or adjective. The homonyms involve postpositions and adjectives which are nevertheless in a limited number.

Although they express some semantics, e.g. purpose, function, reference, ke and ki are deficient in productivity as compared to the instrumental se with which they combine and form compounds. The instrumental/ablative se in contrast is a productive element in Urdu, as it derives a lot of new words from native and loanwords, and conveys meaning. As the name shows, it functions as instrumental, e.g. zəban se 'with tongue' and source of action, e.g. Karachi se 'from Karachi'. Ahmed (2008) states that instrument markers in South Asian languages show syncretism, i.e. the merger of two or more originally different inflectional forms with different markers. There is in fact no syncretism pattern; some South Asian languages have the examples of comitative instrument (marking companions). Urdu/Hindi uses the postposition saath 'with' as the comitative marker, e.g. mere saath 'with me'. But, se shows secondary comitative marking.

The ablative/instrumental *se* forms adverbs of manner and comparative/superlative degrees of adjectives. The formation of compound adverbs is simple and is only made by attaching to nouns. *se* sometimes links two lexemes, particularly two adjectives to construct an adverb.

22a. zor 'force' (N 
$$\rightarrow$$
 zor **se** 'forcefully' (Adv)

b. məkkari 'cunningness' (N) → məkkari **se** 'cunningly' (Adv)

c. 
$$\operatorname{atfhi}$$
 'good' (A) + tarah 'like' (A)  $\rightarrow \operatorname{atfhi}$  tarah se 'well' (Adv)

Both (22a & 22b) show simple adverbial constructions, involving nouns and instrumental, but (22c) shows a combination of three elements i.e.  $A + A + Inst \rightarrow Adv$ . Such formations are however rare. The adverb formations are thus in two categories, although the manner of forming adverbs is the same.

However, there are two ways to form comparative/superlative degrees. In the first, like adjectives in English, Urdu adjectives also take comparative and superlative degree markers, -tər, -tərin, which are borrowed from Persian and attach to adjectives in the same manner:

23a. sərd 'cold' 
$$\rightarrow$$
 sərd tərin 'coldest' (Persian)

b. 
$$b^h \text{ ond } a \text{ 'odd'} \rightarrow b^h \text{ ond } a\underline{\textbf{t}} \text{ ar' odder'} \rightarrow b^h \text{ ond } a\underline{\textbf{t}} \text{ arin' oddest'}$$
 (Native)

Although most formations may be Persian borrowings, as in (23a), the Persian degree markers may also form comparitives and superlatives with native (23b) and Arabic loanwords (23c). Example (23c) is particularly interesting. The base form is already a hybrid compound of the Persian *bod* 'bad' and the Arabic *sorrort* 'face'. Moreover, it takes Persian degree markers for comparitives and superlatives. Persian formations may be of the greatest frequency individually, but the formations with native and Arabic words as a whole are certainly more numerous than them. These formations are primarily inflectional except for a few hybrid compounds, e.g. *bodsorort* (23c), which take the degree markers.

Degrees of adjectives are natively formed derivationally by using the instrumental/ablative se in combination with the Arabic loan adjective zijada 'much' for comparatives and an additional native adverb sab 'all' for superlatives. Such compounds are also not entirely native formations, and there is no distinction of the source of base form, as shown below:

24a. bəţa 'big/old' 
$$\rightarrow$$
 Wania se (ziada) bəţa 'older than Wania'

$$\rightarrow$$
 səb **se** (ziada) bəra 'oldest' (Native)

b. paedar 'durable'  $\rightarrow$  lohe se (zijada) paedar 'more durable than iron'

$$\rightarrow$$
 səb se (ziada) paedar (Persian)

c. həsin 'pretty' → Asifa se (ziada) həsin 'prettier than Asifa'

 $\rightarrow$  səb se ziada həsin 'prettiest' (Arabic)

The three base forms are taken from the three sources. se functions differently but conveys the sense of comparative/superlative degrees. The Arabic adjective zijada 'much' is optional and often ignored, when two objects of comparison are mentioned. Thus in such a case, se for comparative and the phrase sab se (sab 'all') 'among all' for superlative are sufficient for the degree formations.

Both the ablative se and the fused forms of the genitives ka/ki play distinctive roles, but their formations are not as varied as the other compounds, e.g. endocentric compounds. ka and ki are different from the gentitive markers ka, ki and ke in that they do not show possession and gender/number marking. Unlike genitive markers, which only form a phrase, they play a key role in compound formation, but the compounds are relatively few in number. The ablative/instrumental marker in contrast is not so limited in compound formation. It forms qualitative adverbs and degrees of adjectives, for which it combines with the various base verbs of the three sources. Thus, it seems to be productive enough to form certain compounds.

There are a variety of Urdu postpositional compounds. They are unlike the phrasal type of exocentric compounds in English e.g. *showoff*, which are actually made up of a verb + a preposition/adverb/particle, and the derivations are nouns. But, there are also verbal compounds, in which one constituent is a noun, adjective or a verbal noun but the other must be a dummy verb.

### 4.6. Verbal Compounds

Urdu verbs come in two types. A large number of verbs are already present in the base form. For example, a lexical verb pi 'drink' requires the suffix -na to form the infinitive pina 'to

drink'. It takes other suffixes according to tense and aspect needs, e.g. *pija* 'drank' and *piraha* 'drinking'. However, many times verbs do not exist in their verbal capacity to describe an action. Therefore in contrast to those inflectional in nature, a large number of verbs are derivationally constructed in compound forms from nouns and adjectives (particularly loans) in combination with some dummy verbs.

A dummy verb in Indo Aryan languages is in fact a main verb but also functions as a tool, in combination with a noun or an adjective, to form a compound verb in the absence of a base verb. In Urdu, it is often a nativised form of a Persian auxiliary. For example, kərna (from kərdən) 'to do' and hona (from budən/fodən) 'to be' show the nativisation of Persian loans. The Persian suffix -ən shows the infinitive marking, but its nativisation is in two different ways. Urdu borrows a large number of Persian verbs but doesn't accept the infinitive marker -ən, which is replaced mostly either by a dummy verb, e.g. foruxtən becomes fəroxt kərna 'to sell', or with the native infinitive marker -na, e.g. a verb xəridən becomes xəridna 'to buy'. Discussing the Arabic influence on Urdu, Versteegh (2001: 497) states that it appears that all these formations follow the pattern of Arabic loans in Persian. This is possibly because the Arabic loans are commonly used with the Persian verbs kərdən 'to do' and budən/fodən 'to be' etc. However, as noted, some dummy verbs are themselves modified forms of Persian loans, which shows that a compound verb may be a hybrid compound in Urdu.

Moreover, a loan verb alone cannot function as a verb. It is dysfunctional; it merely functions as a verbal noun, and it has to combine with a dummy verb, which carries the inflectional information. The morphological functions of a dummy verb are the same, and only a dummy verb shows gender and number or tense and aspect changes. Thus, all loans, whether nouns, adjectives or even verbs, must be integrated with dummy verbs to form compound verbs. For

example, an Arabic loan *1zzət* 'respect' is used in the verb form with a dummy verb as *1zzət kərna* 'to respect' (infinitve) and *1zzət ki* 'respected' (past). Thus, a loanword functioning as a verb must be used in a compound form.

Sometimes, a verbal base exists and is fully functional, but it is also used as a verbal noun and is combined with a dummy verb for the formation of a compound verb. However, there is a change of semantics. For example, day 'run' is a base verb, but it is also used as a noun. The infinitive form of the base verb is day 'to run'. But the noun day 'a run' combines with a dummy verb lagana 'to touch' to form a simple verbal structure in the infinitive form day lagana 'to run'. It is neverthelss semantically different from day na, as it involves a volitional act of running. The dummy verb laga is a transitive verb and shows subject's intention. So, in day lagana, it expresses subject's willingness for the act of running, which is not seen in day na. The use of a verbal noun with a dummy verb in this manner also shows that a loan verb depends on a dummy verb to form compound verbs.

The number of dummy verbs is judged to be between 12 to 15, but the most common verbs used as dummy verbs are *kərna* (from *kərdən*) 'to do', *hona* (from *budən/fʊdən*) 'to be', *lena* 'to take', *dena* 'to give', *dʒana* 'to go' and *ana* 'to come' etc. Schmidt (1999: 101) explains that in general, Urdu verbs demonstrate a very regular conjugation with the exception of five verbs *ho* 'be', *kər* 'do', *de* 'give', *le* 'take', and *dʒa* 'go'. These five verbs primarily function as main verbs, and secondarily as dummy verbs (Versteegh 2001: 488) for the formation of compound verbs and light verbs<sup>12</sup> (Butt, 1995) in complex predicates.

-

<sup>12.</sup> For the purpose of morphological study, dummy verbs have been discussed here. The term light verb is used in syntax, which is not the interest of this work. However, the difference between the two terms has been explained.

There is a controversy over whether dummy verbs should be termed dummy verbs or light verbs. The two terms are different functionally. A light verb forms a complex predicate, which is a verbal structure of two or more verbs in a sentence in which the main verb is mostly in lexical form, and the light verb forms the tense morphology. It can be said that V (kha 'eat') + IV (lija 'taken') = complex predicate (kha lija 'has/had eaten'). Note that V is the main verb, and it is followed by a light verb (IV) in syntactic order. Tense formation depends not on the main verb but on the light verb (Butt, 1995), as seen in the present example. Thus, the function of a light verb is to support a main verb to form a complex verbal construction V<sub>1</sub>V<sub>2</sub> or complex predicates in other words. By contrast, the function of a dummy verb is to combine with a noun, adjective or a verbal noun to derive a compound verb NV/AV combination. Bukhari (2009: 80) claims that, "creating new verbs by a productive process of combining a nominal (or adjective) with a light verb is a common feature of Indo Aryan languages." He states that a verbal structure of N + V or A + V is a light verb, as it combines with a noun or adjective to form complex predicates. There is a discrepancy between a dummy verb and a light verb, which Bukhari overlooks and it is necessary for this to be clarified. If N +V or A + V is a productive process of creating new verbs, a verb in such a construction must be a dummy verb. Neither the construction N+ V or A+ V is a complex predicate nor is V a light verb in this structure. Dummy verbs are used with their lexical meanings. Light verbs do not express their lexical meanings. Rather, they modify the semantics of a sentence. In simple terms, both types of verbs are in fact main verbs and can be used as dummy verbs and light verbs in two different constructions. In VV formation, they function as light verbs but do not express their lexical meaning. In NV or AV formation, they function as dummy verbs, expressing lexical meaning, to derive compound verbs. The difference is illustrated below:

- 25a. Wania ne Azan ko məʃvəra **drja**Wania.f.s.erg Azan.m.s.acc advice.m.s give.**pst**.m.s
  'Wania advised Azan.'
  - b. Wania ne Azan ko məʃvər*a* **de dija (he/tha)**Wania.f.s.erg Azan.m.s.acc advice.m.s give.m.s.**bse** give.m.s.**perf** (be.3.s.pre/pst)
    'Wania has/had advised Azan'

In examples (25a) and (25b), the verb *de* 'give' has been used in two different roles, as a dummy verb and as a light verb. In (25a), functioning as a dummy verb with its lexical meaning in the past form, it constructs a simple verbal structure in combination with the noun *mafvara* 'advice'. The construction thus expresses the semantics of a past verb. With a basic lexical form *de* 'give', the dummy verb retains this role in (25b) and correlates with the noun *mafvara* 'advice' to form again a simple verbal construction that functions as the main verb of the sentence. However, used in perfective form *dija* in a different syntactic position in the same sentence, it takes the role of a light verb and expresses the tense/ aspect morphology. In this role, it now modifies the semantics of the main verb *mafvara de* 'advise'. The dummy verb *dija* (simple past)' in (25a) and the light verb *dija* (perfective) in (25b) are identical. The perfective aspect in (25b) is only possible when the main verb *mafvara de* (used in base form) combines with the light verb *dija* (expressing the tense morphology). Thus, VV as a whole is a complex verbal construction or complex predicate in Butt's (1995) terms.

In short, the function of the light verb is syntactic in nature, as it forms a complex predicate, i.e.  $V_1V_2$  in which the main verb  $(V_1)$  is mostly in its base form and the light verb  $(V_2)$  shows the tense morphology. The function of the dummy verb is to exhibit morphological information. It derives a compound verb in combination with a noun or adjective, i.e. N V/AV, which is needed when a lexical verb does not exist. Versteegh (2001: 497) considers that the extensive use of dummy verbs in compounds is typical of Hindi-Urdu.

Bauer (2003) defines two types of compounds: 1) root compounds or primary compounds discussed in different classes of compounds, e.g. endocentric, exocentric (see above) and 2) synthetic compounds or verbal nexus in which the verb is the head. As noted, Urdu compound verbs are the combination of dummy verbs with nouns, adjectives or verbal nouns. They are different from Bauer's (2003) synthetic compounds, e.g. *dish washer*, in which a verb is the head element and fully incorporated. The head element in an Urdu compound verb is a noun, adjective or a verbal noun. A dummy verb remains a modifier. It does not incorporate and maintains its distinctive status. It shares the meaning in the formation of a compound verb.

# 4.7. Summary of the Chapter

Urdu compounds show various features. Most compounds are formed by some distinctive compounding words each of which derives a large number of compounds without distinguishing the source language of the constituents. Therefore the formation of hybrid compounds is dominant, and so the major types of compounds were the focus in this chapter rather than distinguishing them with reference to their source language. These various types include endocentric, exocentric, copulative, postpositional and verbal compounds. The majority of compounds are endocentric verbal compounds, but there are also formations of copulatives, genitives like *ke* and *ki*, and the ablative/instrumental *se*. The resulting compounds are mostly adjectives, nouns and verbs, but there are also adverbs.

Urdu compounds are often ambiguous due to their structural nature, as they may be left branching (modifiers come before head) e.g. xofnəsib 'lucky' or right branching (modifiers come after head) e.g. ədalət-e-ozma 'supreme court'. They are also ambiguous because the two constituents are often synonyms, and the compounds may be an endocentric formation,

e.g. *rah-e-həq* 'the right path' or an exocentric one, e.g. *balbətfe* 'family'. The four examples show that Urdu compounds are formed with and without interfixes. Those with interfixes, found in endocentric and copulative classes, show interesting structural and semantic properties. The interfixes in endocentric and copulative compounds, *-e-* and *-o-*, function as genitives *ka* and the conjunctive *or* 'and'. They convey possession, plurality and emphasis. But the interfix *-b-*, in a few determinative compounds, express the semantics of various postpositions, e.g. often *por* 'on'. The compounds without interfixes are mainly found in exocentric and verbal compounds. Exocentric compounds are metaphorical and creative in semantics, but they are not in a large number. Verbal compounds are only hybrid formations in the sense that their key constituents are dummy verbs, often modified forms of Persian auxiliaries. However, there is no distinction in the source languages of the nominal and adjectival constituents.

Taking the morphological structures from the three sources of Urdu words, i.e. native, Persian and Arabic as the base of Urdu morphology was necessary throughout the last four chapters, because there are major changes in Urdu morphology with respect to the sources. Therefore, now, it may be helpful to understand the morphological structures of English loans, and the discussion on the more recent loans can be started. It is necessary to see to what extent English loans follow the same patterns as native or Persian and Arabic loanwords. This will be the focus of the next chapter.

# Chapter 5

# Morphological Adaptation of English Loans in Urdu

#### 5.1. Introduction

In the previous chapters, Urdu morphological changes have been discussed with respect to three sources of words i.e. native Urdu, Persian and Arabic. The discussion in this chapter is in the light of Winter's (2008: 156) discussion of the integration of loans. It looks at some forms of the heavy integration of English loans, and how the adaptations take place. The discussion focuses on two points, i) that the English loans are in principle subject to the same processes as Persian and Arabic loans, but ii) that they are less well integrated than these older Persian and Arabic loans. It is necessary to see the productivity of native Urdu, Persian and Arabic affixes attaching to English loans, and the adaptability of English loans with which of the three source language affixes, the loans are more frequent.

Some aspects e.g. gender, number, case morphology, derivation with various categories of the same word borrowed, formation of hybrid compounds with or without any morphological changes in the English constituents, English affixes in Urdu and creation of new lexical items are taken into consideration. Although the creation of new lexical items is not expected, it is nevertheless possible. As the loanwords are simply more recent, they are felt by general speakers to be 'more foreign'. There is a strong correlation between the length of time a loanword is seen in a language and the degree of integration that takes place. Crawford (2008: 64) notes the same correlation in Japanese, "The degree of nativisation of a loanword also correlates to the length of time it has been attested since 19<sup>th</sup> century showing more nativisation than those words that have only been attested recently." This is seen in the discussion ahead. Some sub conclusions are drawn based on these two points and the morphological structures from the three sources.

## 5.1.1. Extent of English Loans and Morphological Changes in them

New words are not created from nowhere, but are either borrowed or formed by combining words or parts of words. Borrowing may vary in degree from casual to heavy lexical borrowing, and from slight to the significant incorporation of structural features. Winford (2003: 17) states that borrowing begins with vocabulary, and the incorporation of structures comes only after the substantial importation of loanwords. Based on his scale of borrowing, the contact situation between English and Urdu can be illustrated as:

Table 1: Language Contact Situation between Urdu and English

	English $\rightarrow$ Urdu	Urdu → English
Lexicon	very strong	minimal
Phonology	weak	strong
Morphosyntax	weak	strong

English here refers to the Urdu speakers of L2 English. In this context, the influence of English words on Urdu is strong. Conversely, the influence of Urdu words on English is minimal, but the influence of Urdu phonology and morphosyntax seems to be strong. This is natural because the speakers of a language use loanwords which are influenced by their first language, and therefore the phonological changes take place. Winford (2003: 22) considers some major outcomes after the importation of loanwords over a certain period of time. There are three stages of contact situations, i.e. (A) language maintenance (B) language shift and (C) language creation. Despite heavy borrowing from English, Urdu can still be seen to be in the language maintenance stage and has not yet reached the language shift, as has occurred under the Persian and Arabic influence. However, the use of the term language shift here means a community of speakers no longer using their traditional language but instead shifting to the

use of another language. Urdu tends to turn to Arabic and Persian, while Hindi tends to turn to Sanskrit for new words. Due to this fact, the two languages have consistently given a shape to Urdu with the passage of time, and their influence has gone into the creation of words, phrases, and terminologies. With regard to English, the influence is nonetheless not to the same extent. However, the lexical influence and the structural fusion of English loans are considerable, and they need to be examined.

King (2000: 173) observes that the influence of English on PEI (Prince Edward Island) French has been essentially lexical and these lexical innovations "have triggered particular language-internal changes, resulting in the emergence of a number of structural differences in PEI French". Sankoff (2001: 654) states that King bases this analysis, as English *back* corresponds in meaning to one meaning of the French affix *re*- 'return to a former state or place', as in *venir back* 'to come back' (*revenir*); the borrowing of English *wh*-words also ensues changes in French relative clauses. Similar to this, the lexical influence of English loans on Urdu is deep, and there are some structural changes as well. Nevertheless, the language internal changes are not so many.

Winter's (2008: 157) discussion on the issue in light of Haugen's (1950) *Importation and Substitution* of loanwords is useful here. A word is said to be imported, e.g. the English loanword *faithful* in Urdu, if it is not changed structurally and conforms to the source language form. However, a loanword is said to be substituted, e.g. (English) *frad* 'fraud'  $\rightarrow$  *fradia* / *fradi* 'one who gets benefits by treacherous means' (Urdu), when it is structurally changed in the target language and replaces the original one. The words *fradia* and *fradi* are in fact new lexical creations in Urdu, and there is no parallel derivation for their meaning in the English lexicon. Although such new creations are not many, they mirror the start of the

similar influence Persian and Arabic have had on Urdu. The morphological changes are not widespread, but the degrees of change are notable. To understand the degrees of change, it is necessary first to see the general types of English loans and the patterns of adaptations.

# 5.1.2. Types of Loans and the Patterns of Adaptation

Although there are a large number of adjectives and verbs as well, loan nouns are dominant and can be classified into three groups. Almost all scientific names and terminologies, e.g. ozone, oxygen, carbon, computer, TV and engine, have no Urdu substitutes and so are a part of the Urdu lexicon. Others have Urdu substitutes but they are not very commonly used. English counterparts are more frequent e.g. libas 'suit', pathun 'pants' and bæthak 'drawing room' etc. Still others have common Urdu substitutes, and people use them as well. However, using English loans instead is gradually becoming a fashion, e.g. intehan 'exam', dist (m)/sahuli (f) 'friend' and kayaz 'paper'. Native equivalents to English loans are frequently supplemented or replaced. In some instances, the long spelling and difficult pronunciation of Urdu words, e.g. bavərtfixana 'kitchen' and bætolxəla 'toilet' have facilitated the choice of English loans as a compromise. Heavy borrowing has had so much influence that many Urdu words are getting close to becoming obsolete, as the latter of the two examples. There are a huge number of loans used in their original form with no morphological changes. Some lexical items and compounds are shown in the table below:

Table 2: Patterns of English Loans with no Changes

Lexical Items	Compounds
hotel, nursery, agenda, vote, certificate,	all rounder, governing body, income tax,
consultant, dictator, parking, propaganda,	local government, civil society, chief
ideal, challenge, committee, president,	engineer, supreme court, chain smoker,

navy, security, administrator, target, formula, record, control, appeal, judge, bureaucrat, market, publicity, bus, power, criticise, start, programme, contract, media, budget, national.

deputy commissioner, whole sale, swimming pool, work permit, selection committee, network, result card, service centre, homework, book shop, registration form, adhoc judge, super star.

Source: Urdu Daily Express (07/02/2010)

The loans given in the table are very frequent; in some cases they are even more common than their native counterparts and are the only options e.g. *security* and *parking*. It is an irony of fate with Urdu that the names of the organisations and institutions, e.g. the *Daily Express* (Urdu newspaper) and *Urdu Dictionary Board*, related to promoting the Urdu language are themselves in English.

The integration of loans causes various forms of morphological changes, as illustrated below:

Table 3. Patterns of English Loans with Morphological Changes

Pattern of Loans	Examples
1. Gender Distinction	daktər (m) 'doctor' → daktər <b>ni</b> (f) 'lady doctor'
2. Pluralisation	bʊk 'book'→bʊk <b>ẽ</b> 'books'
3. Derivation	frad 'fraud' → fradıa/fradi 'one who gets benefits by treachery'
4. Hybrid compounds	phəl 'fruit' + frut 'fruit' → phəl frut '(a variety of) fruits'
5.Acronyms (English)	HEC - Higher Education Commission
6. Acronyms (Urdu)	PTI - Pakistan Terik-e-Insaf 'Pakistan Justice Movement'
7. Reduplication	taım taım (emphasis) 'with changing time'
8. Echo Reduplication	taım vaım 'time and the like'
9. Affixes	-ean: Lahorean 'citizen of Lahore', -ite: Karachiite

These loans seem well integrated and should be measured by the extent to which they are affixed with Urdu morphological markers. Various forms of morphological changes include initials, signatures, and acronyms e.g. HEC (Higher Education Commission), abbreviated forms of long English loans (Prof.), shortening of two word compounds, e.g. Indo-Pak (India-Pakistan) and reduplication e.g. *time time* 'with the passage of time'. Urdu acronyms are not traditional, but the English loan forms of acronyms are popular, e.g. *HEC* (*Higher Education Commission*) *Pakistan*. Even more interesting are the Urdu compound nouns, which follow the pattern of English acronyms e.g. *PTI* (*Pakistan Tehreek-e-Insaaf*) 'Pakistan Justice Movement' (a political party led by a cricketer turned politician, Imran Khan).

These forms of lexical use of English loans are not exhaustive but suggestive. Morphologically, they have had a minimal effect upon Urdu. It is the native affixes that establish a pattern, according to which a loanword is adapted. English loans in Urdu are restricted in regard to the addition of new items. For example, the loan nouns very often take the native Urdu gender and number markers, but they are unaffected by the Persian and Arabic gender and number marking. The derivational changes are also very limited to some loans, although the forms of changes may be various. The frequency of each change varies depending on the situation involved. The integrated loans and the adaptation depend both on the word class (verb, noun, adjective) and on similarity to loanwords from other languages already existing in the borrowing language. Certainly, loanwords used as nouns differ from those used as verbs, or adjectives, in relation to both morphological marking and their incorporation into borrowing language structures. Reed (1948) discusses the adaptation of English loans in Pennsylvania German Morphology. There are many similarities between the adaptation of English loans in the two languages. One common feature is the gender marking

of the borrowing languages. This is necessary, because both Urdu and Pennsylvania German have overt gender marking, whereas English does not.

### 5.2. Adaptation of Loans: Gender and Number Changes

#### 5.2.1. Gender Treatment and Gender Marking of Loans

This subsection discusses two points, firstly some general principles that determine the gender of English loan nouns in Urdu, and secondly the formation of feminine noun from masculine noun following a rule may be grammatical in some cases but ungrammatical in others. Urdu gender morphology comes from three sources i.e. native Urdu and Persian and Arabic. Any loan noun can be expected to adopt a pattern from either of these sources. The abundance of English loan nouns shows that they adopt the native Urdu gender morphology, which is based on not only sound but also meaning. General gender markers are -a and -i, as noted in Chapter 2. Therefore, all loan nouns that end with -a (m), e.g. sofa 'sofa' and -i (f) e.g. digri 'degree' are treated as masculine and feminine respectively. However, many other loan nouns do not end with -a/-i. In this case, the gender treatment is based on how their Urdu counter parts are taken, whether masculine or feminine. This is particularly necessary, if the loan nouns are inanimate nouns. For example, English loan bed is masculine and tebil is feminine only because their Urdu counterparts, bistor and mez, are masculine and feminine. However, it is difficult to determinie, if Urdu counter parts are not available and English loans are the only words used for certain objects. This is the option particularly for scientific objects and terms. In this case, Rzivi's (2007: 73) resolution, as noted in Chapter 2, is also helpful that huge, heavy, powerful, dominant and big objects are masculine, while small, weak and light ones are feminine. Therefore, refrigerator, computer, TV and engine are masculine and pin, ozone, oxygen and carbon are feminine in Urdu. These inanimate nouns do not have the general

native Urdu suffixes, -a and -i, for determining gender, and therefore their image emerging in mind is the only option left for it.

In addition to -*a* and -*i*, many English loan nouns may also have two other forms of gender, although limited in their use. Thus, there are three forms of nativised gender marking.

The loan nouns in (1a) are inanimate nouns and treated as masculine and feminine, based on their endings which are the primary gender markers in native Urdu. Examples (1b-1c) show that a larger number of English loan nouns, treated as masculine, have a zero suffix and a few may also have -i, both following the secondary native patterns of gender. The feminine patterns in (1b and 1c) are also derived forms in accordance with the native Urdu masculine patterns, and so end with -ni and -ən respectively. However, the masculine gender suffix -i and the two feminine forms -ni and -ən attach to only a few English loans, as they are unproductive in native Urdu also. It is noteworthy that the formation of some feminine nouns following a rule may be grammatical in some cases but ungrammatical in others.

- 2a) daktər (m) 'doctor'  $\rightarrow$  daktər**ni** (f)
- b) master (m) 'teacher'  $\rightarrow$  master**ni** (f)
- 3a) titfər (m) 'teacher'  $\rightarrow$  \*titfər**ni** (f)
- b) mænid3ər (m) 'manager'  $\rightarrow$  \*mænid3ər**ni** (f)

The feminine nouns in (2) take the gender marker -ni in accordance with the masculine nouns, which have a zero suffix for gender marking, as examplified above. However, this rule is not applicable in some other loan nouns, as in (3). This shows that -ni is not as productive as the major gender markers -a and -i, although there are a larger number of English loan nouns treated as masculine with a zero suffix. Secondly, Crawford's (2008: 64) view is also helpful here that the degree of nativisation of a loan correlates with the length of time it has been in a language and its frequency of use. The loan nouns in (2) are quite older than those in (3). They are not treated as loans, but as native words. This is why, possibly, they are used with the gender distinction. But the loans mouns in (3) are more recent and still treated as loans. Therefore, they are used without any gender distinction.

There is another aspect of the case. Note that the nouns in (2) have retroflex sound in the final syllable, while the nouns in (3) are devoid of it. Therefore, the former seem more adaptable and are more in harmony with the native Urdu nouns, for which a retroflex sound is a feature.

The feminine nouns in (3) do not seem to be adaptable, although they are morphologically similar to those in (2). However, it is not the case that if a loan noun is adaptable, the other loan with a similar morphology is also adaptable. The examples in general show the adaptability of some English loan nouns for native gender marking. Thus, the gender morphology of English loans is in conformity with the TLS. In the examples, there are two different processes at work here: one is the gender assignment to a loan word; the other is whether the loanwords can act as the base for further formations, through derivation, i.e. the change of gender from masculine to feminine. The process of derivation is discussed in the sections to come. Gender marking of English loans actually concerns their adaptability in plural marking with native Urdu affixes.

#### 5.2.2. Pluralisation of English Loans

As Winter (2008: 168) points out, Wegener (1999, 2002, 2003, 2004) argues that different plural forms e.g. those in Pennsylvanian German are integrated on a morphological level. The pluralisation of English loan nouns is the most frequent form of adaptation in Urdu, and many of its features are similar to those seen in Pennsylvania German (PG). For example, many English plurals have also entered Urdu, but not all are very commonly used. The plural morphology of English loan nouns shows changes more like those of native nouns. The loan stems are affixed with the plural morphemes, i.e.  $-e(m) - ij\tilde{a}$ ,  $-\tilde{e}(f)$  (nominative forms) and  $-\tilde{o}$  (oblique form). They are the general native Urdu plural markers. The following examples of masculine and feminine plurals in Tables (4a-4d) show that English loans are well integrated. Therefore, unlike Persian and Arabic, there is no morphological ground to differentiate between the pluralisation of the native nouns and that of the English loans.

There are two points to be noted in respect of oblique singular/plural forms. Like the singular oblique of the native masculine nouns, the singular oblique of English masculine nouns is the same as the nominative plural, i.e. it ends with -e. The oblique form of all other singular loan nouns is unchanged like their native counterparts. The oblique form of all loan plurals also follows the native oblique form and ends with -õ. Each of the four sets of masculine and feminine plurals contains one native Urdu example so that the morphological changes in the English loans can be compared and better understood:

Table 4a: Pluralisation of English Loans: Masculine Nouns (ending in -a)

Singular	Nominative Plural	<b>Oblique Plural</b>
1- deta 'data'	→ dete 'data'	/ det <b>õ</b>
2- sof <i>a</i> 'sofa'	→ sofe 'sofas'	/ sof <b>õ</b>

3- formula 'formula'	→ formule 'formulas'	/ formul	lõ
4- propaganda' 'propaganda'	→ propagande 'propagandas	s'/ propag	;and <b>õ</b>
5- pesta 'pesta'	→ peste 'varieties of pasta'	/ pest <b>õ</b>	
6. bet <i>a</i> 'son'	→ bete 'sons'	/ betõ	(Urdu)

Table 4b: Pluralisation of English Loans: Masculine Nouns (all other endings)

Singular	Nominative Plural	<b>Oblique Plural</b>
1- daktər 'doctor'	→ daktər 'doctors'	/daktər <b>õ</b>
2- kəmısən 'commission'	→ kəmı∫n 'commissions'	/ kəmı∫n <b>õ</b>
3- dʒəg 'jug'	→ dʒəg 'jugs'	/ dʒəg <b>õ</b>
4- pepər 'paper'	→ pæpər 'papers'	/ pæpər <b>õ</b>
5- mænidʒər 'manager'	→ menedʒər 'managers'	/ menedʒər <b>õ</b>
6- bhəngi 'toilet cleaner'	→ bhəngi 'toilet cleaners'	/ bhəŋg <b>iõ</b> (Urdu)

Many English loans treated as masculine (4a) end in -a. Their pluralisation is made by the alternation of the suffix -a with -e. The patterns of loan plurals, e.g. formula 'formula'  $\rightarrow$  formule 'formulas' are exactly the same as the native plural, e.g. beta 'son  $\rightarrow$  bete 'sons'. In set (4b), comparing the loans, i.e. (1-5) with the native one (6), shows that many English loan nouns remain unchanged like their native counterparts in their plural forms. This means that masculine pluralisation by alternation of vowel is possible only if masculine loans end with -a. If they end with something other than -a, there are no changes for pluralisation.

The singular oblique patterns of the loans, following morphology of the native nouns, are the same as the nominative plurals. For example the nominative plural *formule* 'formulas', in set (4b), is also the oblique singular noun. There is no difference in the oblique plurals of the loan masculine nouns in the two sets (4a & 4b). The oblique plurals end with the suffix - $\tilde{\mathbf{o}}$ . This

gives the picture that the loan oblique plurals, e.g. *formul***o** and native oblique plural, e.g. *bet***o** (native) are also the same in forms. However, there is another possibility from the examples in set (4b). The oblique form of loan plural nouns, e.g. *kttfin* 'kitchen' may also remain unchanged. The oblique plurals in general end with the suffix -**o**, and there is no difference between masculine or feminine loan nouns. It is the same as the native nouns. The English feminine nouns (4c and 4d) also show this:

Table 4c: Pluralisation of English Loans: Feminine Nouns (ending in -i/-ni)

Singular	<b>Nominative Plural</b>	<b>Oblique Plural</b>
1- laıbrer <b>i</b> 'library'	$\rightarrow$ larbrerij $\tilde{a}$ 'libraries'	/ laɪbrerɪj <b>ő</b>
2- junivəsiti 'university'	$\rightarrow$ juniversities' universities'	/ junivəsitij <b>õ</b>
3- digri 'degree'	$\rightarrow$ digrij $\tilde{a}$ 'degrees'	/ dīgrīj <b>õ</b>
4- edʒənsı 'agency'	$\rightarrow$ edzəns <b>ij<math>\tilde{a}</math></b> 'agencies'	/ edʒənsij <b>õ</b>
5- daktər <b>ni</b> 'lady doctor'	$\rightarrow$ daktər <b>nij<math>\tilde{a}</math></b> 'doctors'	/daktərnıj <b>õ</b>
6. beti 'daughter'	$\rightarrow$ betij $\tilde{a}$ 'daughters'	/ betij <b>õ</b> (Urdu)

Table 4d: Pluralisation of English Loans: Feminine Nouns (all other endings)

Singular	Nominative Plural	Oblique Plural
1- plet 'plate'	→ pletẽ 'plates'	/ plet <b>õ</b>
2- bildin 'building'	$\rightarrow$ bilding $\tilde{e}$ 'buildings'	/ bɪldɪŋg $ ilde{m{o}}$
3- buk 'book'	→ bʊk <b>ẽ</b> 'books'	/ b $v$ k $ ilde{m{o}}$
4- məʃi:n 'machine'	$\rightarrow$ məʃin <b>ẽ</b> 'machines'/ məʃi	$\ln  ilde{\mathbf{o}}$
5- fradən 'cheater	$\rightarrow$ fradən $\mathbf{\tilde{e}}$ 'cheaters'	/ fradən $ ilde{oldsymbol{o}}$
6. bhəngən 'toilet cleaners'	→ bəŋgəne 'toilet cleaners'	' / bəŋg <b>ənõ</b> (Urdu)

In set (4c), English feminine nouns ending in -i, e.g. laubreri 'library', are pluralized as laubrerija 'libraries', the same as the native Urdu plural betija. Further, the oblique plurals in every case continue to end with the suffix  $-\tilde{o}$ . However, set (4d) shows a different picture. Like many native feminine nouns, e.g. bhangan 'toilet cleaner', there are also English loan nouns, which are treated as feminine and do not end with -i, e.g. bilding 'building'. They are pluralized with the suffix  $-\tilde{e}$  e.g.  $bilding\tilde{e}$  'buildings'  $/bangan\tilde{e}$  'toilet cleaners'. All the loan examples in sets (4c and 4d) are evidence that there is a full integration of English loan nouns. The two sets show a general rule, namely, if the English loans end with -i, the plural ends with  $-ij\tilde{a}$ . If the loans end with something other than -i, the plurals end with  $-\tilde{e}$ . Both the sets show that whether masculine or feminine, the oblique plurals end with  $-\tilde{o}$ . Singular oblique remains unchanged in all nouns except for those given in set (4a), the same as the nominative plurals.

Forming plurals in Urdu is far more complex than in English. As seen in chapter 2, there are three types of plural patterns, i.e. native Urdu, Persian loan and Arabic loan patterns. In this scenario, English loan pluralisation seems to be according to native Urdu patterns. Therefore, applying Winter's (2008: 156) criteria of conformity, it can be observed that the English loan nouns are in conformity with TLS. There are some notable features.

Even though the pluralisation of English loans is based on native Urdu plural by suffixation, the suffixation itself is no more than an alternation of vowels. The alternation of vowels is the main feature in the gender and number morphology and the case morphology of the Urdu noun. Phonological alternations come in many shapes and sizes, and the processes behind them are equally varied, as are the kinds of factors which condition them (Davenport and Hannahs, 2005). The alternation forms nativised English plurals both with and without nasalization. With the nasalization, it makes changes at the suffix position, e.g. bok/ boke

'books', as in sets (4b,4c and 4d) to form feminine plurals. To form masculine nominative plurals, the alternation of vowels at the suffix position does not require nasalisation, as in (4a). The oblique forms of plurals however require nasalization. The alternations are mainly caused by nasalised vowels and an allophone /~/of the alveolar nasal consonant /n/. There are a number of certain cases of nasalization in the alternation of vowels. One is grammatical (morphological), i.e. forming plurals, another is lexical, and still another is semantic. However, the lexical and semantic cases are not the issue here.

In addition to the alternation of vowels, all the plurals in the four sets (4a to 4d) also show the productivity of native Urdu plural suffixes and the adaptability of English bases. The plural suffixes are productive enough to attach to any of the loan nouns. Productivity is mainly said to be a function of the frequency of words, and the reason for the connection between frequency and productivity lies in the processing of words in the lexicon. As noted above, many English nouns are gradually becoming more frequent than the native ones. They are adaptable in native Urdu pluralisation. Dressler (2003: 32) postulates productivity as a primitive property of inflectional morphology (in parallel to grammatical productivity), and states, "if we conceive of productivity as a grammatical concept, then morphological productivity can still be defined...as: the possibility for language users to coin, unintentionally, a number of formations which are in principle uncountable." Discussing Wurzel's (1984) productivity rules, he mentions two important elements 1) the integration of loanwords with unfitting properties, which have to be fitted (accommodated) to the system adequacy of a borrowing language; 2) the integration of loanwords with the fitting criteria. The first, the integration of loanwords with unfitting properties is noted in derivation, as discussed in the relevant section ahead. For the integration of loan-words with already fitting properties, the integration must overcome only the obstacle of foreignness. Loanwords with

fitting properties here mean that they are adaptable and can be adjusted in the morphological system of borrowing language. One example, Dressler (2003) gives, is the entrance of German neuter -o into Slavic languages, where -o is the default ending of neuters (Sg.Nom. = Acc.) and the respective microclass the default for neuters: e.g. German das Auto > Polish auto (neuter), Slov. auto (masc.). German. Auto (neuter) becomes a neuter in Polish; but a masculine in Slovene demonstrates non-productivity of the Slovene neuter microclass in -o, but high productivity of its Polish correspondent. In the light of this, the pluralisation of English loans in Urdu can also be noted as showing the integration of loanwords with fitting properties. English loan nouns have fitting properties and are fully compatible in the Urdu plural system. They accommodate the Urdu gender system as well as the number system. Thus, the native suffixes are productive enough in these two aspects. They accommodate English bases without requiring any major changes, i.e. deletion or substitution of any other affixes. This shows that English bases can easily be extended by the addition of new items (the Urdu plural markers), and therefore seem to have been fully integrated into the Urdu pluralisation process with gender distinction, generally.

Gender distinction is visible in the derivation of words also. Derivational affixes possess different degrees of productivity, but this is only partially seen in the English loans. It is not even as much as the productivity of Urdu plural markers. However, the derivational patterns possess a variety of formations.

#### 5.3. Adaptation of Loans by Derivational Means

The adaptation of English loans by derivation can be seen both by affixation and by compounding. This section discusses three major points 1) derivation by affixation 2) derivation by compounding 3) English loan affixes. Dressler (1993: 5028) considers that

morphological innovations (even if derived via the productivity derivation rule) always violate (at least a part of) lexical norms. This violation of norm is also evident in the derivation between Urdu and English morphological items. The major word-formation processes by English loans include affixation, compounding and acronymy, but reduplication, blending and clipping are also evident.

#### **5.3.1. Derivation by Affixation**

Urdu affixes are restricted in derivation as compared to gender and number marking. Some linguists hold the view that productive ways and means of word-formation are only those that can be used for the formation of an unlimited number of new words in a modern language. Bauer (2003) states, "a morphological process can be said to be more or less productive according to the number of new words which it is used to form". In light of this, Urdu derivational affixes are not as productive as gender and number affixes, which is nevertheless explicable in terms of grammatical marking in general: a noun in Urdu must be marked for gender and number. Generally, we don't see a large number of derivational patterns with English loan bases, although a variety of derivational changes can be seen. Moreover, Urdu borrows only one (or two) categories of a word. Other derivations of the same category can rarely be seen. For example the word *fraud* is very commonly used but its other categories i.e. *fraudulent* and *fraudulence* are unknown. Derivations from the loanwords are mostly in the native Urdu style, although some Persian and Arabic affixes are also productive.

#### 5.3.1.1. English Loanwords with Native Urdu and Persian/Arabic Derivational Affixes

This section discusses mainly native Urdu affixation, but some English loanwords taking native Urdu, Persian and Arabic affixes are also highlighted. The following section then focuses on Persian and Arabic affixation separately. As stated above, most word formations

are the result of native Urdu affixation, which may be productive in just one meaning because that meaning is specially needed by the community at a particular time period historically. It can be non productive to coin scientific or technical terms. Only a few English loanwords, e.g. <code>park/parking</code>, and <code>administration/administrator</code> are borrowed with more than one category. Most loans are used in the original forms borrowed and the derivations, if any, are mostly made in the native Urdu manner, although native affixes are fewer than Persian and Arabic affixes. They do not only derive new words from English loans, rather they create new lexical items with loan bases. However, some Persian and Arabic affixes are productive enough to derive quite many words with English loans. This is illustrated in the following data:

Table 5: Some English Loans and their Adaptation in Urdu

```
a. frad 'fraud' (N)
                               1) fradija 'one who gets benefits by treachery' (N) (m)
                                fradije 'cheaters' (plural and sing. obl) / fradijo (pl.obl.)
                               2) fradi 'cheater' (N/A)
                                 fradi 'cheater' (plural and sing. obl) / fradijo (pl. obl.)
b. bəm 'bomb' (N) \rightarrow
                             1) bəmbar 'bomber' (N/A) (m) (used in native manner)
                             2) bəmbari 'bombardment' (N) (f)
                               bəmbarıjā 'bombardments' / bəmbarıjō (pl. obl)
c. fəlsəfa 'philosophy' \rightarrow
                             1) fəlsəfe (sing. obl./pl) / fəlsəfő (pl. obl) (m)
                             2) fəlsəfi 'philosopher' (N/A)
                             3) fəlsəfijana 'philosophical' (A)
                             4) fəlsəfijət 'status of a philosophy' (N)
d. dzografija 'geography' (N) 1) dzografijai 'geographical' (A)
e. titsər 'teacher'
                                titsəri 'teaching' (N)
  daktər 'doctor'
                               daktəri 'doctorship' (N)
```

The loans undergo all the possible inflectional and derivational changes through which a native noun can undergo. Some examples of major changes, from Table 5, in comparison with Urdu counterparts are illustrated below:

vehsi 'barbarian'----- vehsijana 'wild'-----vehsijat 'status of wilderness' (U)

vehsi 'wilderness'

In the given examples, (5a) shows inflectional changes based on native morphology; (5b) also exhibits native based derivational and inflectional changes, and (5c) Persian and Arabic based derivational changes. Almost all loans in the table seem to be integrated, as they show various changes of categories and forms. The changes in the loan noun *frad* 'fraud' (5a) are in two different patterns, i.e. *fradnja* and *fradi* 'one who gets benefits by treachery' (both) which are new lexical creations. They are in conformity with the target language patterns only, and their semantic substitute does not exist in English, as a derivation of *frad*. This shows that native Urdu affixes, with English bases, can create new lexical items rather than merely deriving new words. Lexical creation is more important than the derivation of new words in that it introduces a new lexical item into a lexicon, while aderivation does nothing but changes the

category of already existed word in a lexicon. Both fradija and fradi are new introductions to Urdu lexicon. This shows the complete integration of some English loanwords in Urdu morphology. Both the forms are in concordance with those seen in the native patterns and are in common use. The first one, i.e. fradija is entirely parallel to the native changes, e.g. dak 'mail' (f)  $\rightarrow dakija$  'postman'. This change is from a (m/f) noun to form an agentive masculine noun showing the male specific job or profession. A noun undergoing morphological changes may be masculine or feminine, e.g. the English fradija or the native dakija must be a masculine noun. In addition, it has no feminine form.

The second formation frad 'fraud'  $\rightarrow fradi$  'one who gets benefits by treachery' takes the native derivational affix -i, but the morphological changes in fact parallel the changes in Persian loanwords, e.g. mehnot 'hard work'  $\rightarrow mehnti$  'hard worker'. Both fradi and mehnti are used as a noun and an adjective, and there seems to be no distinction of gender. However, there is a possibility that the English noun takes the feminine form fradon, although it is not traditional. The newly derived noun may not necessarily be agentive.

The morphological changes in all the examples from (5b) onwards show that Urdu borrows one category words and derives, if it does, all other categories with mostly native affixes in a native manner. For example, English noun *bəm* 'bomb' (5b) exhibits much morphological and phonological fusion. Morphologically, the noun is further converted into the noun/adjective *bəmbar* 'bomber', noun *bəmbari* 'bombardment' and a verb *bəmbari kərna* 'to bombard/to do bombing' (for which a dummy verb is required, which will be discussed later). Two of the nouns *bəm* and *bəmbar* are treated as masculine, but *bəmbari* is feminine. All these nouns take nominative and oblique plural morphology, just like the native Urdu

nouns do. This shows that the integration is the same as the integration of Persian-Arabic loans with various morphological changes. However, the number of such English loans is comparatively not large.

The integration of most English loans as in (5c-5d) is generally not so much with any affixes whether native or Persian/Arabic. However, like those in (5a) and (5b), these changes verify the perception that English loans have started undergoing some integration, similar to the ones seen in Persian/Arabic loans, although to a far smaller extent than the older ones. The loan folsofa 'philosophy' (5c) can be pluralised as folsofe. It can also take the suffix -i for the formation of an agentive noun/adjective folsofi 'philosopher'. It shows morphological changes in the bases with even a higher degree of integration by taking Persian affix -ijana and Arabic affix -ijat to form an adjective folsofijana and an abstract noun folsofijot 'status of a philosophy'. It is thus unique in the sense that it takes the native suffix -i, it takes the Persian suffix -ijana, and then it can also Arabic suffix -ijat.

The changes in (5e) and (5f) are not various. They are from a noun to noun and noun to adjective. The nouns in (5e) show an agentive noun, e.g. *titfər* 'teacher' changing into an abstract noun *titfəri* 'teaching'. Like *fəlsəfa*, the noun *drama* in (5f) is unique in the sense that it can be affixed with native and Persian/Arabic derivational affixes. Affixing with the native - *i*, it is changed into the adjective *dramai*, but affixing with Persian and Arabic, it is changed into agentive nouns i.e. *dramanıgar* and *dramanəvis*. Thus, it shows more affixation of Persian morphemes than native morphemes.

All the morphological changes generally show the productivity of native derivational suffixes and the adaptability of English loans. The degree of productivity of affixations very much depends on the structural, lexico-grammatical and semantic nature of bases and the meaning

of an affix. The adaptability of English loans in pluralisation is widespread, but by derivational means it is not seen on a very large scale. However, various derivational patterns can be seen. There are changes in various loans, but each loan is restricted to one change in most cases. Those which show a high degree of integration and complete morphological adaptation are only some words. Complete morphological adaptation means a loan, e.g. *frad*, is able to take a maximum number of possible morphological forms, the same as shown by Persian and Arabic loans, and it looks like a native word. It should show the gender, number changes in the native manner; it should also show the derivational changes in the native manner. With this explanation, it can be said that the loans in (5a-5c) are completely integrated, although those in (5d-5f) are partially integrated.

The ability of the loans to take all forms of structural changes shows that some English loans are fully integrated, and the affixation or inflection is according to native (target language) patterns. The inflectional and derivational changes are both major and minor, and the word formations include all major categories of words i.e. nouns, adjectives and verbs. The new words derived from the English loans seem to be designed systematically, as they follow the native Urdu morphological rules exactly, to make the changes. Plag (1999) terms it the morphological productivity that can be defined as "the property of a given word formation process to be used to derive a new word in a systematic fashion". The newly derived words from English loans are the extended forms and the results of several applications of the word-formation rule in a systematic fashion, as the examples of *frad* 'fraud' shown ahead. The stems of the words form derivational relations in different degrees. What is meant by 'degree' is that the stem of a word, e.g. *art* takes various affixes, one after the other, to derive various formations, e.g. *artistic artistically*. The English loan *frad* 'fraud' in base form shows the zero degree affixation. Then, by applying a suffix -i or -ija to the base, there is first degree

affixation converting an abstract noun into an agentive noun *fradi* or *fradija* 'cheater'. The application of second degree affixation to *fradija* changes it into nominative plural and the singular oblique form *fradije*. The noun takes plural oblique form as *fradijo* in its third degree of affixation. Thus, it can be suggested that in the inflectional and derivational process the English loans are generally adaptable. They can be extended into a variety of forms in the Urdu morphological system, and some of them are completely integrated.

Generally, an integrated or non integrated loanword exists side by side with its native equivalent so that the substitution can be easily identified. It can be true for words like bəmbari 'bombing' but not for fradıja 'cheater'. The native substitute, i.e. golabari 'bombing' for the nativised loan bəm 'bomb' does exist but is rarely used. It may be considered obsolete now, because the nativised loan bəmbari has replaced it. The native substitute for fradıja may be <code>dhokabaz</code>, which is in fact used for 'cheater'. It is fully semantically compatible with fradıja. The English loan is specific in meaning and implies cheating in terms of money, but the native noun refers to cheating in a general sense. This semantic production is also evidence that some English loans are fully integrated and are like native Urdu words.

The changes in (5a)-(5c) bring us to the conclusion that English loans have started taking the status of Persian and Arabic loans, which are so fully integrated (phonologically, morphologically) in Urdu that they seem to be original Urdu words. Their morphological changes are the same as those in native words and Persian-Arabic loans. Words like *frad* contain retroflex sound, which is the specific feature of native words, not found in Persian-Arabic loans. Therefore, the changes in *frad* like English loanwords seem to be more native than in Persian and Arabic loans. The structural changes and the convergence of the loans are

extreme, involving some types of interaction and mutual accommodation by English loanwords and native Urdu affixes.

The structural convergence is similar to the one Winford (2003: 11) notes in Northwest New Britain where languages of the Austronesian and non-Austronesian families and new contact languages i.e. pidgins, creoles, and bilingual mixed languages have become structurally isomorphic i.e. they are ancestrally different, though, their structures are similar. He considers that borrowing is sometimes "the incorporation of foreign features into a group's native language by speakers of that language...the borrowing language is maintained, though changed in various ways by the borrowed features, and that the agents of change are its native speakers." Based on this criterion, it can be said that the English loans are at the maintenance stage. They have converged into native form. However, it is just the convergence of some lexical items, despite the fact that a large number of English loans have become members of the Urdu lexicon. Like the examples in (5d-5f), there are many English loans which, though integrated with Urdu affixes, do not undergo all the derivational changes. They take some forms, but do not take others. Consider the following examples:

```
4a. drama 'drama' (N) → dramai 'dramatic' (A)
b dʒʊgrafija 'geography' (N) → dʒʊgrafijai 'geographic' (A)
c. edʒənda 'agenda' (N) → *edʒəndai
```

The two nouns (4a & 4b) ending with -a can be converted into adjectives by suffixing with -i, but the third noun (4c) is ungrammatical with the same suffix. Discussing the productivity of the English derivational suffix -ist, Katamba (1993: 68) states, "We do not have a carte blanche to use it with any noun. There are unexplainable gaps. For example, a follower of prophet Muhammed is not a *Muhammedist* though the follower of Budha is a *Buddhist*." The

adaptation of English loans with Urdu affixes also shows the same unexplainable gaps and can be widely seen. Although the native Urdu derivational affixes, particularly -i, are productive and form new words with English loan bases, they do not always function as shown in (4c). There is no explanation as to why they do not form new words. The degree of productivity of affixation very much depends on the structural, lexico-grammatical and semantic nature of the bases and the meaning of the affixes. In general, it can be said that native Urdu affixes are productive and the English loans are adaptable, and so integration occurs. Some loans, e.g. *drama* in (4a), are adaptable more with Persian and Arabic affixes than with native Urdu affixes, as shown in the next section.

# 5.3.1.2. English Loanwords with Persian and Arabic Affixes

Some English loan nouns, e.g. *drama* (4a), are unique in their adaptation in that they take native Urdu, Persian and Arabic derivational affixes. Affixing with the native -i, the noun *drama* is changed into an adjective *dramai* 'dramatic', but affixing with the Persian -*mgar*/-*nəvis*, it is changed into an agentive nouns i.e. *dramanıgar*/ *dramanəvis* 'dramatist', and affixing with Arabic -*ijət*, it is converted into an abstract noun *dramanət* 'the state of being dramatic'. Although these are only a few examples, it is interesting to see some patterns of Persian/Arabic affixes attaching to English loans separately from native based changes:

**Table 6: English Loanwords with Non Native Affixes** 

a. laın 'line'	$(N) \rightarrow$	l <i>a</i> ın <b>dar</b>	'(paper) having lines'	(A)
b. fəlsəfa 'philosophy'	$^{\prime}(A) \rightarrow$	fəlsəfi <b>jana</b>	'philosophical'	(A)
c. bor 'bore'	$(A) \rightarrow$	1) bor <b>ıjə<u>t</u></b>	'boredom'	(N)
	$\rightarrow$	2) bor kərna	'to bore'	(V)
	$\rightarrow$	3) bor <b>hon</b> <i>a</i>	'to be bored'	(V)

d. drama (N) 
$$\rightarrow$$
 1) drama**rjət** 'the state of being dramatic' (N)  $\rightarrow$  1) drama**nıgar** 'dramatist' (N)  $\rightarrow$  2) drama**nəvis** 'dramatist' (N)

The English loan nouns *lam* 'line' (6a) and *fəlsəfa* 'philosophy' (6b) change into an adjective. Both take two distinctive Persian affixes. The noun *fəlsəfa* is unique in the sense that it may take native Urdu, Persian and Arabic affixes. The change in *bornjət* (6c) is also unique but different from the first two in that a loan adjective takes an Arabic suffix *-njət* for the formation of a noun, and it takes dummy verbs for the formation of a verb. The noun *drama* in (6d) is the most interesting among the examples in that it takes native Urdu, Persian and Arabic affixes. It changes into all major categories i.e. noun, adjective and verb.

Although English loans are generally adaptable, some of them don't accept Urdu affixes freely. Various affixes e.g. -tja, tjət, -tjana shown in (5a-5b) attach to some items e.g. dʒʊgrafija, titfəri, dramatjət as in (5d-5f). However, the affixation of many others is ungrammatical with the same lexical item e.g. \*dʒʊgrafijana. In general, the attestation of all these affixes is not possible, and the integration of the English loans is generally limited with Persian and Arabic affixes, and mainly occurs with native Urdu affixes. Therefore, it can be said that, although English loans show their adaptability and some variations, they do not generally integrate with Persian and Arabic affixes. Those given are among a few examples. As Winford (2003: 24) points out, early scholars such as Müller (1875) and Jakobson (1938) were of the view that structural (linguistic) constraints were the primary determinants of contact-induced change. Arguing against them he emphasises that, "the wide body of evidence available to us now shows that practically any linguistic feature can be transferred from one language to another, if the circumstances are right. The reason is that extra linguistic

factors – the social ecology of the contact situation itself can override any purely structural resistance to change." In this scenario, we see that the modern vernacular language is based on the *Khari Boli* dialect, of the Delhi region, spoken in 12<sup>th</sup> century, although the name *Ordu* was first used much later in 17<sup>th</sup> century. Since then, Persian and Arabic linguistic influence began with the arrivals of Afghan emperor Mehmood Gaznavi and then Mughals, as two languages got much importance. Therefore, the integration of Persian and Arabic lexical items and their affixes was convenient. However, the same Persian and Arabic affixes have not proved to be productive in their integration with English loans, which are adaptable enough with the native Urdu affixes. They adjust English loanwords very to some extent, as compared to native Urdu affixes. The social setting, the circumstances and the linguistic wellformedness do not allow the integration of English loans in every aspect of its morphology.

Analysing the English loans based on Wegner's criteria (cf Winter, 2008: 168)), it is observed that most loans, as in Table (2), are without any morphological changes, and therefore the zero degree inflection is according to a non-native pattern. The derivation by zero degree inflection is ascribed to simple loans whose stem is homonymous with a word-form and often with a root-morpheme. They are simple lexical items with no derivational changes. Secondly, there are also many English loan nouns, e.g. *titfəri* 'teaching' (5e) whose derivational changes follow native rules. Therefore, the affixation of a morpheme is according to a native TL pattern. Thirdly, all the English loans taking Persian/Arabic derivational affixes cannot be counted as fully integrated, as they neither follow native (Urdu) nor non-native (English) morphological rules. Rather, they follow Persian/Arabic rules, and therefore such loanword integration is according to a peripheral native TL pattern, as defined by Winter (2008: 158).

Although such loanword integration and derivation by affixation, according to the peripheral native TL pattern, shows a few formations, the derivation by compounding is vast enough. The major categories of English loanwords combine with Urdu dummy verbs to derive verbs.

#### 5.3.2. Derivation by Compounding

Derivation by compounding of English loans is seen in hybrid compounds. Most hybrid formations of English loans are not with the native Urdu words but with Persian/Arabic loans. These formations may be divided into two groups, i.e. hybrid compound nouns/adjectives, and compound verbs. Both compound groups show that English loanwords are adaptable with Persian and Arabic loanwords, already exist in Urdu, for the formation of compounds. The following subsection focuses on compound nouns and adjectives.

## 5.3.2.1. Hybrid Compounds: Noun and Adjective Formations

Hybrid compounds are the formation of two or more constituents from different languages. As pointed out earlier, Capuz (1997: 8) considers the terms *hybrid* or *loan blend* for them and states, in Urdu, there are a large number of hybrid compounds formed by one constituent borrowed from English and the second from Urdu, e.g. dabal 'double' (E) + roti 'chappati' (U)  $\rightarrow dabalroti$  'bread'. They are the third most frequent example of English loanword adaptation after pluralisation and compound verbs.

Regarding hybrid compounds in Pennsylvania German, Schach (1948: 121) cites Lambert's (1924) list of over a hundred hybrid compounds, in which the constituents from different languages may be wholly or partly integrated. For example, one loanword *fens* completely supplants the native words, but another loanword *hendal* only partly replaces Pennsylvania German words, as in *griffl handgriff*. There is no partial integration of the Urdu/English constituents, however, the compounds sometimes reflect the assimilation of the constituents.

For example, *səbzimarkit* 'vegetable market' (8d), given below, is so commonly used in Urdu that the two constituents, i.e. *səbzi* 'vegetable' (Persian) and *markit* 'market' (English) seem to be assimilated with each other, so much so, that the compound seems to be a single entity. Sometimes, constituents may be synonyms and similar in meaning. Like those in Pennsylvania German, there may also be a division into two groups depending on those designating objects for which there are no Urdu words, and those which exist beside an Urdu doublet e.g. *phəl frut* 'a variety if fruits'. The Urdu constituent in the compound is semantically an Urdu doublet of the English constituent and both have the same meaning.

A hybrid combination of a native and a loanword is a general phenomenon. As there is a heavy influence of Persian and Arabic on Urdu, naturally there is a possibility that an English loan may blend with a Persian or Arabic word already present in Urdu. Classifying the hybrid formations of English loanwords with native Urdu, Persian and Arabic, it is interesting to see that most hybrid formations are those with Arabic loanwords and then Persian loanwords. In Kent's (1999) terms such hybrid formations as pseudo loanwords. The examples in the tables (8)–(10) show that most hybrid formations are pseudo type of loan formations. However, the combinations with native Urdu words as well as original English compounds are also seen. Most compounds are nouns, and refer to food, administration and institutions etc.

#### **Table7. Hybrid Compounds: NN formations**

a - b (NATIVE)—NN/NV

a. phal 'fruit' (N) + fruit 'fruit' (N) 
$$\rightarrow$$
 phal frut '(a variety of) fruits' (N)

b. ræl 'rail' (N) + gari 'vehicle' (N) 
$$\rightarrow$$
 ræl gari 'train' (N)

c. mader 'mother' (N) + t $\int$ od 'fuck' (V)  $\rightarrow$  mader t $\int$ od 'abusive remarks for somebody' A/N c - d (PERSIAN)

c. tʃikən 'chicken' (N) + kəbab 'roast meat' (N)  $\rightarrow$  tʃikən kəbab '(a dish of) roasted chicken'

d. 
$$s \ni bzi^{13}$$
 'vegetable' (N) + markit 'market' (N)  $\rightarrow$   $s \ni bzimarkit$  'vegetable market' (N)

e - f (ARABIC)

e. siti 'city' 
$$(N)$$
 + nazim 'administrator'  $(N)$   $\rightarrow$  siti nazim 'city mayor'  $(N)$ 

f. səfəha 'page' (N) + nəmbər 'number' (N) 
$$\rightarrow$$
 səfəha nəmbər 'page number' (N)

## **Table 8. Hybrid Compounds: AN formations**

a - (PERSIAN)

a. Imtehani 'examination' (A) + farm 'form' (N)  $\rightarrow Imtehani$  farm 'examination form' (N)

b-d (ARABIC)

b. behri 'naval' (A) + kəmando 'commando' (N) → behri commando 'naval commando' (N)

c.  $t \text{ fif 'chief' (A)} + \text{ sahib 'Mr.' (N)} \rightarrow t \text{ fif sahib 'a title formerly given to a politician'}$  (N)

d. qomi/subai 'national/provincial' (A) + əssembli 'assembly' (N) → qomi/subai əssembli 'national/provincial assembly'
 (N)

#### Table 9: Hybrid Compounds: NA

a - b (ARABIC)—NA

a. sekretəry 'secretary' (N) + xardʒa 'foreign' (A) → sekretəry xardʒa 'secretary of foreign affairs' (N): Similarly: sekretəry xəzarna 'secretary of finance', sekretəry daxla 'secretary of internal affairs'

b. rope 'rupees' (N) + fi 'per' + kilo 'kilo' (A)  $\rightarrow$  rope fi kilo 'rupees per kilo' (N)

175

<sup>&</sup>lt;sup>13</sup>. Although used in many south Asian languages, *səbzi* is originally a persian word for vegetable (Persian online dictionary: *Aria Dic--* http://api.ariadic.com/?word=&lcode=20)

Most combinations are NN and AN, but there are some examples of NA and NV as well. Classifying the hybrids within these categories, it can be seen that the NN class has English loanword combinations with native Urdu words (7a–7b), Persian loanwords (7c–7d) and Arabic loanwords (7e–7f). Most of these compounds refer to food items, transport, and administration. The AN class is mostly formed with Arabic loanwords (8b-8d) and then Persian loanwords (8a). Semantically, these compounds generally refer to political institutions. The NA and NV classes may be formed with native Urdu and Persian loans, but most structures are formed with Arabic loans. They mainly refer to government administrative machinery (9a). One formation (9b) refers to an interesting fact that a compound may also be a trilateral structure, i.e. a formation of native Urdu, (Arabi/Persian) and English loanword. Here *rope* is plural of Sanskrit based word *ropeja*, *fi* is Persian and *kulo* is an English loan.

Like general Urdu compounds, most compounds with English constituents are from the endocentric class. The other classes of compounds, the copulative compounds *siti nazim* 'mayor', are few in number. The patterns also reveal that the English loanword may appear both as a head and a modifier disregarding whether the compound is right headed or left headed. For example, *qomi əssembli* 'national assembly' (8d) is right headed, but *sekretəry xardʒa* 'secretary of foreign affairs' (9a) is left headed. The English loan constituent is the head and the native one is the modifier in both compounds.

There are also some compounds, e.g.  $ræl\ gari$  'train' in which the English constituent may function as a modifier and the Urdu constituent as the head of the hybrid compound. The two compounds, i.e.  $ræl\ gari$  and  $siti\ nazim$  show that whether the head constituent is the native gari or a Persian/Arabic loan nazim, the appearance of an English constituent as modifier also forms a variety of compounds. These examples are similar to the ones Schach (1948: 123)

observed in hybrid compounds in Pennsylvania German. For example, he considered the English loanword in compounds, e.g. *ke spai* 'cheese pie' as the final or basic element, but others, e.g. *paigrust* 'pie crust', *pai* is the descriptive or definitive element.

AN or NA compounds in Tables 8 and 9 need some explanations that they are compounds and not just an adjective modifying a noun syntactically. Hüning (2008) considers that the sharp distinction between syntactic structure of an adjective modifying a noun and morphological word formation by an adjective and a noun, found in many grammatical theories, fails to account for the functional similarities between phrases and A+N compounds. However, there is a division of functions between syntax and word formation. Some of the functional differences, he mentions, are as follow: 1) the naming function is typical for the compounds. It is the task of compounds with an adjectival first element to characterize or set apart something as a special category. In contrast, phrases have the task not to name, but to describe entities. Some of the phrases in Dutch lose their inflectional marking at the end of the adjective when used as names. 2) Compounds are stressed on the first constituent, phrases on the second e.g. bláckbird vs. black bírd. 3) Compounds are written as one word, phrase as two. Thus it is often difficult to define the distinction between compounds and phrases, but there are certain elements e.g. naming, stress pattern and orthography. Looking at Tables 8 and 9, the examples given only seem to show names of some animate or inanimate objects, although the other criteria Hüning (2008) defines, i.e. stress patterns and orthography, may be seen in other compounds. Unlike English, stress pattern plays no role. There may be some formations orthographically seem to be single words, but there is non among the given ones. One point of difference is that the adjectival constituent is not the first element, whether it is an English or Urdu constituent. The compounds formed may be left branching e.g. qomi *assembli* 'national assembly' or right branching e.g. *sekretary xardʒa* 'secretary of foreign affairs' disregarding the position of English constituent.

Hybrid compounds are sometimes formed due to the absence of a convenient general term in Urdu and an overabundance of specific terms in English loans for the same purpose. The formation thus becomes a necessity. This is because Urdu has no adequate vocabulary in some aspects of life, particularly in science and law. For example, there is no native substitute for the English loan constituent *nəmbər* 'number' in the compound (7f). Urdu vocabulary for much of the government machinery has also long been discarded and thought to be obsolete now. Therefore, the word *sekretəry* in *sekretəry xəzana* 'secretary of finance' (9a) and the like do not have Urdu counterpart. Thus, the English loans are convenient to use in the hybrids.

The compounds formed about government machinery are the most among all compounds, which cover various aspects of life. The formations are generally related also to some food items, court and transport, and show various semantic features similar to those seen in hybrid compounds in Pennsylvania German. For example, *phol frut* '(a variety of) fruits' (7a) shows that they sometimes exhibit semantic duplications in the two constituents (one Urdu and the other English), which are synonyms. However, the compound refers to 'a variety of fruit'.

In short, the hybrid compounds of nouns and adjectives have various formations belonging to various aspects of life. Most of them are from the endocentric class, but there are a certain number of other examples, e.g. copulative. The English constituents may function both as head and modifier without any distinction whether a compound is right headed or left headed. Most of these compounds are on a peripheral target language pattern. They are formed by English loans with the combination of Persian and Arabic loans that already exist in Urdu.

This shows that English loans are not adaptable with Persian and Arabic affixes, but that they conveniently adapt with Persian and Arabic loanwords in compound forms.

This adaptability is not restricted to the formations of nouns and adjectives only. The dominant majority of these hybrid compounds by English and Perso-Arabic constituents is in verb formations. Winter (2008: 168) observes that it is doubtful whether it is possible to have borrowed verbs without any morphological integration at all. Borrowed verbs are normally either inflected according to a productive pattern in the TL or used in periphrastic forms, which avoid an inflection of the borrowed item proper. For example, an English loan verb *kritisaize* 'criticise' is used as *kritisaise kərna* 'to finish', because a loanword for expanding verbal repertoire must be facilitated by a device called a dummy verb.

# 5.3.2.2. Hybrid Compounds: Verb Formations

The use of verbs without any morphological integration in the native morphology is not possible. No verbs enter with their original structure, probably due to Urdu's morphologically complex verbal system. As seen with Persian and Arabic loan verbs, the verbal structure is formed by compounding the English adjective, noun, verbal noun or even a base verb with an Urdu dummy verb, e.g. kərna 'to do' (for active) and hona 'to be' (for passive). As noted in Chapter 4, the two are restructured forms of the Persian auxiliaries kərdən and budən/ʃodən respectively. Thus, the integration in compounding is on a peripheral TL pattern, and Persian auxiliary verbs used as dummy verbs in Urdu play a key role. It can be said that English loanwords combining with Urdu dummy verbs are the most possible and the most frequent forms of their adaptation with Persian-(Arabic) loanwords. Consider the following examples:

# **Table 10: Hybrid Verbs with English Loans**

a. kritisaize 'criticise' → kritisaize kərna 'to criticise'/ kritisaize hona 'to be criticised'
b. ərændʒ 'arrange' → ərændʒ kərna 'to arrange' / ərændʒ hona 'to be arranged'
c. renəveit 'renovate' → renəveit kərna 'to renovate' / renəveit hona 'to be renovated'
d. eksept 'accept' → eksept kərna 'to accept' / eksept hona 'to be accepted'
e. əpruv 'approve' → əpruv kərna 'to approve' / əpruv hona 'to be approved'
f. beg → beg kərna 'to beg' / beg hona 'to be begged'
g. responsəbilti → risponsəbiləti dena/ lena 'to assign/take responsibility'
h. rispons 'response' → rispons dena 'to respond'/rispons lena 'to take response'
i hæpi 'happy' → hæpi kərna 'to make happy' / hæpi hona 'to be happy'
j. æŋgri 'angry' → æŋgri kərna 'to make angry' / æŋgri hona 'to be angry'

The examples (10a-10f), e.g. *kritisaize* (10a), show that English base verbs are the bases with Urdu dummy verbs to derive verbs. The dummy verbs are in active and passive forms, but those in (10g-10h) are only in active forms. The latter ones i.e. *dena* 'to give' and *lena* 'to take' are also different in that they show handing over and taking over something e.g. a certain job. Unlike Persian and Arabic base verbs that may undergo modification to attach to a dummy verb, English base verbs, as in (10a-10f), do not undergo any morphological changes. An English base verb can only be used as a verbal noun, and the function of verb is adopted by a dummy verb. In general terms, verb borrowing is supposed to be the most complicated in the derivation of new words. Reed (1948: 243) notes regarding English loans in Pennsylvania German morphology that the verb conjugation is the same in the case of loanwords (English) as for native words (Pennsylvania German), and phonetic rules are observed accordingly; if the English verb ends with [d/t], it is conjugated just like German verbs with root-endings in

[d]. This shows morphological changes in the loan constituents. In contrast, there is no alternation in the English loan base words, in their integration with Urdu dummy verbs.

Moreover, in contrast to Persian/Arabic bases which may be nouns, adjectives or verbal nouns, the English bases may also include basic forms of verb. As mentioned earlier, discussing degrees of grammatical productivity, Dressler (2003: 35) also points out the integration of loanwords with unfitting properties in which foreign words which are not adaptable in the normal circumstances can be integrated into the system by a given derivational rule. For this, a rule must have maximum productivity in order to overcome the two obstacles of foreignness and unfitting properties. For example, English (and German) verbs have no thematic vowels, thus a thematic vowel must be added when adapting an English loan-verb into a Romance language: E. to dribble > It. dribbl-a-re, Sp. dribl-a-r. The English base verbs in Urdu i.e. kritisaize (10a), renoveit (10c) and beg (10f) with unfitting morphological properties can be integrated into the Urdu verbal system by obligatory derivational verbal constituents, e.g. kərna 'to do' (dummy verb) to form verbs as kritisaize kərna 'to criticise'. This shows that although Persian and Arabic affixes aren't productive and derive few words in combination with English loans, the dummy verbs are productive. They can accommodate not only English loan nouns, adjectives and verbal nouns but also base verbs, which cannot be integrated into other forms and with other categories of words for verbal capacity. Matras (2009: 182) argues that languages tend to adopt morphological adaptation strategies to accommodate complex loan verbs, which can be interpreted from full acceptance of the "verbness" to the denial and consequent need to reconstruct the verbness. Therefore, it can be said that the Urdu system of dummy verbs functions perfectly for the acceptance of English verbs which are otherwise not attested in Urdu morphology. There seem no alternations in English loan bases in the compound verbs.

The discussion of hybrid compounds, whether they are nouns, adjectives or verbs, shows that the integration of English loanwords is based on the pattern of Persian and Arabic loanword integrations with native Urdu words. However, this recent loanword integration may be far less (in amount) than the older ones.

Spread across a long period of time, the integration of Persian and Arabic words with native Urdu words caused an integration of the loan affixes in the Urdu morphological structures. The time period of English loanword integration is far shorter than that of Persian and Arabic. Therefore, it is noteworthy that English loanword integration has not yet caused any integration of English loan affixes in Urdu morphology. However, certain examples show that there is some usage of English loan affixes, mostly in an informal language.

# 5.4. English Loan Affixes in Urdu

For morphologically rich languages, like Urdu, a lot of grammatical information is expressed via affixes. Its rich variety in the affixation process is due to its nature of borrowing and absorbing a large number of loanwords and their affixes. Therefore, it can be expected that English affixes may also find some place in Urdu morphology. However, in contrast, English affixes have not yet been able to get a place in its morphological structures. They are used in certain positions but mainly in the informal language or in the media by the elite. Rubba (2004: 2) thinks that English has three inflectional categories. They are correlated with nouns in number, verbs in Tense/Aspect, adjectives (and adverbs) in comparison. The continuous use of English loanwords and code switching has caused these three types of loan inflections to enter Urdu. There may seem plural markers, adjectival markers and progressive markers. Although most are restricted to informal use, some may be coming up parallel to some of the present Urdu affixes whether native or Persian and Arabic loans, as shown below:

5a. Plural markers: -ies and -s

fərabies 'drinkers' / sərdars 'tribal lords'

b. Adjectival markers (with city names): -ite, -ean, -i

Karachiite / Lahorean 'a citizen of Karachi/Lahore':

c. Progressive marker (used as an inflectional suffix): -ing

kha 'eat' + ing  $\rightarrow kha$ ing 'eating'

d. Nominal markers: (emphasis) -ation

gərbər 'disorder' + -ation → gərbəration 'disorder'

The use of the same word in the increasing exposure to English (Smeaton 1973) can lead to the introduction of the patterns of such lending languages into the receiving language. This can be observed in English plural patterns, which are gradually getting attraction from the speakers of Urdu. The loan plural markers -s and -ies seem to make a place in Urdu morphology, if persistently used in the media. They are often used with English loan plurals, but they are also used with Urdu plurals, e.g. mullahs 'religious scholar' and sərdars 'tribal lords' in informal language. Although Urdu has gender marking, with this pluralisation by English markers, there is no distinction of gender, e.g. tfəpaties 'chapatis' (f) and nəmazies 'people praying' (m). It is only by assumption that a noun is differentiated in its gender.

The second important affixation comes with the adjective markers -ite and -ean that are affixed with city names. However, they are mainly restricted to print media. The two affixes -ite and -ean were known after they were first used in English papers, and it was gradually increased in talk shows in the electronic media. Otherwise, they are not generally used. The general public would rather use -i with any city name, e.g. Hyderabadi or even Lahori. This marker seems more adaptable in the context of its use with country names, e.g. Pakistani or

*Israeli*. Despite its common use internationally with many other countries, it is no less than a native adjectival marker with Pakistani city names e.g. *Mirpuri*, *Sejalkoti* and *Islamabadi* etc. The extent of its adoption cannot be claimed, though it looks like a native adjectival marker.

Other possible affixes are the English progressive marker -ing and the nominal marker -tion, used only rarely in the informal language. People may say kha 'eat'  $+-ing \rightarrow khaing$  (me 'I' +a:m 'msango' +khaing 'eating'  $\rightarrow$  me a:m khaing) rather than use a progressive from Urdu. However, it is not in accordance with the Urdu syntactic structure for progressive sentences, and does not seem to integrate in Urdu morphology. Although the nominal marker -tion has a higher chance of fusing into the system, it is equally uncommon. A little semantic change is seen in the use of -ation e.g. garbaration 'disorder'. The noun with its use is more emphatic.

Apart from plural markers, other affixes have fewer chances of integration. An interesting observation is the use of some affixes by a specific class of people. For example, the adjectival markers for city names, i.e. -ite and -ean, are mostly used by the elitist or educated people. This is also because they are frequently used in English newspapers e.g. Daily Dawn, the largest English newspaper. The progressive and the nominal markers i.e. -ing and -ation on the other hand are used by the common people, not very educated. Unlike the adjectival markers -ite and -ean, the progressive and the nominal markers -ing and -ation are restricted to informal language. Both do not seem to be integrated. Only English plural markers among all affixes have some possibilities of integration due to their use without any distinction.

The discussion on the English affixes and on the adaptation of English loans is summarised in the following section with some results drawn.

# 5.5. Summary of the Chapter and the Results Drawn

English loanwords are commonly used in various forms, and the nativisation shows morphological changes in gender marking, pluralisation, derivation, hybrid compounds, reduplication and the affixation of English morphemes. However, the discussion was restricted to a few most important points.

The native Urdu gender and number marking of English loans, e.g. daktərnijā 'doctors' (f), is in common use. The adaptability of loans is restricted in derivation by affixation, although they exhibit various affixation types from native Urdu, Persian and Arabic sources. The evidence of lexical creation, e.g. fradija/fradi, by native Urdu affixation shows that English loans are adaptable in Urdu morphology and may take the status of Persian and Arabic loans. The affixation of Persian and Arabic morphemes, e.g. fəlsəfijana 'philosophical' and borijəţ 'boredom', can also be seen with some loanwords. The derivation by compounding with Persian and Arabic constituents, e.g. finif kərna 'to finish' and siti nazım 'city mayor', is the most widespread adaptation of English loans after pluralisation. However, English loan constituents show no morphological changes. This restriction is clearly evident also in the nativisation of English affixes except for some instances e.g. the plural markers -s and -ies.

In sum, English loans are adaptable and native Urdu affixes are productive, in the process of inflection and derivation by affixation. Derivation by compounding of English loans is more frequent with Persian and Arabic loans. There are some results drawn from the discussion on the whole adaptation process.

# 5.5.1. Results Drawn from the English Loanword Adaptation

The description of the morphological adaptation of English loans in light of the three sources of morphological structures draws some findings, as shown below.

- 1. The productivity of the native Urdu affixes and the adaptability of the English loanwords show that Urdu affixes whether the gender and number markers (Section: 5.2.) or derivational affixes -i/-ija (Section: 5.3.) are productive and English loans are adaptable. Therefore, the integration takes place with the native Urdu affixes only.
- 2. Although the Persian and Arabic loan affixes function in Urdu, and may also be productive in the original languages as well, they are not generally productive with the English loanwords (Section: 5.3.) except for a few dummy verbs (modified from Persian auxiliaries), Persian adjectival marker -ana and Arabic nominal marker -ıjət. In addition to these few, there can rarely be any integration of English loanwords with the Persian and Arabic loan affixes. This is despite the fact that English loanwords are adaptable in Urdu, but their adaptability does not extend to the Persian and Arabic loan affixes.
- 3. When Urdu borrowed a word from Persian and Arabic, it also borrowed the derivational patterns (Chapters 3 and 4), and so various other derivational categories of the same word. In contrast, when Urdu borrows from English, it does not generally borrow the other categories of the same word. Rather, it derives from the loanwords according to its own derivational system. Words like *bəmbari* 'bombing' and *fəlsəfi* 'philosopher' (Section 5.3.) are good examples of the adaptation.
- 4. As derivational changes of English loanwords by affixation are far fewer than derivational changes of Persian and Arabic loanwords, it is clearly seen that the formation of new words by the combination of native Urdu affixes and English loanwords is limited. However, as the examples show (Section 5.3.), it is interesting to see a variety of new words derived. A process of lexical creation, e.g. *fradia/fradi*, with relevance to native Urdu morphology, is

also evident and even more interesting. The new words are a lexical creation rather than simply derivation, as there is no such derivation in the English lexicon for their meaning.

- 5. More frequent derivation of words is achieved however by hybrid compounding (Section 5.3.2.1.), not by affixation. The hybrid compounds derived by English loanwords with Persian and Arabic loanwords are greater in number than those with native Urdu words. It is noteworthy that the adaptability of English loans does not take place with Persian and Arabic affixes, but the formation of compounds with the older loans is not prohibited.
- 6. Although the patterns of English loanwords adaptation are the same as the patterns of Persian and Arabic loanword adaptations, the extent of English loanword adaptations is far less than Persian and Arabic loanword adaptation.
- 7. English loans generally show no structural changes with any affixes from the three sources i.e. native Urdu, Persian and Arabic. This is particularly seen in the English loan constituents in hybrid compounds (Section: 5.3.2.1) and is unlike the older Persian and Arabic loans whose constituents were often incorporated, as in Urdu copulatives (Chapter 4).
- 8. Section (5.4.) shows that English affixes have not yet found any proper place in Urdu morphology, except in some instances of plural markers -s and -ie. On rare occasions, they are used in informal language e.g. *voderas* feudal lords'. Moreoves, they are used as nominative plurals only, and there is no use as oblique forms.

These results drawn from the discussion on the use of English loan affixes and English loanwords adaptation require some general conclusion to be made in a separate chapter.

# Chapter 6

# **General Conclusion from the Discussion**

#### 6.1. Introduction

Persian, Arabic and English have been very influential languages in the world. A large number of the world's languages have borrowed from each of them. Urdu is unique in that it is a recipient language from them all with a large number of loanwords. However, the manner of borrowing is different. In the case of Persian and Arabic, the borrowing is historical and very old. The influence began from the early development of Urdu after the 12<sup>th</sup> century, and became even stronger in the Mughal era in the Indian subcontinent. The influence of English loanwords is more recent, and seen particularly in the last decade mainly after the introduction of electronic media.

There are various aspects of the linguistic influence from the three great languages, but unfortunately no research has been done into any of these aspects relative to borrowing into Urdu, whether lexical, phonological, morphological or syntactic. The aim of this thesis is to see how Persian, Arabic and English loanwords are morphologically adapted. Any research into the issue becomes even more interesting when one considers that two prominent languages, Persian and Arabic, have had a historical influence on Urdu with various morphological changes. The loanwords from them have played a key role in its early lexical and morphological development. Thus, Urdu morphological structures come from the three sources i.e. native Urdu, Persian and Arabic. Considering the morphological structures from these three sources as three aspects of Urdu morphology, the study looks at how the loanwords from English are adapted. English loawords are more recent and so treated as foreign. The hypothesis at the beginning of the study was that the affixation, whether

inflectional or derivational, might be on native Urdu patterns but that the compounding of English loans is more with the older loans (Persian and Arabic).

The present study is descriptive due to the non availability of relevant literature and non availability of any theory dealing with the morphology of loans. The bigger part of the study deals with Urdu morphological structures with reference to the three sources of lexical items. The next sections summarises the Urdu morphological structures first and then the adaptation of English loans in the light the three sources. They are followed by some discussion.

# 6.2. Summary of the Discussion on Urdu Morphology: Native Urdu, Persian and Arabic

The discussion begins in the first chapter with an introduction into the study and an overview of the sociolinguistic background of Urdu. Persian and Arabic loanwords play a part in the morphology of Urdu in particular, and south Asian languages in general. English loanword influence is much more recent, and so a brief comparison is made between the morphology of English loanwords in five south Asian languages i.e. Urdu, Punjabi, Sindhi, Pashto and Marwarhi. The study focuses in particular on gender and number morphology and the formation of compounds.

Chapters 2, 3 and 4 discuss various aspects of Urdu morphological features. Morphology is a comprehensive field of study. All of its aspects cannot be covered in a short work. Therefore, only some selected points, i.e. the gender, number and case morphology of nouns, derivational affixes and some types of compounds are elaborated on.

The native Urdu gender, number or case morphology is based on the alternation of vowels, which also shows nasalisation in feminine plurals. The native gender markers -a (m) and -i (f) are heavily dominant. Masculine is the basic gender and feminine is only a derived form.

Some masculine nouns also end with -i, and in this case the feminine nouns must end with -an. A large number of nouns do not have any specific masculine marking, and their feminine forms must end with -ni. These markers in general express gender rule. However, some native nouns deviate from this rule. The rule becomes further irrelevant in loanword morphology. Some Persian and Arabic loans take the Arabic feminine marker -a, which is a homophone of the native masculine gender -a. Persian in fact expresses no gender marking, and its loans in Urdu adopt native gender mostly.

There are also deviations in pluralisation. Native masculine nouns generally end with -e, if singular nouns end with -a, otherwise they remain unchanged. Feminine plurals end with  $-ij\tilde{a}$ , if their singular form takes -i. All other feminine nouns take the plural marker  $-\tilde{e}$ . Although most loans take native plural markers, some e.g. -an, and  $a\underline{t}$  (both masculine) from Persian and -in (m) and  $-a\underline{t}$  (f) from Arabic have also entered Urdu morphology and function like native plural markers. The most interesting factor in pluralisation is the non Arabic loans' formation of broken plurals, a feature that is only associated with Arabic loanwords. It shows the nativisation of the Arabic patterns of broken plurals, but it also shows deviation from the native Urdu and the Arabic pluralisation rules.

There are also deviations in the derivation by affixation. Urdu contains a variety of native and loan derivational affixes, discussed in Chapter 3. There are more prefixes than suffixes, but most derivations are made by suffixes. Native affixes are few and based on vowels mainly, but they are the most productive of all. Persian affixes are mainly prefixes and though largest in number, they only derive a few words. However, some affixes, e.g. *na* 'no/not' and *-ana* are quite productive and may attach to non Persian words very often. There are not many Arabic affixes, but some, e.g. *la* 'no/not' and *yær* 'not/without' (both negative markers) also

show productivity. The dominant among loan derivational affixes are negative markers, e.g. na, be from Persian and la, bila and yer from Arabic. All loan affixes express semantics. Some of them have interchange ability with or without semantic changes in the derivations. This shows variations in the Urdu derivation. They seem more often in the various patterns of Arabic derivations by modification of the base, as the patterns are nativised by taking affixes from native Urdu, Persian and even from Arabic. For example, the Arabic tri consonantal root zlm derives the noun zolm 'tyranny' and adjective zalm 'tyrant'. But the native -i and Persian -ana, make further derivations of adjectives, zolmi 'cruel man' zalimana 'oppressive'. Such derivations are frequent, although there are some irregularities in their formations.

Arabic loan derivational patterns by modification of bases also show an interaction between morphology and phonology. Many morphological changes are caused by phonological changes. Therefore, a discussion on some important aspects of Arabic loanword phonology was necessary. Although most Arabic phonemes lose their phonetic value and are replaced by native Urdu phonemes, some are still part of the Urdu phonetic inventory. Fricatives /x/,  $/\gamma/$  and the uvular stop /q/ were not only diachronically substituted for native phonemes in native words but also created new lexical items. Thus, they are now part of the Urdu phonetic inventory and function like native phonemes. However, as recently noted, in a shift back the borrowed velar fricatives /x/,  $/\gamma/$  and the uvular stop /q/ are now substituted by the velar stops  $/k^h/$ , /g/ and /k/. The data confirms Wohlin's (2010: 38) observation of the reversal of sounds regarding Indo European and Germanic languages, but the features in Urdu are different.

Phonology affects not only derivation by affixation but also derivation by compounding. In Urdu, compounding of constituents from two different sources is a frequent function. These hybrid compounds are more frequent than normal, i.e. native + native, combinations.

Therefore, a discussion with respect to the source languages in this aspect seemed irrelevant in chapter 4, which focuses rather on the various types of compounds. By types of classification, most compounds are endocentric and verbal compounds. Then, they include exocentric, copulative and postpositional compounds.

Urdu compounds may be left branching (modifiers that come before head) or right branching (modifiers that come after head). Both constituents are often synonyms or near synonyms, although they may be from two different sources e.g. in *zolm-o-sitom* 'tyrannies' N<sub>1</sub> is Arabic and N<sub>2</sub> is Persian. Urdu compounds are generally formed by distinctive compounding words, each of which may form several combinations; they may also have infixes that structurally link both constituents. There are three infixes *-e-*, *-o-* and *-b-* that perform grammatical, semantic and phonological functions. They are mostly found in endocentric and copulative compounds. No verbal compounds contain any infixes. The key constituents in verbal compounds are dummy verbs, the modified forms of Persian auxiliaries.

Dummy verbs are a major tool for verb formation in Urdu, particularly by compounding with loanwords. This phenomenon of hybrid verb formation is also very frequent in English loanword adaptation, which is discussed in in Chapter 5.

### 6.3. English Loanword Morphology in the Light of Persian and Arabic Morphology

The central hypothesis is shown to be true, that the affixation of English loanwords, whether inflectional or derivational, tends to be on native Urdu patterns. The compounding of English loans is more like that of the older loans (Persian and Arabic), which form a bigger part of the Urdu Lexicon and function like native Urdu affixes. The study has also found that English loans are integrated more with native Urdu affixes. The adaptation of English loans is not as complete as that of the two older loan influences, and so Urdu does not generally borrow any

derivations of a loan, as it did with Persian and Arabic with or without any morphological changes. The related supplementary questions regarding the structural patterns, the manner of loanword modification, the elements affected and the English loan affixes support this fact. Urdu does not generally borrow English affixes, but some, e.g. plural markers, are in the informal language.

Examining the questions, in detail, the study has found that English loans are adaptable and their integration takes place with native Urdu affixes, which are productive and conveniently attach to loanwords. There are both inflectional and derivational changes. Inflectional changes are extreme and only based on native Urdu patterns. English loan nouns take the native gender markers -a and -i and the plural markers -e (m) and  $-ij\tilde{a}/-\tilde{e}$  (f).

In derivation by affixation, English loans are generally more adaptable with native affixes, which are productive enough. Words like *fradija*, 'one who commits fraud' even show lexical creation. Although Persian and Arabic loan affixes function like native affixes in Urdu, and may be productive in the original languages as well, they are not generally productive with English loanwords. Except for a few dummy verbs, such as the Persian adjectival marker -*ana* and the Arabic nominal marker -*ıjət*, no other Perso-Arabic element is integrated with English loans. This is despite the fact that the latter is adaptable, but the adaptability does not extend to the former and so the integration cannot occur freely.

However, frequent derivation is made mainly by the verb formation of English loanwords with Urdu dummy verbs. Although the pattern of verb formation is native, the major dummy verbs are the restructured forms of Persian auxiliaries. This makes compound verbs according to the peripheral target language system. These hybrid formations are greater in number than those with any other words from the three sources. It is interesting to see that English loans

are not adaptable with Persian and Arabic affixes, but the formation of compounds with the older loans is not prohibited.

The adaptability of the recent loans is limited irrespective of the morphological patterns they follow. It is far less than the adaptation seen in Persian and Arabic loans. There are two factors, 1) the influence spread across centuries' long period and 2) Urdu's own developmental stage differentiate the capacity of the past and present influences. English loanword influence is spread over barely a century and a half. Its extreme form is seen during the last few decades, particularly with the introduction of electronic media. Urdu is already a well established language and the huge language contact has caused only some morphological changes. In contrast, the Persian and Arabic influence began with the early development of Urdu, nearly a thousand years back.. Therefore, it is natural to see far more morphological features in the two older languages than in the latter one.

Another point is the morphological systems of the lending languages, Perso-Arabic and English in this case, that may affect borrowing languages, as Urdu in the present case. AlJabri (2010: 3) argues that English has a morphology which is not as productive in its overall derivation as some other morphological systems such as those found in Semitic languages (e.g. Arabic). With lexical borrowings from Persian and Arabic, Urdu also borrowed derivational patterns. Therefore, a number of other derivational categories of the same word can be seen. These derivational patterns are now part of Urdu morphology, as the affixes in the loan derivational patterns may attach to the bases other than from their source languages. Thus, they function like native affixes. However, this is not true with English loanwords. Urdu does not normally borrow from English the other categories of the same word. Rather, it derives words according to its own derivational system. Words like *bombari* 'bombing' are

good forms of the adaptation. This shows that English derivational (and inflectional) morphemes have not been integrated into Urdu. This is why English loanwords depend on native Urdu morphemes or in some cases Persian and Arabic morphemes already present. The only English loan affixes, sometimes used mainly with English loanwords, are the plural markers -s and -ies. On rare occasions, their affixation with Urdu bases, e.g. mullahs 'priests' also occur in informal language. However, the proposition here is that there is some lexical conditioning for it. The nouns ending in -a and -i e.g. mulla(h) 'priest', vədera 'feudal lord' and vahabi 'a religious sect' may be the convenient bases for English loan plural markers. The nouns are similar to English nouns e.g. visa, baby, which are more easily adaptable than others, possibly because of the harmony in the endings of such nouns in the two languages. This harmony in the endings suggests a possibility of English plural markers entering into Urdu. Presently, as mentioned above, their occurrence is very limited to a specific class of people. The adjectival markers -ite and -ean are in even more limited use.

Thus, English loanwords have brought into Urdu no inflectional or derivational patterns, unlike Persian and Arabic morphological patterns. Therefore, the creation of new lexical items and new forms also cannot be expected. It is possible when some patterns have been adopted, which can be noted in the case of Arabic loanwords, e.g. a broken plural form *xəvaţin* and a formation of a noun from a noun e.g. *xəbərnjəţ* 'status of news'. However, there are some examples of neologism, e.g. *fradna/fradi*. The new words are a lexical creation rather than simple derivation, as there is no derivation parallel to these in the English lexicon for their meaning. Al-Jabri (2010: 8) argues that, in general, current approaches to lexicalisation tend to treat concepts as related entities; words, on the other hand, are viewed as isolated items that just happen to be attached to concepts. Both the creations *fradna* and *fradi* refer to a person who gets personal gain by treacherous means. They fulfil the lexical need to represent

a person of treacherous acts. Although the Persian loan <code>doybaz</code> 'cheater' or native Urdu derivation <code>dhokebaz</code> are its synonyms and already present in Urdu, there is a key semantic difference between the older forms and the new creations. The older forms are used in the generic sense, while the new creations are used in a specific sense for the one who gets some financial benefits by treacherous means. This shows that English loans have the ability of lexicalisation for a specific semantic concept.

Nevertheless, this explanation is only partly true. More importantly, many English loanwords have retroflex sounds which make them distinct from Persian and Arabic loans. It is also necessary to note that retroflex sounds are seen in South Asian English, not in British or American English. The lexical creations are not due to loanwords' phonemic influence, as seen in the creations by Arabic sounds. Along with lexical borrowings from Arabic, some Arabic phonemes, e.g. the velar fricatives /x/ and /y/ also entered Urdu. In contrast, no English phonemes entered Urdu. However, there is a phonological harmony between many (South Asian) English loanwords and native Urdu words, which have the retroflex sounds /t/ and /d/. This common feature increases the chances of neologism by English loanwords in general. This is verified by many other similar loans, when they conveniently take less frequent feminine markers, e.g. -ni and -on in daktorni 'lady doctor' and fradon. Due to the gender distinction in Urdu, the original loans are treated as masculine, the basic gender, and they only need to be feminised. Thus, it can be claimed that any English loan with retroflex sounds will probably be more conveniently adapted and lexicalised than other loans. This is expected also due to the fact that the loans with a retroflex sound, e.g. total and darve more conveniently enter the Urdu lexicon and substitute the Urdu counterparts, i.e. kol and t/holan, which are nearly obsolete now.

This further confirms the adaptability of English loanwords however it may be limited when compared to that of Persian and Arabic loanwords. Lee (2004: 1) considers that change in a language occurs either quickly or slowly; the rate of change is unpredictable. There are only some morphological changes in English loanwords, but there is a process of change and English loanwords adapt in Urdu morphology.

However, there are no morphological changes in the English constituents in the hybrid compound verbs when combined with Urdu dummy verbs. Even if there are English loan verbs, they cease to function as a verb, and the verbal capacity is adopted by the dummy verbs. Urdu adaptation morphology is functional in establishing the verbness of loan verbs, but without any morphological changes in them. This is unlike the older Persian loan auxiliaries which function as dummy verbs in their restructured forms and have diachronically helped in establishing a system of compound verb morphology in Urdu in the absence of a base verb. Matras (2009: 182) argues that languages adopt morphological adaptation strategies to accommodate complex loan verbs. English loan verbs alone cannot function with their verbal capacity. Therefore, the Urdu system of dummy verbs functions perfectly to accommodate them, which are otherwise not adaptable. The dummy verbs function the same way with bases from native Urdu, Arabic or English. As required, they show tense/aspect changes, but no changes are seen in the English loan constituents. Therefore, the adaptation of English loans in compound verbs seems no different from the older loan bases. Like older loans, they only follow the set patterns established by Persian auxiliaries. However, certain examples, e.g. Persian noun xəriq 'purchase', show that older loans may adopt the status of base verbs without requiring a dummy verb. It is not possible with English loans, and so their adaptation is far less than that of Persian and Arabic loans.

Morphological changes in English loans follow the morphological patterns formed by native Urdu, Persian and Arabic. This seems to be a similar situation to the one seen in Turkish loanwords in Aivaliot (a dialect of Greek). Ralli (2011: 185) argues that Turkish loans are adapted to Aivaliot morphology, following specific requirements, which are imposed by the most productive derivational and inflectional processes of Greek, but they are also constrained by features innate to the donor language, i.e. Turkish. There are constraints in English morphological rules functioning in Urdu, as the loan affixes are not active. English loanwords only follow some requirements imposed by derivational and inflectional patterns of the native Urdu, Persian and Arabic. The changes in them are restricted to pluralisation mainly, and therefore they have not yet played any role in Urdu development.

Baik and Shim (1998: 276) advise, "One must not forget that language change is the result of dynamic interaction between internal linguistic development and the external socio-psychoecono-political factors in language contact situations". Both the factors have played a role in the language change in Urdu as well. Some major factors are given in the following section.

# 6.4. Factors Influencing the Degree of Adaptation

Internal Linguistic development is an ongoing process, which is also evident in Urdu. One major source of linguistic development is language contact. Persian and Arabic language contact with Urdu contributed to its linguistic development from the earliest times. In contrast, the role of English is very limited due to the shorter period of language contact. Although English has lent no morphological patterns, it does play a role in the enrichment of the Urdu lexicon. The use of the same word in the increasing exposure to English with time and frequency (Smeaton 1973) in the long run, can lead to the introduction of the patterns of such a lending language into the receiving language (Urdu in the present case). This can be

seen in the original plural patterns of English loans, which are gradually increasing in their uses by the speakers of Urdu.

Contact between the two languages correlates the linguistic nature of both, which is also a major factor influencing the adaptation process. The mechanism of native Urdu word formation is important. The productivity of its affixes associates with the adaptability of English loanwords to adopt a morphological pattern. English loans are adaptable as much as native affixes are productive. They do not associate with Persian/Arabic affixes, which though productive in Urdu and in their original languages are unproductive with English loans in Urdu. However, their compound formation with the older loans is not prohibited. Thus the productivity and the adaptability function in due course for their integration.

Ralli (2011: 185) argues that the morphological adaptation of loans is mainly constrained by three factors: (i) the degree of intensity of socio-cultural contact in a language-contact situation (Thomason and Kaufman 1988); (ii) the recipient-language word-formation mechanisms, and (iii) the productivity of morphological patterns. These three factors in fact help in understanding Urdu morphological patterns. The second and third have just been seen, and so the focus is on the first one now. Language contact between two languages depends mainly on external socio-psycho-econo-political factors, which are thus even more important generally and even more extreme in Urdu. The first among them is the use of electronic media. Consistent use of English affixes, e.g. plural markers -s and -ies by the elite in the electronic media may cause the formation of an English morphological pattern in Urdu. As noted in the first chapter, English is considered to have a high status and prestige as a global language. The use of English loanwords in daily Urdu is not just a status symbol it has become a social habit. The use of plural markers is rare, but it is the new phase of the social

habit. The use may vary and may thus influence the degree of the adaptation process. It is in the highest degree with plural marking and the hybrid compounds but not with the derivation by affixation. This is mainly because of the attitude of the speakers for whom the use of nativised English plurals and the hybrid compounds is convenient.

The increasing education and literacy rate is also a factor for this. Although the huge majority of people are still uneducated, ordinary people continue to get education. English is the medium of instruction in all private educational establishments and from higher secondary classes and onwards in all public institutes. This makes the use of English loans – or at least exposure to them – increase day by day. The dominant role of electronic media and the presence of a large number of English language centres have made it even more rapid. The rapid increase in their use causes loanwords to undergo more nativisation and makes it more possible for their adaptation.

# 6.5. Implications of the Study

There is a scarcity of linguistic research in Pakistan except for a small amount of work done by a few individuals. This conclusion can be reached from the fact that there is just one Department of (theoretical) Linguistics established by Azad Jammu and Kashmir University a few years back, and one Journal of Linguistics published by the same university. Urdu is no exemption from this scenario. Thus, this study is the first attempt in the research on Urdu morphology. It therefore forms the basis for further research in various aspects of this field.

The first area of further research is the adoption of a theoretical approach. It is now convenient in terms of data analysis, as a description is already present. It is suggested that the lexeme based approach is suitable for future study. Words as the bases are present whether

they are native words or loanwords, and the affixes of one source attach to the words of the other source.

The study of loanword morphology in Urdu within a lexeme based model can help the development of a theory of loanword morphology in general. The absence of such a theory was strongly felt at the beginning of this study, and so it became the reason for the descriptive work done. There are two major points that can specifically be dealt with by the development of a theory of loanword morphology. First, it can help in the evaluation of the morphological adaptation phenomena by loanwords in general. Secondly, it can provide a tool to compare and discuss the linguistics issues involved in the case of a language which is the recipient language from more than one language. The data description of Urdu provides a good base.

There is also a possibility that an OT based approach may be adopted to see the constraints in the morphological adaptations of loans in Urdu. Loanwords from various sources present an interesting base for the comparative analysis of the constraints.

OT can be particularly helpful in the morphology-phonology interface, which is also observed in this work, e.g. affixation of the Arabic loan definite article  $\sigma l/\partial l$  and assimilation of the lateral. This can be a rich area of linguistic research, for Urdu, as only some surface elements were highlighted. There are other important aspects, for example the role of a non neutral affix, e.g. -gi that causes the phonological changes in the stems. There must be other non neutral affixes, which need to be seen.

Due to this morphology-phonology interface, Arabic loanword phonology functioning in Urdu had to be highlighted. This shows that some Arabic loan phonemes not only substitute for native phonemes in native Urdu words but also function in combination with native

retroflex phonemes in lexical creations. The introduction of Arabic phonemes in Urdu was a diachronic process and provides us with another rich area to study the process and its present function as compared to the native Urdu phonemes. Moreover, as many phonemes are now being substituted by native phonemes even in Arabic loanwords, it shows the reversal of Arabic sounds from Urdu. This presents another idea which can broaden the discussion introduced, on the rise and fall of Arabic phonemes in Urdu.

\*

# References

- Abdallah, A. S. 2006. *Teaching Word Roots and the Broken Plural*. London: A Project by CETL School of Oriental and African Studies, University of London.
- Abbas, S. (2002). "Language of the Armies, Urdu: A Derivative of Persian and Avestan" In *Persian Language and Literaterature*. Iran Chamber Society. Retrieved: 02/2011 http://www.iranchamber.com/literature/articles/language\_of\_armies.php.
- Ahmed, R. 2008. "Unpacking indexicality: Urdu in India". Texas Linguistic Forum, 52: 1-9.
- Ahmed, T. 2008b. "Instrument as Path: Reconciling the Semantic Usages of Urdu *se* and Pashto *pa*." Paper presented at *Universität Konstanz*, Germany. Retrieved. 9/2010. http://www.ling.uni-konstanz.de/pages/home/tafseer/instrument.pdf.
- Aronoff, M. 1976. Word Formation in Generative Grammar. Cambridge, MA: MIT Press.
- Aronoff, M. 1996. Morphology by Itself. Cambridge, MA: MIT Press.
- Azim, A. (1975) "Urdu A Victim of Cultural Genocide?" In Zafar Imam (ed.) *Muslims in India*. New Delhi: Orient Longman, pp. 256-272.
- Baik, M. J. and Shim, R. J. 1998. "Martin Jonghak Baik and Rosa Jinyoung Shim: Reply" *World Englishes*, 17(2): 273-279.
- Bauer, L. 2003. Introducing Linguistic Morphology. Edinburgh: Edinburgh University Press.
- Bauer, L. 2005. "Les Composés Exocentriques de L'anglais". In D. Amiot (ed). *La Composition Dans une Perspective Typologique*. Arras: Artois Presses Université. pp 35-47.
- Baugh, A. and Cable, T. 1978. *A History of the English Language* (3<sup>rd</sup> Edition). Englewood Cliffs: Prentice-Hall Inc.
- Baumgardner, Robert J. 1998. "Word-Formation in Pakistani English" *English World-Wide*, Volume. 19, Number 2, John Benjamins Publishing Company. pp. 205-246(42).
- Baumgardner, Robert J. 1990. "The Indigenisation of English in Pakistan". *English Today*.6: Cambridge University Press. pp 59-65.
- Benczes, R. 2004. "On the Analysability of English Exocentric Compounds". *Jezekoslovija*, 5.1-2: pp 1-21.
- Bogel, T., Butt, M., Hautli A., and Sulger, S. 2007. "Developing a Finite-State Morphological Analyzer for Urdu and Hindi". In *Proceedings of the Sixth International Workshop on Finite-State Methods and Natural Language Processing*. Potsdam.
- Booij, G. 1992. "Compounding in Dutch". Rivista di Linguistica, Volume 4(1) pp. 37-59.

- Booij, G. 2005a. *The Grammar of Words: An Introduction to Linguistic Morphology*. Oxford: Oxford University Press.
- Booij, G. 2005b. "The Interface between Morphology and Phonology: A Comment on Ellen Kaisse: Word Formation and Phonology". *Skase Journal of Theoretical Linguistics*, pp. 246-99.
- Bukhari, N. H. 2009. *The Syntax of Serial Verbs in Gojri*. Ph.D Thesis, Newcastle University, United Kingdom.
- Butt, M. 1995. *The Structure of Complex Predicates in Urdu*. Stanford, California: CSLI Publications.
- Capuz, J. G. 1997. "Towards a Typological Classification of Linguistic Borrowing. *Revista Alicantina de Estudios Ingleses*, 10: 81-94.
- Carlisle, J. F. 1987. "The Use of Morphological Knowledge in Spelling Derived Forms by Learning-Disabled and Normal Students". *Annals of Dyslexia*, (volume 27): 90-108.
- Carstensen, B. 1986. "Euro-English". In D. Kastovsky and A. Szwedek (eds). *Descriptive, Contrastive and Applied Linguistics* Berlin, New York: Mouton de Gruyter, 827-835.
- Crawford, C. (2008). "An Evolutionary Account of Loanword-Induced Sound Change in Japanese". *University of Pennsylvania Working Papers in Linguistics*. Vol. 14. Iss:1. Proceedings of the 31<sup>st</sup> Annual Penn. Linguistics Coloquium.
- David, A., Maxwell, M., Browne, E. and Lynn, N. 2009. *Urdu Morphology*. Maryland: Centre for Advanced Study of Language. University of Maryland.
- Davenport, M. and Hannahs, S.J. 2005. *Introducing Phonetics and Phonology*. London, New York: Hodder Arnold.
- Dressler, W. U. 1993. "Word-Formation: Poetic License". *The Encyclopaedia of Language and Linguistics*, 9: 5028-5029.
- Dressler, W. U. 2003. "Degrees of Grammatical Productivity in Inflectional Morphology". *Italian Journal of Linguistics/Rivista di Linguistica*, 15(1): 31-62.
- Durani, N. 2007. *Typology of Word and Automatic Word Segmentation in Urdu Text Corpus*.

  M.S Thesis, National University of Computer and Emerging Sciences. Lahore, Pakistan.
- Grimes, B. 2000. "Pakistan". In Ethnologue. *Languages of the World*. 14<sup>th</sup> ed. Dallas, Texas Summer Institute of Linguistics.
- Gumperz, J. J. and Wilson, R. 1971. "Convergence and Creolisation: a Case from the Indo Aryan/Dravidian Border in India". In D. Hymes (ed.). *Pidginization and Creolization*

- of Languages. Cambridge: Cambridge University Press, 151–167.
- Hafez, O. 1996. "Phonological and Morphological Integration of Loanwords into Egyptian Arabic". Cairo. *Monde Arabe, Première Série*, 27: 383-410.
- Hardie, A. 2003. *The Computational Analysis of Morpho-Syntactic Categories in Urdu*, PhD Thesis, Lancaster University.
- Hardie, A. (2003). "Developing a Tagset for Automated Part-of-Speech Tagging in Urdu". In Archer, D, Rayson, P, Wilson, A, and McEnery, T (eds.) *Proceedings of the Corpus Linguistics*. pp 298-307
- Haugen, E. 1950. "The Analysis of Linguistic Borrowing". Language, 26: 210-231.
- Hoffer, B. 1980. "English loanwords in Japanese: some cultural implications". *Language Sciences*, 12 (1): 1-21.
- Hüning, Matthias (2010). "Adjective + Noun Constructions between Syntax and Word Formation in Dutch and German". In Sascha Michel and Michael Onysko (eds.), *Cognitive Perspectives on Word Formation*. Berlin: Mouton de Gruyter. pp 195-215.
- Hussain, S. 2004. "Letter-to-Sound Conversion for Urdu Text-to-Speech System". In the *Proceedings of Workshop on Computational Approaches to Arabic Script-based Languages*, COLING, Geneva.
- Ijaz, F. and Moin, M. R. 2003. Gender and number generation of Urdu nouns. Islamabad. *CRULP*. pp 52-56.
- Jakobson, R. 1938. "Sur la theorie des affinities phonologiquesentre des lang" ues. *Actes du Quatrieme Congres Internationalede Linguistes*, pp 48-59.
- Junior, G. R. (2005) "The Languages of the World" In *Ethnologue* 15<sup>th</sup> Edition Dallas Tx SIL International Online Version. <a href="http://www.ethnologue.com/">http://www.ethnologue.com/</a>. Retrieved (04/2008).
- Kahnemuyipour, A. 2000. "On the Derivationality of Some Inflectional Affixes in Persian" Chicago. Paper presented at the Linguistic Society of America Annual Meeting, *LSA*.
- Kaisse, E. 2005. "The Interface between Morphology and Phonology". In: P. Štekauer and R. Lieber. *Handbook of English Word-Formation*. Dordrecht: Springer, 25-47.
- Kashif, D. 2004. Jadeed Dersee Urdu. Karachi: Qamer Kitab Gher 'Qamer Book House'.
- Katamba, F. 1993. *Morphology*. London: Macmillan Press Limited.
- Kent, D. B. 1999. "Speaking in Tongues: Chinglish, Japlish and Konglish". *KOTESOL* Proceedings of PAC 2. (The Second Pan Asian\_Conference). Seoul.
- King, R. E. 2000. The Lexical Basis of Grammatical Borrowing: A Prince Edward Island

- French Case Study. Amsterdam, Philadelphia: John Benjamins.
- Keane, E. 2001) Echo Words in Tamil. PhD Thesis, Trinity College, Oxford.
- Koul, O. N. 2008. Modern Hindi Grammar. Springfield, USA: Dunwoody Press.
- Lambert, M. B. 1924. *A Dictionary of the Non-English Words of the Pennsylvania German Dialect*. Lancaster: PA
- Lee, J. J. 2004. "The Impact of English on the Post-1945 South Korean vernacular. In *Karen's Linguistic Issues*. Retrieved: http://www3.telus.net/linguisticsissues/koreanvernacular.
- Lieber, R. and Scalise, S. 2006. "The Lexical Integrity Hypothesis in a New Theoretical Universe". *Lingue e Linguaggio*,1: pp 7-32.
- Martindale, J. 2009. *Languages of India*. Sydney: Martindale's Language and Translation Centre.
- Matras, Y. 2009. "Formal and Functional Aspects of Verb Borrowing". In *Morphologies in Contact*. An International Symposium Financed by the Volkswagen-Foundation. Universitat Bremen.
- McGregor, R.S. 1972. Outline of Hindi Grammar. Oxford: Clarendon Press.
- Megerdoomian, K. 2000. "Unification-Based Persian Morphology". In *Proceedings of CIC* Ling. Alexander Gelbukh, ed. Centro de Investigación en Computación IPN, Mexico.
- Mohanan, T. 1994. Argument Structure in Hindi. PhD Thesis, Stanford University, CA.
- Moizuddin, M. 1989. Word Forms in Urdu. Islamabad: National Language Authority.
- Müller, M. 1875. Lectures on the Science of Language. New York: Scribner.
- Nawaz, S. 2002. "Deletion Rules in Urdu Language". CRULP, pp 220-222.
- Naim, C.M. 1999. *Introductory Urdu*. Cambridge: University of Chicago.
- Olsen, S, 2000. "Copulative Compounds: A Closer Look at the Interface between Morphology and Syntax". In G. Booij and J. van Marle (eds.). *Yearbook of Morphology*. Dordrecht: Kluwer. pp 279-320.
- Olsen, S. 2000. "Composition". In G. Booij, C. Lehmann and J. Mugdan (eds), *Morphologie/Morphology* 1. Berlin, New York: Mouton de Gruyter, pp 897-916.
- Paradis, C. and LaCharité, D. 2001. "Guttural Deletion in Loanwords". *Phonology*, 18.2: pp 255-300.
- Plag, I. 1999. Morphological Productivity: Structural Constraints in English Derivation,

- Berlin: Mouton de Gruyter.
- Plag, I. 2003. Word-formation in English. Cambridge: Cambridge University Press.
- Poplack, S., Wheeler, S. and Westwood, A. 1989. "Distinguishing Language Contact Phenomena: Evidence from Finnish-English Bilingualism". *World Englishes*, pp 389-406.
- Qureshi, A. H. and Akram, M. 2008. "Hyberdization in Urdu and Inter-Cultural Perspective". Jamshoro, Pakistan. Sine nomine (no named publisher). pp -1-12.Retrieved. 06/2011 http://www.victimsofterror.info/wp-content/uploads/2011/03/HYBRIDIZATION-IN-URDU-Jamshoro.pdf.
- Radcliffe, R. 1990. "Arabic Broken Plurals". In M. Eid and J. McCanhy (eds). *Perspectives on Arabic Linguistics II*. Amsterdam: John Benjamins, 94-119.
- Rai, A. 2000. Hindi Nationalism. New Dehli: Orient Longman Private Limited.
- Ralli, A. 2011. "Morphology in Language Contact: Verbal Loan-Blend Formation in Asia Minor Greek (Aivaliot)". In Th. Stolz, M. Vanhove, H. Otsuka and A. Urdze (eds.), *Morphologies in Contact (Studia Typologica 12)*. Berlin: Akademie Verlag, 185-201.
- Reed, C. E. 1948. "The Adaptation of English to Pennsylvania German Morphology". *American Speech*, 23: pp 239-246.
- Rehman, T. 1998. "Linguistics in Pakistan [Canada]: A Country Report". In Rajindra Singh *The Yearbook of South Asian Languages and Linguistics* (ed) (Delhi: Sage Publications,). pp 184-196.
- Rehman, T. 2004. "Language Policy and Localization in Pakistan: Proposal for a Paradigmatic Shift, Crossing the Digital Divide". *SCALLA Conference on Computational Linguistics* 5th to 7th January, Katmandu.
- Rehman, T. 2005. "The Muslim Response to English in South Asia: With Special Reference to Inequality, Intolerance and Militancy in Pakistan". *Journal of Language Identity and Education*. Volume 4, Number 2. Lawrence Erlbaum Associates Inc. pp 119-135.
- Riaz, K. 2007. "Challenges in Urdu Stemming (A Progress Report)", In *Proceeding of BCS IRSG Symposium: Future Directions in Information Access (FDIA)*.
- Rizvi, S. M. J. 2007. *Development of Algorithms and Computational Grammar for Urdu*. PhD. Thesis, Pakistan Institute of Engineering and Applied Sciences.
- Rubba, J. 2004. *An Overview of the English Morphological System*. San Luis Obispo: California. Polytechnic State University.
- Ryding, K. C. 2005. *A Reference grammar of modern standard Arabic*. London: Cambridge University Press.

- Sankoff, G. 2001. "Linguistic Outcomes of Language Contact". In P. Trudgill, J. Chambers and N. Schilling-Estes, eds. *Handbook of Sociolinguistics*. Oxford: Blackwell, pp 638-668.
- Sapir, E. 1921. *Language: An Introduction to the Study of Speech* (Chapter IX. P. 2.). New York: Harcourt Brace, New York: www.bartleby.com (Online Books Publisher) (Date accessed: January 2009).
- Schach, P. 1948. "Hybrid Compounds in Pennsylvania German". *American Speech: A Quarterly of Linguistic Usage*, 23: pp 121-134.
- Schmidt, R. L. 1999. Urdu: An Essential Grammar. London. Rutledge.
- Seidenberg, M.S. and Gonnerman, L. M. 2000. "Explaining Derivational Morphology as the Convergence of Codes". *Trends in Cognitive Sciences*, 4 (9): pp 353–361.
- Shahid, S. R. 2002. "A Study on Glottal Stops in Urdu". Islamabad. CRULP Annual Report.
- Shah, S. M. 2002. "Urdu Nasal Consonant and Phonological Behaviour". Islamabad: *Akhbar-e–Urdu*. Muqtadra Qaumi Zuban. 'Urdu Language Authority'.
- Shapiro, M. C. 2003. "Hindi". In C. George and D. Jain (eds.). *The Indo-Aryan Languages*. Routledge. pp 250-285.
- Smeaton B. H. 1973. Lexical Expansion due to Technical Change, as Illustrated by the Arabic of Al Hasa, Saudi Arabia. Bloomington: Indiana University Publications.
- Sogaard, A. 2004. "Compounding Theories and Linguistic Diversity". In: Frajzyn-Gier, Zygmunt, Hodges, Adam & Rood, David S. (eds.), *Linguistic Diversity and Language Theories*. Amsterdam: John Benjamins, pp 319-337.
- Takashi, K. (1990). "A Sociolinguistic Analysis of English Borrowings in Japanese Advertising Texts", *World Englishes*, Volume 9, 327-341.
- Tatsioka, Z. 2008. "Code-switching and English Loanwords in the Greek Media". *Newcastle Working Papers*, 14: 129-144.
- Thakur, A. K. 1997. *Genitive Construction in Hindi*. M. Phil Thesis, University of Delhi.
- Thomason, S. G. 2001. Language Contact. Washington, DC: Georgetown University Press.
- Thomason, S. and Kaufman, T. 2008. *Language Contact, Creolization, and Genetic Linguistics*. Berkeley: University of California Press.
- Vajpeyi, K. 1958. *Hindi Shabdanushasan*. Kashi: Nagri Pracharni Sabha.
- Verdaguer, I. 1996. "Making Sense of Neologisms". English Teaching Forum, 34 (3): 98-106.

- Versteegh, K. 2001. "Linguistic contacts between Arabic and other languages". In *Arabica*, *Linguistique Arabe: Sociolinguistique et Histoire de la Langue*. pp 470-508.
- Waqar, A. and Waqar, S. 2002. "Identification of Diphthongs in Urdu and Their Acoustic Properties". *CRULP*, pp 16-26.
- Watson, J. C. E. 2002. *The Phonology and Morphology of Arabic*. London: Oxford University Press.
- Wegener, H. 1999. "Die Pluralbildung im Deutschen ein Versuch im Rahmen der Optimalitäts theorie". Linguistik Online 4, 3/99. <a href="http://www.linguistikonline.de">http://www.linguistikonline.de</a>
- Wegener, H. (2002). "Zur Funktion des s-Plurals im heutigen Deutsch". In Wiesinger, P.(ed.), Akten des X. Internationalen Germanistenkongresses Wien 2000 "Zeitenwende– Die Germanistik auf dem Weg vom 20. ins 21. Jahrhundert". Vol. 2: Entwicklungstendenzen der deutschen Gegenwartssprache. *Lexikologie und Lexikographie*, Bern et al.: Lang. pp 109-116.
- Wegener, H. 2003. "Normprobleme bei der Pluralbildung fremder und nativer Substantive". *Linguistik Online* 16, 4/03. <a href="http://www.linguistik-online.de">http://www.linguistik-online.de</a>.
- Wegener, H. 2004. "Pizzas und Pizzen die Pluralformen (un)assimilierter Fremdwörter im Deutschen". *Zeitschrift für Sprachwissenschaft*, 23: pp 47-112.
- Winford, D. 2003. An Introduction to Contact Linguistics. London: Blackwell.
- Winter, F. E. 2008. "Studying Loanwords and Loanword Integration: Two Criteria of Conformity". *Newcastle Working Papers*, 14: 156-176.
- Wohlin, A. 2010. "A 5-Dimensional Analysis of Linguistic Data"- Part II'. Retrieved on 22 February 2011, from: www.u5d.net/Linguistics/Part%20II-45p.pdf
- Wurzel, W. U. 1984. Flexionsmorphologie und Natuerlichkeit. Berlin: Akademie-Verlag.